

Received 3 July 2023, accepted 13 July 2023, date of publication 17 July 2023, date of current version 28 July 2023. *Digital Object Identifier* 10.1109/ACCESS.2023.3296152

# **SURVEY**

# **Transmission Network Planning in Super Smart Grids: A Survey**

## MUHAMMAD ADNAN<sup>®1</sup>, (Member, IEEE), YAZEED GHADI<sup>®2</sup>, (Member, IEEE),

IJAZ AHMED<sup>1</sup>, (Member, IEEE), AND MANSOOR ALI<sup>103</sup>, (Member, IEEE)

<sup>1</sup>Department of Electrical Engineering, National University of Computer and Emerging Sciences (FAST), Chiniot-Faisalabad Campus, Chiniot 38000, Pakistan <sup>2</sup>Department of Software Engineering, Al Ain University, Al Ain, United Arab Emirates

<sup>3</sup>Electrical Engineering Department, École de Technologie Supérieure (ETS), Montreal, QC H3C 1K3, Canada

Corresponding author: Muhammad Adnan (m.adnan@nu.edu.pk)

**ABSTRACT** For a utility company to reduce  $CO_2$  emissions along with managing load demands, it must strive for a 100 percent renewable electrical power generation. Europe takes initiatives to achieve this goal by developing a super smart grid (SSG) based on renewable energy resources (RERs) by 2050. The SSG is based on two exclusive alternatives: wide area and decentralized power generation using large number of RERs. Before developing such SSG, there is a need to address the critical issues associated with RERs, i.e., load flow balancing and transients stability. Considering a reliability issue involved with RERs and the random deviations between demand response and generation response patterns, load flow balancing and transient stability become challenging research issues in SSGs. These technical issues are also considered to be more challenging, if an unexpected outage in the form of an occurrence of three phase (L-L-L) faults (TPF) arises in SSGs, due to power quality disturbances. To address this problem, load flow balancing probabilistic modeling is performed in this research paper in order to formulate the complexity of randomness between generation and demand response patterns through transmission network planning (TNP) in the form of a super smart node (SSN) transmission network infrastructure. Moreover, a further optimization in SSN transmission network has been done with the addition of a cooperative control strategies in terms of an integrating vehicle to grid (V2G) technology in SSN transmission network in order to achieve further enhancement in load flow balancing and transient stability in SSGs. Furthermore, as SSGs power infrastructure is based on different clusters, therefore in order to accommodate various clusters for load flow balancing, a continuous spinning reserve (CSRs) probabilistic modeling has also been performed in this paper in terms of its integration in SSN transmission network. Considering above probabilistic analysis, future contingencies are easily predictable, before any kind of disruptive changes arises in a SSGs due an occurrence of a TPF. Moreover, from simulation results as performed in this paper, we can easily verified that our proposed probabilistic algorithm of load flow balancing and transients stability outperforms existing literature work and can also achieved near optimal performance, even for a broad range of variations in load and also in case of an arising of significant power quality disturbances in SSGs due to an occurrence of TPF.

**INDEX TERMS** Super smart grid, renewable energy resources, transmission network planning, continuous spinning reserve.

NOMEN		TNP	Transmission Network Planning.
SSG	Super Smart Grid.	SSN	Super Smart Node.
RERs	Renewable Energy Resources.	V2G	Vehicle to Grid.
TPF	Three Phase Fault.	CSRs	Continuous Spinning Reserve.
		GW	Gigga Watt.
The associate editor coordinating the review of this manuscript and		HVDC	High Voltage Direct Curren.
approving i	t for publication was Ali Raza <sup>(D)</sup> .	HVAC	High Voltage Alternating Current.

This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

RIPG	Renewable Integrated Power Grid.
MG	Microgrid.
KW	Kilo Watt.
DRM	Demand Resource Management.
EVs	Electric Vehicles.
SOC	State of charge.
DERs	Distributed Energy Resources.
PFL	Parametric Feedback Linearization.
CPI	Consensus Proportional Integral.
DESS	Distributed Energy Storage Systems.
ESS	Energy Storage System.
WTT	Wind Turbines Tripping.
WT	Wind Turbine.
PG	Power Grid.
MW	Mega Watt.
SAARC	South Asian Association for Regional
	Cooperation.
VPP	Virtual Power Plants.
CSP	Concentrating Solar Thermal Plants.
FACTS	Flexible AC Transmission System.
DLR	Dynamic Line Rating.
HTS	High Temperature Superconductors.
V2X	Vehicle with Everything.
GWO	Grev Wolf Optimization.
HVDC	High Voltage Direct Current.
IDEA	Improved Differential Evolution algorithm.
PV	Photovoltaic.
PSO	Particle Swarm Optimization.
SSO	Salp Swarm Optimization.
HESS	Hybrid Energy Storage System.
MABO	Modified African Buffalo Optimization.
EMA	Exchange Market Algorithm.
MPPT	Maximum Power Point Tracking.
SOP	Sequential Quadratic Programming.
SIMBO-O	Swine Influenza Model Based Optimization
~~~~ <b>(</b>	with Ouarantine.
SVCs	Shunt VARs Compensators.
DR	Distributed Network.
AI	Artificial Intelligence.
SHF	Shunt Hybrid Filters.
AFNN	Adaptive Fuzzy Neural Network.
ANN	Artificial Neural Network.
NEA	North East Asia.
ASEAN	Association of Southeast Asian Nations.
AAG	Australian–Asian Grid.
UHVAC	Ultra High Voltage Alternating Current.
V2G	Vehicle to Grid.
PEV	Plug in Electric Vehicle.
AMI	Advanced Metering Infrastructure.
MDN	Modern Distribution Network.
DAO	Davahead Optimization.
RTO	Real-time Optimization
FCR	Frequency Containment Reserves.
RFC	RainFlow Counting.
CPDs	Custom Power Devices.
-	

DVR	Dynamic Voltage Restorer.			
VSC	s Voltage Source Converters.			
LCC	Line Commutated Converters.			
BESS	Battery Energy Storage System.			
PLF	Probabilistic Load Flow.			
EHV	Extra High Voltage.			
IoE	Internet of Energy.			
MTHVDC	Multi-terminal High Voltage Direct Current.			
STA-SMC	Super-twisting Algorithm-Sliding Mode			
	Controller.			
VCM	Voltage Control Mode.			
CCM	Current Control Mode.			
EMSs	Energy management systems.			
THD	Total Harmonic Distortion.			
ICPT	Inductively Coupled Power Transfer.			
EMS	Energy Monitoring System.			
CI	Computational Intelligence.			
FRCUC	Frequency Risk-Constrained Unit Commit-			
	ment.			
SEIG	Self-excited Induction Generator.			
PMSG	Permanent Magnet Synchronous Generator.			
SCIM	Squirrel Cage Induction Machine.			
MLC	Multi Level Converter.			
LFAC	Low Frequency Alternating Current.			
K.E	Kinetic Energy.			

## Symbols

n <sub>c</sub>	Normal Operating States of Cluster.
$\mathbf{p}_c$	Power Outage in Cluster.
d <sub>ek</sub>	Delay in Generalized Clusters.
$\mathbf{G}^{f}(t)$	Forecasted Generation.
$D^{f}(t)$	Forecasted Demand.
$D^{a}(t)$	Actual Demand.
$G^{a}(t)$	Actual Generation.
$\mathbf{R}_D(t)$	Randomness in Demand Side of Power System.
$R_G(t)$	Randomness in Supply Side of Power System.
B(t)	Backlogged Demand.
λ	Time Delay.
r(t)	Reserve Capacity.
$P_i(t)$	Actual power.
C(t)	Price Signal.
V(t, E)	Cost function to minimize E with respect to
	time.
М	Mass of an object.
V	Velocity of an object.
D	Distance.
W	Work.
F	Force applied an object.
А	Acceleration produced an object.
u <sub>f</sub>	Final velocity.
u <sub>i</sub>	Initial velocity.
Р	Power of the wind.
$V_a$	Velocity of wind.
ρ	density of air.
S	Area covered by wind.

- $P_r$  Active power of the wind.
- $V_f$  Upstream velocity of the wind.
- $V_i$  Downstream velocity of the wind.
- $w_h$  Betz limit.
- $\omega$  Angular frequency.
- $\delta_d$  duty ratio of converter in d frame.
- $\delta_q$  duty ratio of converter in q frame.
- $E_d$  Grid input voltage in d frame.
- $E_a$  Grid input voltage in q frame.
- $P_{V2G}$  Active power from vehicle to grid.
- $Q_{V2G}$  Reactive power from vehicle to grid.

## I. INTRODUCTION

Nowadays the importance of electrical energy increases in every field of life therefore the conventional grid is unable to fulfill the requirements of the 21st century. To resolve this issue the concept of super and smart grids have been presented that provide better sensing ability, control mechanism and advanced security and communication system, that support our conventional gird [1], [2], [3]. The super grid is considered as that it supplies power in large scale while smart grid provide power in small scale having decentralization system and both of them are alternative of each other. Nonetheless, we contend that the two ideas complement one another and must operate in as to provide a transfer of energy across various locations. As a result, a SSG [1] and [4] are essential as a power infrastructure to transport electrical energy from various centralised and decentralised generating sites dispersed over a large region with the capacity to handle both fluctuating

RER supply and the periodic variations in demand and supply reaction times. Using RERs in the form of various clusters to fulfil the load demands of different areas and to provide them with cheaper power is one of the fundamental characteristics of SSGs [1], [2]. However, inspite of an achieving cost effectiveness as in case of using RERs in SSGs, there are also number of deficiencies involved with the RERs, and the most significant one is the reliability issue needs to be resolved first so that various regions and the power grid stations that serve them can share an effective and sustainable kind of electrical energy [5]. To offer the best load flow distribution across various areas dispersed over a big area of network, SSGs are built on the interconnection between several RERs arranged in clusters. Second, a widespread [4] roll out of RERs presents wholly novel operational and technological difficulties for the present power infrastructures. However one problem is power quality perturbations, which have a negative impact on power network infrastructure and affect load balancing and peak switching stability. These technical issues associated with RERs need to be properly addressed before developing such huge network of RERs in a form of SSGs. Load flow balancing is a challenging research case in SSG due to its congestion and scalability. Additionally, the reliability involves with RERs, a load flow balancing can be considered as a promising future challenge in SSGs. Things become considerably more difficult when a three phase (L-L-L) fault (TPF) caused by a power quality issue results in an unexpected interruption in SSGs.

A common task in the design and management of power systems is contingency analysis (CA). The system operator provides information on stable security of the power system in line with CA outcomes. However, in terms of voltage instability, conventional CA with corrective action schemes is unable to discern between safe operating conditions and possibly dangerous ones. In reality, one of the primary risks that contribute to the unreliability of the power supply is voltage instability. As a result, an enhanced contingency analysis (ECA) is proposed in paper [6], which extends the traditional CA by include constant voltage analysis based on modal analysis. The research showed the benefits that energy storage and distributed generating resources may provide the stability of voltage, where voltage stability is defined as the capacity of an electric power system to maintain a suitable voltage at all of the grid's buses both during routine operation and after being subjected to disturbances is known as voltage stability [6].

N-1 contingency analysis is one of several features included in energy management systems (EMSs), which are used to regulate how the power system operates. In reality, N-1 contingency analysis is an operation that examines the consequences of a single power network component failure on operational conditions for the power system after the single component has been taken out of the system. In general, computed branch flows are compared with the maximum loading of transformer branches and the current carrying of transmission lines while conducting N-1 contingency analysis, and voltage variations of network nodes are compared with the permissible deviations from nominal values. Since the implementation of the aforementioned N-1 criteria is required by the majority of grid codes, contingency analysis is also applied throughout the design and development stages [7].

Electrochemical conversion is the method through which BESS are able to store energy. When BESS are charged, electrical energy is transformed into chemical energy, which is then transformed back into electricity when the BESS is discharged. BESS allow separate control of active and reactive power supplying into the grid since they include both an energy storage unit and a VSC [8]. The feature of dynamic models, renewable energy, and the dependability of power systems are highlighted in term of cascading failure issues that are now posing concerns. Grid forming technology is the approach to research how renewable energy affects cascade failure because of the new issues that result from the high penetration of renewable energy [9].

From technical perspective, the SSG is the key enabler for the massive integration of RERs in the power system as proposed in [10], [11], [12], and [13]. RERs are widely accessible in outlying areas, especially in European nations. For instance, the best wind locations are those that are offshore or near to the coast [14]. These RERs are not reliable and to a certain degree, they have a variable and an unpredictable generation output. Moreover, these variabilities in different RERs poses an entirely new challenges in terms of its integration with one another as proposed in [15]. Load flow balancing and transients stability are noted as one of the promising problem in this case, which was clearly highlighted in [16]. Considering a realistic scenario of SSG, various topologies have already been studied by different organizations as proposed in [17]. Moreover, environmental organizations [18] should also consider the future SSGs development to be necessary in order to reduce CO2 emissions through maximum penetration of RERs in the existing power grid stations. Projects and concepts based on SSGs have gotten favorable coverage from both the political and media spheres. Although, considering a technical community, a great skepticism exists. The main reason behind this is that, The primary cause of this is the lack of technology that is now available and well suitable for the construction of future SSGs. i.e., technical research at their initial phases. Therefore, detailed simulation studies using appropriate RERs models is essential to be carried out for satisfactory integration of large wind farms in a form of SSGs, before its actual implementation in future. The simulation analysis of load flow balancing and transient stability in future SSGs is the foundation for this research area. Moreover, this research study will not only address these issues in terms of theoretical perspectives, but also in terms of technical perspectives in order to elaborate these issues more clearly in future SSGs.

As discussed previous, that SSG is considered to be a future energy project and European countries are the first one to take initiatives to develop such SSGs [4]. Therefore, the problem formulated in this paper is latest and contrast to all previous ones, as described in the literature, it is new. As a result, the proposed approach for load flow balancing and transients stability in SSGs in this research work cannot be directly compared with the approaches presented in previous literature studies, which is purely based on smart grids. SSG, which is based on different RERs.

Clustering is unsupervised classification approaches that used to recognize the patterns of power usage in a network [19], [20], but in this work clusters are extension of smart gird, having specific electric power comes from renewable energy sources to communicate within SSGs. The following are characteristic of the cluster [21]:

- Clusters have to be reliable;
- Cluster should communicate to linked area having denser region of data space;
- The region in data space that communicate to cluster having some characteristic like convex or linear characteristic;
- A minimal set of variables should be enough to describe the clusters.

To overcome this difficulty, we consider single cluster smart grid RERs approaches for load flow balancing and transients stability and then incorporating it in SSG to show its effects on different clusters of RERs. This shows the deficiencies of utilizing previous proposed algorithms for load flow balancing and transients stability based on smart grids (one cluster approach) in SSGs, which is based on an interconnectivity between various clusters. Moreover, it also shows a significance of our proposed probabilistic algorithms, which is suitable for both case scenarios, i.e., smart grids as well as SSGs, as compared to existing research in the literature, which is only limited to smart grids. Considering a probabilistic modeling for load flow balancing and transients stability in case of smart grids (single cluster), a similar kind of proposed methodology was proposed in [22]. To illustrate that SSG is a transformation of both super grid and smart grid, a three zone strategy for SSG was proposed in [9] as shown in Fig. 1. Both systems could be able to develop with the help of the three zone SSG technology plan while preserving a high degree of consistency. The three zones are super grid, smart grid and the transformation pathways between these two technologies.

A smart grid is a two-way, cyber secure communication system that integrates computational intelligence with electrical energy generation, transmission, substations, distribution, and consumption in order to achieve a free, system that is neat, secure, stable, efficient, and sustainable [23] while Super smart grids are a kind of wide area transmission system that primarily employs renewable energy sources, lowering greenhouse gas emissions while also maintaining the electrical infrastructure of multiple countries. The SSGs allow for two-way communication between sources and loads in numerous countries, and these loads may typically be serviced by a variety of RERs connected to grids [24].

A smart grid may incorporate new energy sources into the system and address intermittent power supply issues, ensuring stability independent of working conditions [25]. The term "smart grid" refers to a system that combines advanced technology, modern sensors, and control techniques [26]. If a need cannot be met by the smart grid environment, the super smart grids (SSGs) is an independent entity that ensures the balance condition between the supply and demand. In other words, it reduces the burden and focuses on a secure, dependable, and sustainable power supply [27]. SSGs are a hybrid system, which combines two or more energy sources. It is possible for this system to combine electric power from different sources with renewable energy sources, either on its own or in conjunction with the recent grid system [28]. Fig. (1) shows relationship between these 3 zones. The zone can be defined as the power network that split into different regions in order to comply with various planning and operating rules for the power system, these regions is known as zones [29].

The methods of transformation between super and smart grids will be effectively eliminated by performing future experiments in the form of SSG in order to developed SSGs infrastructure. At this stage, as super grid and smart grid are considered to be a matured technology, therefore options for SSGs development are widely open. We envision a SSG that plans to resolve the problem of exceeding capacity, by fusing



Excess Supply Capacity

FIGURE 1. A zoning strategy for development of SSGs [13].

the benefits of developed super grid and smart grid technologies, which will most probably be the outcome of zoning.

Now, before discussing the strategies and techniques, that are being used and developed for smart grid load flow balancing and transient stability and then incorporating these in SSGs, the TNP for a SSGs should be properly analyzed in order to address the critical issue of an inter connectivity between various clusters of different regions. To address the variability of RERs for different clusters in SSGs, the TNP will might need an extra levels of fast ramping spinning reserves in order to handle the critical situations that lies in power systems which is caused by unreliable property of the RERs. The severity of the situation is increased by the presence of various contingencies, which make the relationship among deployment and the prime reserve siting, the quantity of load curtailment in the event of using an unreliable RER, the construction of new lines, and the choice and placement of the ideal renewable sites to be established even more difficult. Therefore, in order to reduce these contingencies issues in SSGs, a proactive approach for TNP in SSGs has currently noted an attention of many scientific researchers and power regulating authorities as proposed in [30].

Considering proactive technique, the transmission network planners anticipate the optimum decisions about the generation investments. Moreover, it also helps the transmission network planners in a way, that they may be able to induce generation and transmission expansions with a significant potential benefit to both utilities and customers. Several research works have shown the significance of using a proactive TNP approach as proposed in [31], [32], and [33]. Proactive TNP is a kind of co-optimization technique, that is especially useful in case of large transmission network investments in order to interconnect different regional areas in a form of clusters, i.e., SSGs, with having a large amount of renewable generation potential and with corresponding curtail load demands. These co-optimization techniques for generation and transmission development have been proposed in several research works and technical reports [34], [35], [36], [37]. Considering co-optimization scenario of TNP, it is easily possible for power system operators to easily gain high renewable penetration goals [21], while considering the complex relationship between the new renewable sites with the existing one and their candidate transmission lines. Moreover, the potential of different reserve levels and considering the functioning of power systems in a reliable condition in the existence of huge amount of RERs is also considered to be a timely topic. Clearly, the integration of large amount of RERs as in case of SSGs vary the pattern of power systems operations, and so, it greatly effects the TNP and the minimum reserve sitting in case of SSGs. Furthermore, these RERs are often placed far from load demand network and traditional spinning reservoirs, in a variety of clusters. Therefore, proper TNP commonly impact on the capacity of the power systems to sure the reserve vulnerability, whose significance has also mentioned in [38] and [39].

A V2G can be able to control the power system during disruptions in the quality of the power and quickly stabilize the whole power system. In case of fault in SSGs, V2G provides a high quality of power with a minimum time because it includes a battery management system that gives faster energy with low value of transient and balancing the flow of power [40].

The V2G technology implies that the interaction of electric vehicle with grid. If grid is unable to fulfill the requirements of the load, then energy flow from vehicle to grid while in case of excess of energy to the grid then power flow from grid to electric vehicle [41]. For real time communication it must be necessary to involve the status of charge and demand of load for better operation of V2G mode [42]. Therefore, the primary activity in vehicle-to-grid really consists of the two-way exchange of information and energy between EVs and the grid [43]. The vehicle to grid system must guarantee that an adequate number of EVs are available as energy storage resources within a certain amount of time in order to fulfill the needs of power load management. The specific number of EVs must be combined through an aggregator, and track the associated data of the EVs, such as the position of the EVs, the state of charge of the batteries, the planned departure time, and the real-time battery capacity chargeable and dischargeable, etc. so that the control center can effectively plan the load requirements of the grid based on the EVs charge and discharge. The aggregators can be ordered to deliver vehicleto-grid services by the smart grid control center in order to give the smart grid feedback-related data.

Simultaneously aggregators provide the gathered information to the control center. The control center may now track the user's locations and examine the users private information if the original location data has been provided [44].

Different options are available for transmission network upgrade using TNP proactive approach in SSGs. One possibility is using local reinforcements to strengthen the network. Large power transfers need greater transmission capacity, hence it is important that these reinforcements take place throughout the chosen transmission channel at numerous places. Installing direct transmission lines close to important generating and demand site in the shape of a grid, sometimes known as a "Super grid," as suggested in [45], is another potential proactive strategy of TNP. HVDC or HVAC technology may be used to build such transmission lines. In this research work, an equalized power sharing between different power grid stations of their corresponding clusters is proposed using both TNP proactive approach, i.e., local reinforcement and direct transmission line, which provides load flow balancing and resolves the future contingencies issues in SSGs. Local reinforcement approach is utilized inside a cluster to strengthen the transmission inter connectivity of different RERs, whereas direct transmission line approach is utilized for cluster to cluster interconnection in order to supply an effective load flow sharing between different clusters. By providing an equalize power sharing among various clusters of SSGs through TNP, it guarantees that not every cluster's component will experience an early energy shortage, so the full power capacity of all clusters in SSGs is available to regulate each other.

Equalize power sharing between different power grid stations and their corresponding clusters in SSGs is technically and economically preferred due to the following reasons as proposed in [46]:

- 1) Elimination of the circulating currents among transmission lines and relief overloading conditions of different power grid stations of various clusters in SSGs.
- Better utilization of the local energy resources in SSGs at the receiving side of different power grid stations,
- Equal revenue distribution among investors and owners of various clusters and local energy resources located at different regions of SSGs.

Now, in order to address various strategies and methodologies that have already been studied for load flow balancing and transient stability in case of smart grids, many proposals have already been discussed in [47], [48], [49], [50], [51], and [52]. The techniques related to load flow balancing can be further categorized into different sections. These approaches optimize an issue of load flow balancing with the help of a storage and flexible loads in a RIPG as proposed in [28]. Considering a dynamic optimal framework for flow of power for multiple time periods to represent load curtailment issues of distributed generation resources based on renewable, an optimum solution based on an energy storage systems and flexible demands was already discussed in [29]. Similarly, in case of load flow balancing, another approach is proposed in [30] to balance the power generation of the RERs in order to maximize or minimize the usage of power in the microgrid (MG) through controlling a stored energy. Recently, different probabilistic approaches are also studied in the literature by considering issue of uncertainty, which exist with the RERs. It includes designing of a probabilistic and also an optimum power flow algorithms in order to mitigate an issue of uncertainties, which occurred due to an effect of an environmental impact on generation resources based on PV and wind farms as proposed in [50], [51], and [52].

Now, in order to incorporate these techniques in SSGs for load flow balancing to show its effects on different clusters, we must first clarify a technical question, that is it suitable to incorporate these smart grids load flow balancing approaches in SSGs By answering this question, we must first consider the common drawback involves in above listed techniques. The main limitation in proposed approaches is that, all of these load flow balancing techniques required flexible loads, control mechanisms and also sophisticated storage by concerning optimum flow of energy to the load receiving station of different power grid stations. Additionally, these recommended techniques are thought to be computationally demanding, and the best load flow balancing solution is not always assured, if we considered a larger deviation between loads on the receiving station (KW) of different power grid stations (MW), which is the main reason behind developing a future SSG infrastructure. In order to tackle this problem, in this article, a new approach to solving this issue is presented; a super smart node (SSN) transmission network topology using TNP proactive approach, that interconnects various RERs with one another in an intelligent way for enabling load flow balancing in SSGs while taking variable heavy loads into consideration. This is believed to be the first piece of work that discusses the process of developing a specialised SSN, and to the greatest of our understanding, in case to provide synchronous stability to the power system in terms of load flow balancing on the receiving side of different power grid stations, even in case of an occurrence of TPF in SSGs.

The SSN is basically a transmission network technique that emerge various transmission networks in an intelligent way, so that these transmission networks must be able to takes into account different uncertainties, that happen in power system network caused by variations in loads and RERs, especially TPF event in SSGs and improvised the balancing of the receiving side load flow, subjected to different potential vulnerabilities in SSGs. It contains both types of nodes, IT node is responsible for flow of information while electrical node is responsible for flow of electricity. The Fig. 2 shows the combining working of both electrical and IT nodes via different entities [53].

Moreover, this SSN transmission network topology for load flow balancing should be effectively utilized in both smart grids as proposed in [22] and SSGs power infrastructure as compared to previous proposed approaches in literature for load flow balancing, which is only limited to smart grids.

Similarly due to congestion issues in SSGs, further variations in terms of load flow arises on the receiving side of different power grid station, especially considering an

## **IEEE***Access*



FIGURE 2. The combine function electrical and IT nodes in SSGs system.

occurrence of TPF with corresponding larger delay, which causes the tripping of different wind turbines nearest to the fault location. Therefore, the receiving side of various power grid stations in various clusters may easily be improved further in terms of load flow management and transient stability, by providing a cooperative control strategy in terms of an integrating V2G technology in SSN transmission network of SSGs.

There are different cooperative control techniques available, however, a major of them can be divide into three different group. The first technique is based on a centralized control scheme by utilizing a central controller in power system which analysis all energy storage systems as proposed in [54] and [55]. As suggested in [56], this sort of approach involves communication connections between the centralized unit and each and every part of the power systems in order to gather data on a global scale and analyze it before sending it to the controller. Thus, this technique is not a cost effective option and it is also vulnerable to a single point failures, due to which it cannot be utilized in SSGs, as single point failure

causes a destructive effect in a form of cascading failures in SSGs. The second technique is fully based on a decentralized control scheme, which consider local information only, like droop control as mentioned in [57]. This type of cooperative control technique is less expensive and it is also robust when communication medium is not essential. Although inspite of this advantage, there is still an issue of an absence of a vast amount of an available information. Therefore this technique may not be effective at all, especially if we intend to use every resource in the electricity network as effectively as possible, as suggested in [58]. The third method relies on a distributed control strategy and merely calls for information sharing between power system components over a local communication network. In order to make the power infrastructure of an SSG economically viable, these administrative and control solutions must be effective for the SSG's power infrastructure due to the potential diversity and distribution of its components. According to the theory put forth in [56], distributed control strategies that have been optimized and well designed are flexible, efficient, scalable, and inexpensive to implement. For these reasons, these techniques are seen as a promising alternative for offering the best control for load flow balancing in SSGs. In order to manage the charge of the battery packs linked to it and use the power capability to handle various clusters of SSGs, especially in those case scenarios where a power quality disturbance issue occurs in SSGs due to an occurrence of TPF, we provide such a distributed supportive control schemes using V2G.

The SSN transmission network in our case scenario will only improvise the SSGs for load flow balancing, whereas, V2G is regarded as a useful technology, both for load flow balancing and transients stability in SSGs. Moreover, considering from literature point of view, a V2G importance for DRM in smart grid was already proposed in many literature studies, but it's utilization for load flow balancing and transients stability analysis in case of SSGs is not discussed in literature. Before discussing, that why V2G techniques as already proposed in literature for smart grids is not suitable for SSGs, as compared to our probabilistic V2G proposed approach, which is suitable for both smart and SSGs power infrastructure, some of these V2G techniques for DRM should be properly addressed.

For DRM employing V2G in the context of smart grids, numerous approaches and algorithms have previously been addressed. These DRM methods may also be decomposed into other groups. For instance, in [59] suggested an optimization technique based on DRM to stabilise an entire power network employing PV inverters and EV chargers. In [60] and [61], researcher suggested to use an implementation of renewable energy resources relying on PV and wind turbines to address the issue that EVs experience due to their inconsistent charging and discharging behavior. In [62], it has been observed that the load profile may be shaped using DRM to balance both demand and supply of energy. The author offered a solution in [63] for scheduling electricity generation in accordance with engagement plans created by the consumers using DRM. Moreover, a game theoretic strategy incorporating DRM was used in [64], [65], and [66], to take into account the advantages of both consumers and utility. Similarly, in order to control the household appliances energy utilization through shaping of a load profile and also to reduce an electrical energy cost in considerations of residential grid, the scholars resolved this issue in intelligent way using DRM in [67] and [68]. Moreover, due to technological advancement in terms of the Internet of vehicle [69], [70], [71] and vehicular ccloud [72], [73], [74], Real time flexible DRM is made possible for EVs via V2G technology. In [75], it has proposed to investigate how EV mobility can affect DRM in V2G systems in smart grids.

Moreover, in DRM using EVs technology, various approaches have been considered for charging and discharging mechanism of EVs. In [76], the author proposed a solution in terms of analyzing the DRM performance during an off peak charging of an EVs. Employment of dynamic pricing in order to control the EVs users charging and discharging behaviors using DRM was proposed in [77]. Moreover, satisfying EV charging demand from electricity generation due to wind power and photo voltaic power using DRM was proposed in [78] and [79]. Analyzing the control of V2G on battery SOC has presented in [80] in order to obtain regulation of frequency for power grid stations and EV charging using DRM techniques. Additionally, to enhance the performance of DRM in smart grids, many topologies have been considered in literatures, i.e., the profit of EVs users in terms of DRM was proposed iin [81]. Analyzing the effect of DRM on various costs of power grid stations was proposed in [57] and [82]. Moreover, the concept of fairness, in terms of power delivery by utility and power consumption by customers using DRM schedule was also proposed in [83]. Considering all of these techniques, power in the region grids will be addressed by V2G technology, which have a significant contribution in the ancillary services sector, i.e., one cluster (smart grid) constant stabilization required to balance supply and demand. whereas, addressing of V2G technique for load flow balancing in multiple regional power grid stations in a form of SSGs is absent in literature. Therefore, in order to make these V2G techniques in smart girds to be effectively utilized for load flow balancing in SSGs. There is a need to redesign a transmission network for one cluster and transformed it into for various clusters in order to provide an effective load flow sharing between different clusters. This can be done through TNP proactive approach, i.e., developing a SSN transmission network topology as previously discussed. According to the researcher opinion, the proposed work is a new that explored the issue of load flow balancing using V2G cooperative control strategies in SSGs, for balancing the power system on the receiving side of different clusters, even in case of an occurrence of TPF with corresponding larger delay in SSGs.

Similarly, V2G is also considered to be a valuable resource for transient stability analysis in smart grids, i.e., upto one clusters. However, its utilization in terms of transient stability in case of SSGs, i.e., cluster to cluster interconnections is not discussed in literature. Various strategies have been proposed for transient stability analysis using V2G in case of smart grids in [84], [85], [86], [87], [88], [89], and [90]. These techniques can be further divided into different categories. Transient stability analysis in terms of reactive power compensation in smart grids using V2G was proposed in [64]. Active power regulation in terms of load flow balancing by valley filling technique using V2G was proposed in [65]. Moreover, transient stability analysis for current harmonic filtering; in order to supply ancillary facility, by considering frequency control using V2G technology was proposed in [66], [67], [68], and [69]. Similarly, enhancing grid efficiency for maintaining stability in power system using V2G was proposed in [70].

Nowadays, incorporating EVs into power grids allows them to serve as energy storage and power generating units in addition to serving as charging loads [91]. V2G may also be regarded as a useful resource to engage in grid ancillary services to preserve the safety and reliability of power grids, like frequency control [92], voltage management [93], spinning reserve [94], and load peak changing [43]. Nevertheless, one EV cannot engage in these grid ancillary services because to its limited capacity. In order to regulate a large number of EVs and offer grid ancillary services, the model of a V2G aggregator has presented in [95]. All of these techniques provide an effective solution in terms of transient stability analysis using V2G in case of smart grids. However, the utilization of V2G for transient stability analysis due to power quality disturbances in case of an occurrence of TPF in SSGs is not discussed in literature. As, in case of SSGs, due to inter-connection between different RERs clusters and considering the reliability issues involve with RERs, transients stability should be consider to be a challenging research issue in SSGs.

Moreover, inspite of utilizing distributed based V2G cooperative control strategies in SSGs to enhance transients stability of the power system. we can also utilize other distributed controller based methods to control many DERs in SSGs in order to enhance transients stability of the power systems. However, a major drawback of utilizing these distributed controller based approach in SSGs is that, it will adds a significant delay in a power systems. Before discussing certain drawbacks involves in case of utilizing other distributed controller based approaches in SSGs, some of these controller based techniques to enhance transients stability in power system should be properly addressed.

Recently, several distributed controller based techniques in Fig. 3 have also been employed in a control loop for the enhancement of the transient stability of power systems. A transient stability enhancement in terms of distributed controller by using flocking theory was proposed in [96]. Although, the results obtained from simulation is represented in [77] show that controller operations may be steady as transients delay large due to the power quality issues in power

VOLUME 11, 2023

systems.Farraj et al. [97] operate rapidly energy storage devices to swiftly balancing the grid using a PFL approach inside a distributed controller architecture. The flocking based controller in [96] and the CPI controller as described in [98], [99], [100], [101], and [102] have mathematically compared to the suggested PFL controller. The PFL controller gave a more significant result as compared to flocking and CPI controller in terms of transients stability enhancement by reducing the corresponding delay issues in power systems. The PFL controller's flaw is that, as a consequence of using Kron's technique, which was suggested in [103], for system reduction, it implied that all generating angles and voltages were known, which was essentially incompatible with the idea of a distributed method. Therefore, a distributed nonlinear robust controller has suggested in [104], which resolves the above issue in PFL controller and achieve the transient stability enhancement of power system within around 3s with the delay in a power system corresponds to 400ms. The suggested controller in [78] and the PFL controller both accomplished identical performances, when the time of delay is 50ms in case of transients, while for other cases it fails. All of these proposed distributed based controllers approaches provides an effective approach in terms of enhancing transient stability in power systems, but considering the delay up to small scale, i.e., 50ms - 400ms. Therefore, there is a need to operate the fast acting energy storage systems in such a way, that they can compensate for time delays due to transients issues and input, while the addition uncertain load along with varying time disturbance also considered, such as in case of SSGs. Our approach is that, to manage different fast acting DESS in SSGs, new distributed controller-based methods have been developed, we utilized the concept of distributed cooperative control strategies using only one ESS, i.e., V2G. V2G employs an energy storage device that reacts swiftly to stabilize the power grid in a small interval amount of time, due to the occurrence of TPF, even with the improvement of appropriate delay in parameters of transient conditions in a power system. The simulation results have been created based on data of observation and in terms of comparison analysis between our propose V2G distributed cooperative control approach with the previous distributed based proposed controller approach in [85] in order to show effectiveness of V2G cooperative control strategies in SSGs. Moreover, considering the phenomena of TPF occur in SSGs, a similar kind of proposed technique for stabilizing of load flow and analysis of stability in term of switching peak in SSGs with the help of V2G technology has discussed in [22].

Moreover, in this research work, we studied the factors that affect the quality of power under disturbances, i.e., transients on SSGs that occurred due to transients stability issues, which arise in power system caused by TPF that happen in SSGs. Although transients stability is a term, that is used especially for an angle stability in power system literature, as in case of an occurrence of larger disturbances in power systems, whereas, power quality disturbances is a term that

is especially reserved for voltage sag and swell, transients, interruption, voltage fluctuations, waveform distortion, and variations in frequency, but the relationship between these two terminologies is very effective, means both of these terminologies are interlinked with one another as proposed in [105] and [15]. Transient stability can be elaborated in the power system to cater the synchronous stability in power system network, when it is tested in the existence of unbounded disturbance, i.e., as in our case phenomena of TPF event with corresponding larger delay in SSGs. These disturbances have a severe impact on power system network and it can also cause a enough abnormality in wind turbine generator rotor angles in a form of large excursion, which observe a enough variations in the speed and position of the rotor. as a consequence of these variations, a serious power quality problem, i.e., transients arise in SSGs. The oscillations caused by transient stability problems in a power system network, as well as in the event of the occurrence of many kinds of faults that occurs in a power system network, are regarded to be a serious problem for power system stability, and they should be managed as such. These transient stability oscillations are explored in this article from the perspective of power quality. It is argued in this research that transient stability fluctuations in a power system network may cause the development of serious power quality perturbations, or transients in power systems, as in the case of SSGs power infrastructure. This study describes the transient stability problems in power systems proposing both numerical and conceptual studies, which define a number of power quality phenomena that may be driven on by the oscillations in transient stability as suggested in [87], and [15]. Contrary to the widely held empirical presumption that these both issues (stability of the power system along with production of quality power) are unrelated to one another, the simulation results from [87], and [15] clearly demonstrate that transients stability oscillation caused by an TPF that happen in a power system infrastructure in fact it can a vital role for the arises a disturbance which led to power quality issue, i.e., transients that occur in the power grid stations at receiving side.

As V2G provides an effective solution for the analysis of both load flow and transient stability, but up to its small scale utilization, i.e., as we don't have an enough EVs available in order to provide large scale backup to multi power grid stations of various clusters in SSGs. Therefore, to study the problem of larger power outages in SSGs, due to an occurrence of TPF with corresponding larger delay, which causes a tripping of wind turbines nearest to fault locations, a noncooperative control based probabilistic model of CSRs has also been formulated in SSGs in terms of its integrations in SSN transmission network. CSRs non-cooperative control based probabilistic model mainly optimizes the SSGs for the balancing of load flow but the effect of transients due to an occurrence of TPF in SSGs is still there. Therefore, the above simulation results between V2G distributed based cooperative control and the proposed distributed controller based approach in [85] for transients stability is further

verified in terms of comparison analysis of probabilistic modeling of V2G cooperate control based technique and CSRs non-cooperative control based technique in order to show effectiveness of V2G in terms of compensating transient stability issues in SSGs. The difference between cooperative and non-cooperative control was clearly formulated in [36] and [106]. In the case of cooperative control in a power system, each subsystem is conscious of the dynamic model of the other subsystem in order to forecast future choices more precisely, particularly in the event of concerns with power quality disruptions that happen in power systems as a consequence of the occurrence of various types of faults. Our V2G based cooperative control probabilistic model for SSGs power infrastructure utilizes the same approach for effectively reducing the issue of transients stability under the event of TPF in SSGs. Whereas, for non-cooperative control based approach the transmission of information in order to communicate between different subsystems is limited. Therefore, compromises on an accuracy has been made in case of predicting an accurate future decisions for power systems. This shows the effectiveness of cooperative control based V2G approach as compared to CSRs non-cooperative control based approach for reducing future contingencies issues in a form of load flow balancing and transients stabilities in SSGs. The proposed work that discusses the evaluation based on load flow balancing probabilistic modelling in order to describe the complex nature of randomness between generation and demand response behavior through an SSN, CSRs, and V2G, and transients stability probabilistic modeling through only V2G in SSGs, even under consideration of TPF. The novelty of proposed work is mentioned in below points:

- A qualitative overview of major enabling technologies for smart grids is explored that used in SSGs. Additionally, the challenging tasks related to the implementation of smart grid technologies in SSGs have been explained for its smooth operation.
- 2) The TNP proactive method is used to design a customized SSN transmission network, compensating for any interruptions in the form of TPF with correspondingly smaller delays as load flow balancing on the receiving side of various power grid stations in SSGs.
- 3) The concept of cooperative V2G control technique along with other non-cooperative techniques can be implemented in SSN transmission in case of TPF event with corresponding larger delay due to which causes tripping of wind turbine that linked with cluster near the fault location. As a result, SSGs will be optimized for load flow distribution between different clusters, and transient stability problems also be reduced.
- 4) A probabilistic model is used to express the complexity of randomness in terms of load flow balancing in the SSN transmission network, CSRs, and V2G, as well as the analysis of transient stability via V2G topology.

5) In order to demonstrate the effectiveness of the V2G cooperative controller-based approach in order to reduce the transient stability issues in SSGs with the corresponding improvement in time delay due to an occurrence of TPF in SSGs, a comparative analysis between V2G cooperative control-based technique, CSRs non-cooperative controller-based technique, and distributed non-linear robust controller-based technique have been carried out in [104].

The rest of the article is split into the sections that follow: The methodological portion is mentioned in Section II. Section III includes the result generated from the simulation for comparison purpose of the proposed technique with current load flow balancing and transient stability models. Section IV provides a brief survey about countries moving towards developing future SSGs. Section V illustrates about the risks and open questions raises in case of development of future SSGs. Section VI gives a brief overview of technological uncertainties during development of future SSGs. Section VII provides an overview of an infrastructure of European SSGs. Finally, Sections VIII, IX and X include the conclusion and future work.

#### **II. METHODOLOGY**

The suggested technique is categorized into two sections. Initially, the level of power system vulnerability is assessed in the event of a TPF in SSGs, which are made up of four clusters and twelve RERs. The study of an uncertainty approach depends on three phase (L-L-L) faults delay is presented to distinguish the phenomena of occurrence of TPF with associated smaller or greater delay in a power system in order to research power system vulnerability assessment. Furthermore, challenges with power quality caused by the occurrence of TPF in SSGs and irregularity of load flow caused by specific RERs tripping are also taken into account. A power system may be vulnerable to TPF if the delay is either smaller or bigger, i.e., fault occurrence interval time, then Algorithm 1 or Algorithm 2 uses the SSN, CSRs and V2G scenario to use probabilistic modelling to examine and anticipate the predicted irregularity in a power system.

We considered the generic case because the concept of SSGs has not simulated and implemented yet in real life scenario. It will be implemented in particle life in future 2050 [4]. Therefore, we consider these two duration for the delay as per choice to implement the algorithm 1 and algorithm 2 for 0.01s and 0.1s respectively. The flow chart for implementing of algorithm 1 and 2 are represented in Figs. (4) and (5) respectively.

Algorithm 1 uses SSN solution to detect and compensate smaller delay that occurred in the form of TPF, with no tripping of wind turbines. Whereas Algorithm 2 uses SSN, CSRs and V2G as a solution to detect and compensate larger delay caused by TPF, that occurred. i.e., tripping of wind turbines of cluster 1.

Secondly, Algorithm 2 is used in order to provide transients stability enhancement in SSGs with the help of probabilistic modeling, such as an occurrence of TPF in cluster 1 with corresponding larger delay, a choice on the switching of V2G on the receiving side of cluster 3 is executed to compensate transients stability issues in cluster 3 of SSGs. SSN are functioned at the receiving end in both cases, i.e., occurrence of TPF with equivalently shorter or longer intervals, which causes a different variation in  $d_{ek}$  parameter, i.e., greater than predefined threshold of 0.01s, which make the potential vulnerability that depends on the event of TPF with corresponding shorter or longer delay in power systems. For TPF with smaller delay, which doesn't cause the tripping of wind turbines as mention in Algorithm 1, i.e.,  $d_{e1} = d_{e1} + d_{e1}$  $d_{e1+0.01s}, d_{e2} = d_{e2} + d_{e2+0.01s}, d_{e3} = d_{e3} + d_{e3+0.01s}$  and  $d_{e4} = d_{e4} + d_{e4+0.01s}$  of corresponding four clusters, only SSN is operated. Whereas, if delay parameter  $d_{ek}$  is greater than approximate range of threshold, i.e., 0.01s, considering a larger delay due to an occurrence of TPF in cluster 1 as mention in Algorithm 2, i.e.,  $d_{e1} = d_{e1} + d_{e1+0.1s}$ ,  $d_{e2} = d_{e2} + d_{e2+0.1s}, d_{e3} = d_{e3} + d_{e3+0.1s}$  and  $d_{e4} =$  $d_{e4} + d_{e4+0.1s}$ , it initiates the next transition stage tcj + 1. Furthermore, in order to fully optimize the SSG for load flow balancing and transient stability, a choice on the switching of CSR 1, CSR 2, V2G, and CSR 3 in the SSN transmission network is required. CSR 1, CSR 2 and CSR 3 only optimizes SSG for load flow balancing, i.e., compensating power drops issues of cluster 1  $(n_{c1} - p_{c1})$ , cluster 2  $(n_{c2} - p_{c2})$  and cluster 3  $(n_{c3} - p_{c3})$ , whereas V2G is considered to be a precious resource, both for optimizing load flow stability and reducing transients stability problems in SSGs, i.e., compensating cluster 3 power drop issues  $(n_{c3} - p_{c3})$  as well as compensating the transients delay  $d_{e3} = d_{e3} + d_{e3+0.1s}$ , which arises due to transients stability task under consideration of an occurrence of TPF with corresponding larger delay in SSGs.

The closed loop control system demand response probabilistic model for optimizing the stability of both load flow along with enhancement of transient stability in SSGs is shown in Fig. 3. With a closed loop control system power network in the existence of an SSN, CSRs, and V2G as illustrated in Fig. 3, the constantly monitoring have been readily conducted in SSGs for balancing of load flow analysis and transients stability improvement. By considering the probabilistic model which is closed-loop represented in Fig. 3, power system operators can quickly consider the necessary corrective action to improve transient stability and offer the best load flow balance, particularly when TPF occurs in SSGs. When TPF occurs in SSGs with a smaller corresponding delay, SSN effectively enhanced the power system for the balancing analysis of load flow. However, when TPF occurs in SSGs with a larger corresponding delay, which trips the wind turbines closest to the fault location, switching of an already integrated CSR and V2G in SSN transmission network is required to completely optimize the power system for load flow balancing and transients stability. The concepts



FIGURE 3. Demand response schematic model for SSGs.

implemented in Fig. 3 are explained in further description in the next part of the paper.

## A. SUPER SMART NODE TRANSMISSION NETWORK PLANNING PROBABILISTIC MODELING

The optimal generations from RERs will be determined using the suggested RERs probabilistic TNP model of the SSN transmission infrastructure, taking contingencies evaluation in SSGs into account, as well as power quality issues cause by TPF that occurred in SSGs, while also ensuring spinning reserve deliverability. Considering these power quality disturbances due to an occurrence of TPF in a synchronized complex network of four clusters with every one which consist of three RERs in a form of SSGs, a load flow balancing problem using TNP proactive approach is investigated by using the concept of SSN transmission network. Through designing of SSN transmission network by using TNP proactive approach, we can actually adjust our forecasted supply, i.e.,  $G^{f}(t)$  in the context of our forecasted demands, i.e.,  $D^{f}(t)$ reducing the needs and expectations in SSGs i.e., F(t) as represented in Fig. 3. For achieving this objective, a transmission grid closed loop control system model is formed which depicts in Fig. 3, by considering a larger penetration in:

- 1) Demand Response, and
- 2) Power quality disturbances.

A similar kind of proposed work in order to provide sustainability of an elastic demands in case of smart grids was clearly formulated in [107]. Moreover, in our case the penetrations in demand response patterns occurred caused by the tripping of the RERs with an occurrence of TPF with corresponding

#### Algorithm 1

Input : A group of normal operating states of different clusters in SSG

 $(n_{c_1}, n_{c_2}, \ldots, n_{c_n}),$ 

 $\backslash \backslash n_{c_1}, n_{c_2}, n_{c_3}, n_{c_4}$ , corresponds to normal operating states of cluster 1, 2, 3 and 4, while  $n_{c_n}$  represents the normal operating states in case of generalized clusters. **Input** : An unexpected phenomena of three phase

- **(**L L L) fault ( $f_{c1}$ ) in cluster 1 with corresponding smaller delay, which doesn't causes wind turbines tripping (*wtt*) in cluster 1 of SSG
- **Input** : A set of delay in all clusters cause by an occurrence of TPF in cluster 1 of SSG  $(d_{e_1}, d_{e_2}, \dots, d_{e_k})$

 $\backslash \langle d_{e_1}, d_{e_2}, d_{e_3}, \tilde{d}_{e_4}$ , corresponds to delay in cluster 1, 2, 3 and 4, while dek represents the delay in case of

generalized clusters

**Output:** A set of the following transition states for all clusters in order to make up delays caused by the occurrence of TPF at cluster 1  $(t_{c_i} \rightarrow t_{c_{i+1}})$ 

while  $(f_1)$  do

assign next transition state to compensate for delay due to an occurrence of three phase (L-L-L) fault  $\ Fault f_{c_1}$  occurred at cluster 1, which causes a

delay  $d_{e_1}$ ,  $d_{e_2}$ ,  $d_{e_3}$  and  $d_{e_4}$  in corresponding 4 clusters

**if** Cluster  $1 \rightarrow f_{c_1} \rightarrow d_{e_1} + d_{e_{1+0.01s}}$  **then** | Next transition,  $t_{c_{j+1}} \rightarrow$  Super Smart Node \\ Fault  $f_{c_1}$  at cluster 1 causes a delay  $d_{e_1}$  in clusters 1

```
else if Cluster 2 \rightarrow f_{c_1} \rightarrow d_{e_2} + d_{e_{2+0.01s}} then
Next transition, t_{c_{j+1}} \rightarrow Super Smart Node
\setminus Fault f_{c_1} at cluster 1 causes a delay d_{e_2} in
clusters 2
```

else if Cluster  $3 \rightarrow f_{c_1} \rightarrow d_{e_3} + d_{e_{3+0.01s}}$  then Next transition,  $t_{c_{j+1}} \rightarrow$  Super Smart Node  $\setminus$  Fault  $f_{c_1}$  at cluster 1 causes a delay  $d_{e_3}$  in clusters 3

else if Cluster  $4 \rightarrow f_{c_1} \rightarrow d_{e_4} + d_{e_{4+0.01s}}$  then Next transition,  $t_{c_{j+1}} \rightarrow$  Super Smart Node  $\setminus$  Fault  $f_{c_1}$  at cluster 1 causes a delay  $d_{e_4}$  in clusters 4

else if

Cluster 1 + Cluster 2 + Cluster 3 + Cluster 4  $\rightarrow$   $f_{c_1} \rightarrow d_{e_1} + d_{e_{1+0.01s}} \rightarrow d_{e_2} + d_{e_{2+0.01s}} \rightarrow d_{e_3} + d_{e_{3+0.01s}} \rightarrow d_{e_4} + d_{e_{4+0.01s}}$  then Next transition,  $t_{c_{j+1}} \rightarrow$  Super Smart Node  $\setminus$  Fault  $f_{c_1}$  at cluster 1 combine effects on all clusters else

 $\begin{vmatrix} next transition, t_{c_{j+1}} \rightarrow (n_{c_1} \rightarrow n_{c_n}) \\ All clusters operating in its normal states$ **end** $\\ Send t_{c_{j+1}} transition state as an input to Algorithm$  $1 \\ end \end{vmatrix}$ 



FIGURE 4. Flow chart of Algorithm 1.

larger delay in SSG. Power quality disturbances got more attention in case of an occurrence of TPF with corresponding larger delay, which causes a significant transient stability issue in SSGs. Therefore, in order to analyzed these two future contingencies issues in SSGs, we consider a scenario of  $D^{f}(t)$  and  $G^{f}(t)$ . Additionally, to support load flow balancing

across distinct RERs of four clusters in SSGs, the  $G^{f}(t)$  must be synchronised with  $D^{f}(t)$  in order to meet the criteria for synchronous consistency between the generation and demand response behavior, i.e.,

$$G^{f}(t) = D^{f}(t) + r_{0}$$
 (1)



FIGURE 5. Flow chart of Algorithm 2.

where,  $r_0$  denotes the nominal reserve, i.e., supply from the SSN transmission network, in order to achieve synchronization between  $G^f(t)$  and  $D^f(t)$ . In this case scenario, we considered an effect of  $G^{f}(t)$  and  $D^{f}(t)$  response patterns by regularly providing the returning demand B(t), in terms of adjusting  $r_0$  through SSN transmission network, in order

## Algorithm 2

1

I

1

1

1901	1000 Z
nput	: A set of normal operating states of different clusters in SSG
	$(n_{c_1}, n_{c_2}, \ldots, n_{c_n}),$
input	: An unexpected occurrence of three phase (L-L-L) fault $(f_{c_1})$ in
	cluster 1 with corresponding larger delay, which causes wind
	turbines tripping (wtt) in cluster 1 of SSG
nput	: A set of power outage in all clusters due to tripping of wind
-	turbines in cluster 1 of SSG $(p_{c_1}, p_{c_2}, \ldots, p_{c_k})$
$\langle p_c \rangle$	$p_{c_2}, p_{c_3}, p_{c_4}$ , corresponds to power outage in cluster 1, 2, 3 and 4,
while	$p_{\rm ex}$ represents the power outage in case of generalized clusters
nnut	• A set of delay in terms of transients in all clusters due to
nput	$\alpha$ occurrence of TPE in cluster 1 of SSG (d d d )
$\setminus J$	occurrence of TFT in cluster 1 of SSO $(a_{e_1}, a_{e_2}, \ldots, a_{e_k})$
$\langle a_{e_1} \rangle$	$a_{e_2}, a_{e_3}, a_{e_4}$ , corresponds to delay in cluster 1, 2, 5 and 4, while dek
epres	ents the delay in case of generalized clusters
Jutp	ut: A set of next transition state for all clusters in terms of
	compensating power outages $(t_{c_j} \rightarrow t_{c_{j+1}})$
vhile	$(f_1)$ do
	assign next transition state to compensate for delay due to an
0	occurrence of three phase (L-L-L) fault
1	$\land$ Fault $f_{c_1}$ occurred at cluster 1, which causes a delay $p_{c_1}, p_{c_2}, p_{c_3}$
	and $p_{c_4}$ in corresponding 4 clusters
i	<b>f</b> Cluster $l \rightarrow Power Drop = n_{cl} - p_{cl}(wtt \rightarrow cluster 1)$ Cluster $l \rightarrow$
	$d_{e_1} = d_{e_1, oo_1}$ then
	Next transition, $t_{Ci+1} \rightarrow \text{Super Smart Node} \rightarrow \text{CSR 1}$
1	Fault $n_{c_1} - n_{c_2}$ and $d_{a_1} = d_{a_2}$ gives power drop and transients
	delay in cluster 1 due to tripping of its RFRs (Incorporating CSR)
	1 in SSN transmission network is necessary)
	In SSIV transmission network is necessary)
<b>`</b>	ense in Causier 2 $\rightarrow$ Power Drop = $n_{c_2} - p_{c_2(wtt \rightarrow cluster2)}$ Cruster 2 $\rightarrow$
	$a_{e_2} = a_{e_{2+0.01s}}$ then
	Next transition, $t_{c_{j+1}} \rightarrow$ Super Smart Node $\rightarrow$ CSR 2
	$\langle Fault n_{c_2} - p_{c_2} \text{ and } d_{e_2} = d_{e_{2+0.01s}}$ gives power drop and transients
	delay in cluster 1 due to tripping of its RERs (Incorporating CSR
	2 in SSN transmission network is necessary)
	else if Cluster 3 $\rightarrow$ Power Drop = $n_{c_3} - p_{c_3(wtt \rightarrow cluster3)}$ Cluster 3 $\rightarrow$
1	$d_{\ell_3} = d_{\ell_{3+0.01}}$ then
1	Next transition, $t_{cit} \rightarrow \text{Super Smart Node} \rightarrow \text{V2G}$
	Fault $n_{\alpha} - n_{\alpha}$ and $d_{\alpha} = d_{\alpha}$ gives power drop and transients
	stability delay issues in cluster 3 due to tripping of RERs in cluster
	1 in case of an ecourrence of TPE (Incornerating V2G in SSN
	The case of an occurrence of TPF (incorporating v20 in SSN
	transmission network is necessary)
- ·	else il <i>Cluster</i> 4 $\rightarrow$ Power Drop = $n_{c_4} - p_{c_4(wtt \rightarrow cluster 4)}$ Cluster 2 $\rightarrow$
0	$d_{e_4} = d_{e_{4+0.01s}}$ then
1	Next transition, $t_{c_{j+1}} \rightarrow$ Super Smart Node $\rightarrow$ CSR 3
	\ Fault $n_{c_4} - p_{c_4}$ and $d_{e_4} = d_{e_{4+0.01s}}$ gives power drop and transients
	delay in cluster 1 due to tripping of its RERs (Incorporating CSR
	3 in SSN transmission network is necessary)
6	else if Cluster 1 + Cluster 2 + Cluster 3 + Cluster 4 $\rightarrow$ Power Drop
=	$= n_{c_1} - p_{c_1(wtt \rightarrow cluster_1)} \rightarrow n_{c_2} - p_{c_2(wtt \rightarrow cluster_2)} \rightarrow n_{c_3} - p_{c_3(wtt \rightarrow cluster_2)} \rightarrow n_{c_3} - p_{c_3(wtt \rightarrow cluster_3)} \rightarrow n_{c_3(wtt \rightarrow cluster_3)} \rightarrow $
1	$p_{c_3(wtt \rightarrow cluster 3)} \rightarrow n_{c_4} - p_{c_4(wtt \rightarrow cluster 4)}$ and Clsuter 1, 2, 3 and 4
(	$Delay) \rightarrow d_{e_1} = d_{e_1} + d_{e_1+o_1} \rightarrow d_{e_2} = d_{e_2} + d_{e_2+o_1} \rightarrow d_{e_2} = d_{e_2} + d_{e_2+o_2} \rightarrow d_{e_2} \rightarrow$
	$d_{21,01} \rightarrow d_{21} = d_{21} + d_{21,01}$ then
	Next transition $t_a \rightarrow \text{Super Smart Node} \rightarrow \text{CSR}_1 \rightarrow \text{CSR}_2 \rightarrow$
	$CSR_2 \rightarrow CSR_4$
	Equalize power sharing is achieved by incorporating CSP 1 CSP
	2 V2C and CSD 2 in SSN transmission naturals in order to
	2, V2O and CSK 5 III SSIV transmission network in order to
	accommodate power drop in each cluster. These power drops,
	1.e., $n_{c_1} - p_{c_1}$ , $n_{c_2} - p_{c_2}$ , $n_{c_3} - p_{c_3}$ and $n_{c_4} - p_{c_4}$ in cluster 1, 2, 3 and
	4 will continues the feedback loop as shown in Fig. 1 and it can be
	determined from frustrated demand $F(t)$ , i.e., referred equation 15.
	Moreover, in this case scenario only V2G is considered to be a
	valuable resource to compensate transients delay issue of cluster 3,
	i.e., $d_{e_3} = d_{e_3} + d_{e_{3+0.1s}}$ Whereas, transients delay issues in cluster 1,
	2 and 4, i.e., $d_{e_1} = d_{e_1} + d_{e_{1+0,1s}}$ , $d_{e_2} = d_{e_2} + d_{e_{2+0,1s}}$ and $d_{e_4} = d_{e_4} + d_{e_{2+0,1s}}$
	$d_{e_{A+0,1e}}$ are still there.
6	else
`	next transition, $t_{c_{i+1}} \rightarrow (n_{c_1} \rightarrow n_{c_n})$
6	end
	Send $t_{c}$ , transition state as an input to Algorithm 1
end	

to maintain balanced power or in other words a synchronous stability in  $G^{f}(t)$  and  $D^{f}(t)$ .

In order to confirm the aforementioned case situation, and taking into account the previously mentioned SSG contingency problems, we presume that an SSG may be sensitive to TPF. Suppose that  $\frac{1}{\lambda}$  is the average delay time slot. The average delay *A* caused by a TPF occurring in SSG can be calculated as follows;

$$A = \frac{1}{\lambda} \tag{2}$$

The average delay in eq. (2) caused by a TPF in SSGs equal to only one time frame. So, a more generalized version of eq. (2) may be represented as follows in eq. (3) in the existence of a closed loop probabilistic model depicted in Fig. 3:

$$A = \frac{1}{\sum_{i_{TPF}=1}^{n_1} \lambda_{i_{TPF}}}$$
(3)

where,  $\lambda_{i_{TPF}}$  denotes the TPF delay for every single iteration, in terms of a closed loop probabilistic model is depict in Fig. 3 and  $n_1$  represents the impact of TPF on a network which is based on synchronization of four clusters includes every one having three RERs, consisting of total twelve RERs, i.e.,  $(n_1 = 12)$ , having an interconnection in a form of SSGs power infrastructure.

Similarly, for addressing an issue of demand response pattern in a real time scenario, as in case of SSGs, a case scenario is considered, in which the actual demand  $D^a(t)$  and  $D^f(t)$  are synchronized with each other, by considering the effect of some irregularities in the form of randomness  $R_D(t)$  in a power system, i.e.

$$D^{a}(t) = D^{f}(t) + R_{D}(t)$$
(4)

where  $R_D(t)$  represents the difference or variations between the actual and forecast demand Da(t) and Df(t) respectively. It is a stochastic because the nature of renewable energy source is stochastic due to fluctuating and uncontrollable source of power [108]. It can be described by stochastic nature of uncertainty. The stochastic uncertainty problem can be solved by stochastic programming techniques proposed in [109], [110], and [111]. By incorporating the average delay model in eq. (3) due to an occurrence of TPF into eq. (4) as,

$$D^{a}(t) = \left\{ \begin{bmatrix} D^{f}(t) \times \left(\frac{1}{\sum\limits_{i_{TPF}=1}^{n_{1}} \lambda_{i_{TPF}}}\right) \end{bmatrix} \right\}$$
(5)

eq. (5) can be expressed as a generalized equation to constantly monitor SSG in the form of a closed loop control system probabilistic model as illustrated in Fig. 3.

$$D^{a}(t) = \sum_{i=1}^{n} \left\{ \left| D_{i}^{f}(t) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_{1}} \lambda_{i_{TPF}}} \right) \right| + R_{D_{i}}(t) \right\}$$
(6)

where,  $R_D(t)$  shows the randomness in  $D^a(t)$  and  $D^f(t)$ , and can be calculated by using probabilistic model of an autocorrelation, i.e.,

$$R_D(t) = E[D^a(t)D^f(t)], \tag{7}$$

The synchronous stability obtained when,  $D^{a}(t) \rightarrow D^{f}(t)$ , the deviation of random  $R_{D}(t) \rightarrow 0$ ,

$$G^{f}(t) = D^{f}(t) \tag{8}$$

Likewise, for exploring an issue of generation response behaviour in practical time scenario, as in case of SSGs, a case scenario is considered in which, the true supply  $G^{f}(t)$ and early supply G(t - 1) are synchronized with one another and  $G^{f}(t)$  in addition to some randomness  $R_{G}(t)$  in a power system,

$$G^{a}(t) = G(t-1) + G^{f}(t) + R_{G}(t)$$
(9)

where, G(t-1) is the parameter which is used for controlling purpose, which moves back the closed loop power system network model to previous time frame in practical scenario to constant a balanced between demand and generation response pattern in order to achieved an optimum load flow balancing, as in case of SSGs power infrastructure.

Now, eq. (9) may be written as a generic equation by including the average delay model in eq. (3) caused by TPF into eq. (9) and also in perspective of a closed loop control system probabilistic model as illustrated in Fig. 3.

$$D^{a}(t) = \sum_{i=1}^{n} \left[ \left( G_{i}(t-1) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_{1}} \lambda_{i_{TPF}}} \right) \right) + \left( G_{i}^{f}(t) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_{1}} \lambda_{i_{TPF}}} \right) \right) + R_{G_{i}}(t) \right]$$
(10)

where,  $R_G(t)$  shows the deviation patterns between  $G^a(t)$  and  $G^f(t)$ , and can be calculated by proposing probabilistic model of an autocorrelation, i.e.,

$$R_G(t) = E[G^a(t)G^f(t)]$$
(11)

Similarly, when,  $G^a(t) \rightarrow G^f(t)$ , the random deviation  $R_G(t) \rightarrow 0$ , which stabilizes the SSGs in terms of an achieving synchronous stability between  $G^f(t)$  and  $D^f(t)$ .

Furthermore, to make sure that  $R_G(t) \rightarrow 0$  i.e., the tuning of control parameter should be done for achieving synchronization between  $G^f(t)$  and  $D^f(t)$ .

Moreover, the shortage of power due to an occurrence of TPF in SSG can be expressed in terms of frustrated demand F(t) as,

$$F(t) = E^{a}(t) - G^{a}(t)$$
 (12)

Consequently, in order to obtain the best load flow balancing between generation and demand response patterns, the phenomenon of an expressed demand  $E^a(t)$  must always be satisfied at a defined delay time.

The F(t) take place, when

$$E^a(t) > G^a(t) \tag{13}$$

Examining the average delay model in eq. (3) caused by TPF, eq. (12) can be rewritten as,

$$F(t) = \left\lfloor \left( E_i^a(t) - G_i^a(t) \right) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_1} \lambda_{i_{TPF}}} \right) \right\rfloor$$
(14)

Substituting the generalized expression in context of closed loop probabilistic model as illustrated in Fig. 3, eq. (14) can be rewritten as,

$$F(t) = \sum_{i=1}^{n} \left[ \left( \frac{(E_i^a(t) - G_i^a(t))}{n} \right) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_1} \lambda_{i_{TPF}}} \right) \right]$$
(15)

where F(t) acts as a feedback channel and is sent back into the power system network as B(t), or returning demand, plus closed loop lag, i.e.,  $\lambda_{c_1}$ . Hence, the term for the backlogged demand B(t) is the product of F(t) and closed loop corresponding delay  $\lambda_{c_1}$ ,

$$B(t) = \left(\frac{1}{\sum\limits_{c_1=1}^{n_1} \lambda_{c_1}}\right) \times \left(E^a(t) - G^a(t)\right)$$
(16)

Incorporating the average delay model due to TPF in eq. (3) into eq. (16) and also expressing B(t) in a form a generalized expression, eq. (16) can be expressed as

$$B(t) = \left(\frac{1}{\sum_{c_1=1}^{n_1} \lambda_{c_1}}\right) \times \sum_{i=1}^{n} \left[ \left(\frac{(E_i^a(t) - G_i^a(t))}{n}\right) \times \left(\frac{1}{\sum_{i_{TPF}=1}^{n_1} \lambda_{i_{TPF}}}\right) \right]$$
(17)

Similarly, the reserve r(t) expression can be shown as,

$$r(t) = G^{a}(t) - E^{a}(t)$$
 (18)

The SSG should be in a reserve state, when

$$G^a(t) > E^a(t) \tag{19}$$

Now, by taking into account the average delay model in eq. (3) as well as an extended expression in the context of a closed

loop probabilistic model, as illustrated in Fig. 3, eq. (18) may be recast as follows;

$$r(t) = \sum_{i=1}^{n} \left[ \left( G_i^a(t) - E_i^a(t) \right) \times \left( \frac{1}{\sum_{i_{TPF}=1}^{n_1} \lambda_{i_{TPF}}} \right) \right]$$
(20)

The following should be the threshold policy in the context of reserve r(t) criteria, if

$$r(t) < r_0 \tag{21}$$

Then we have to increase the  $G^{a}(t)$  through SSN transmission network, in order to approach r(t) to *ro* as closely as possible (considering the ramp down constraints), if

$$r_0 < r(t) \tag{22}$$

then, minimizes the  $G^{a}(t)$  in order to approach r(t) to *ro* as closely as possible (considering the ramp down constraints). where ramping limitations are given;

$$r_0 \le G(t) - G(t-1) \le r_0 \tag{23}$$

From eq. (9), we can easily expressed G(t) - G(t - 1) in terms of an expression of  $G^{f}(t) + R_{G}(t)$ , so now eq. (23) will becomes,

$$r_0 \le G^f(t) + R_G(t) \le r_0 \tag{24}$$

The primary issue that has to be solved in this scenario is maintaining constant stability of the returning or backlogged demand or B(t), in every condition. By reducing  $R_G(t)$  via the SSN transmission network, this may be accomplished. The ramping up and ramping down requirements from eqs. (21) and (22), we must regulate the parameter r(t) in order to synchronize it with  $r_0$  for achieving the goal. As a result, eq. (24) may be expressed as follows by minimizing  $R_G(t)$ ;

$$r_0 \le G^f(t) \le r_0 \tag{25}$$

from eq. (1),  $G^{f}(t)$  can be synchronized to  $D^{f}(t)$ , i.e., eq. (25) can be expressed as,

$$r_0 \le D^f(t) \le r_0 \tag{26}$$

Through eq. (26), synchronous stability or in other words in the event that TPF in SSGs occurs, load flow balancing between the demand and generation response pattern may be accomplished, by utilizing SSN transmission network using TNP proactive approach. In this research work, our main claim is that a large penetration of RERs in a form of different clusters in case of SSGs should be properly coordinated with the different receiving side power grid stations through probabilistic model of TNP, i.e., SSN. The SSN properly accommodate load flow balancing between different clusters of SSGs, even in case of an occurrence of TPF with corresponding smaller delay in SSGs. The same mathematical modeling has been formulated for V2G and CSRs, with the corresponding difference in a TPF delay time, i.e.,  $\lambda_i$ , that depend on TPF that occurred along with corresponding

VOLUME 11, 2023

larger delay, it give rise to tripping of certain RERs in SSGs. Therefore, V2G and CSRs must be incorporated in SSN transmission network to completely optimized the SSG for load flow balancing and transients stability. CSRs further optimizes SSG for load flow balancing, whereas V2G incorporation in SSN transmission network further contributes to both load flow balancing and transients stability improvement in SSGs.

If the value of  $R_G(t)$  is greater this mean that the shortage of power is large in SSG due to fault that occurred with larger delay. If CSR is unable to fulfill the requirement of the power in SSGs then V2G is enable to provide power to the SSGs. Therefore, the same mathematical model for both CSR and V2G, only the difference is the availability of power, however the minimum time (delay) required for battery (V2G) than CSR to supply power to the SSG [40]. The mathematical model of spinning reserves based on electric vehicle has been developed and optimizes the scheduling cost for spinning reserve and users in V2G network [112]. The electric vehicle operates as a spinning reserve power source, it provides quick power to grid according to its requirements [113].

The following assumptions can be considered by performing probabilistic load flow analysis [114];

- The load flow equations are to be linearized;
- The power injected from different sources at node are independent;
- The network configuration is assumed to be a constant parameter;
- The discrete and Gaussian probability distribution function are assumed for the conventional generation source and load respectively.

In this work an analytical procedure has consideres for the probabilistic load flow analyses with help of uncertainty modelling as shown in Fig. (46). Convolutional methods and cumulant methods are two categories of analytical techniques. The primary idea behind the analytical technique is simply mathematical, including the convolution of known probability density distribution functions of generated and utilized power in order to get the probability density distribution functions of the associated stochastic variable system states [115]. For some predicted input variable values, load flow equations are linearized by developing them in the Taylor series [116].

A conventional convolution approach was used in the initial generation of analytically based PLF algorithms [117]. Even after using a discrete convolution, such as a discrete convolution in the frequency domain by performing Fourier Transform [118], this still involves an intensive convolution computation for PLF. Convolution methods are rarely used in contemporary studies. Gaussian mixture models and sequence operation theory were used in a few recent efforts to improve the convolution method's performance [119], [120].

The idea of the cumulants was first presented to reduce the load of computing and convolution calculation. These techniques work by substituting the distribution's cumulants for the probability distribution's instances, then conducting the computation using the cumulants. The system's inputoutput sensitivity ought to then be determined, often utilizing a number of deterministic load flow simulations. The output of cumulants is then computed using this level of sensitivity [121].

## B. CONTINUOUS SPINNING RESERVES NON-COOPERATIVE CONTROL BASED PROBABILISTIC MODELING

As a result of TPF occurring in SSGs, it is necessary to be able to tolerate abrupt power interruptions, an integrated CSRs non cooperative control based model in SSN transmission network is operated. A CSR may be able to provide services in response to various power system emergencies, and provide benefits in terms of sustainability and economical operations of power systems, as in case of SSGs.

In the event that an SSG becomes more vulnerable to TPF with a corresponding greater delay, which causes the tripping of wind turbines nearest to fault locations. Therefore, in order to compensate these power outages due to tripping of certain RERs, switching of already incorporated CSRs in SSN transmission network must be necessary. These CSRs optimizes efficiently the load flow balancing of all clusters in SSGs, but the power quality issues due to an occurrence of TPF in SSGs is still there, which causes a significance delay in the operations of different CSRs. The average delay *A* in case of an operation of CSRs can be written as, i.e., eq. (2) will become,

$$A = \frac{1}{\lambda_{i_1} + \lambda_{i_2} + \lambda_{i_3}} \tag{27}$$

The delay incurred due to an operation of CSRs in SSGs in eq. (27) is upto three time slots, as CSR 1, CSR 2 and CSR 3 are operated separately at cluster 1, 2 and 4 of SSGs. Now, by considering eq. (27) in terms of CSRs delay  $\lambda_{CSRs}$  i.e.,

$$\lambda_{i_{CSRs}} = \lambda_{i_1} + \lambda_{i_2} + \lambda_{i_3} \tag{28}$$

then, eq. (27) will become,

$$A = \frac{1}{\lambda_{i_{CSRs}}} \tag{29}$$

The above average delay model is only applicable for three clusters, as we have three CSRs that are correspondingly operates between these clusters. In order to generalize above model to make it applicable for any number of clusters, the eq. (29) can be rewritten as;

$$A = \sum_{i_{CSRs}=1}^{n} \left[ \frac{1}{\lambda_{i_{CSRs}}} \right]$$
(30)

A probabilistic model has been designed to handle the issue of power outages, in terms of an occurrence of TPF with associated higher delay in SSGs, in order to estimate the demand of r(t) and it can be expressed as, i.e., eq. (20) will

become,

$$r(t) = \sum_{i=1}^{n} \left[ \left( G_i^a(t) - E_i^a(t) \right) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) \right]$$
(31)

Similarly,  $D^{a}(t)$  and  $G^{a}(t)$  in eqs. (6) and (10) will now becomes,

$$D^{a}(t) = \sum_{i=1}^{n} \left[ \left( D_{i}^{f}(t) \right) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) + R_{D_{i}}(t) \right] \quad (32)$$

$$G^{a}(t) = \sum_{i=1}^{n} \left[ \left( G_{i}(t-1) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) \right) + \left( G_{i}^{f}(t) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) + R_{G_{i}}(t) \right) \right] \quad (33)$$

where, F(t), B(t) in eqs. (15) and (17) will now becomes,

$$F(t) = \sum_{i=1}^{n} \left[ \left( E_i^a(t) - G_i^a(t) \right) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) \right]$$
(34)  
$$B(t) = \sum_{i_{CSRs}=1}^{n} \left[ \frac{1}{\lambda_{i_{CSRs}}} \right] \times \sum_{i=1}^{n} \left[ \left( E_i^a(t) - G_i^a(t) \right) \times \sum_{i_{CSRs}=1}^{n} \left( \frac{1}{\lambda_{i_{CSRs}}} \right) \right]$$
(35)

## C. COOPERATIVE CONTROL BASED V2G PROBABILISTIC MODELING

In order to address the problem of irregularity in the form of randomness that exist in the power system such as TPF occurred along with corresponding larger delay, a cooperative control probabilistic model based on V2G incorporation in SSN transmission network of SSGs is also designed. In the event that TPF occurs in SSGs, a V2G has the capacity to regulate the power system owing to the resulting power quality disturbances, stabilizing the entire power systems in a short amount of time. Moreover, V2G not only optimizes SSGs for load flow balancing, but also should be consider as a valuable resource for minimizing transients stability issues in SSGs.

In order to verify the potential of V2G in SSGs, we suppose that SSG is now potentially vulnerable to TPF where vulnerability is the inability of the system to withstand an unwanted event. The purpose of a vulnerability assessment is to identify a power system capacity to continue providing service when an unexpected severe situation event happened [122], which mean that SSG must be able to maintain its stability in term of high-quality power flow based on proposed algorithm in case of occurrence of Three Phase Fault (TPF). The average delay *A* in case of an operation of V2G in SSGs, i.e, eq. (29) will now become,

$$A = \frac{1}{\lambda_{V2G}} \tag{36}$$

The above average delay model is only applicable for one cluster, as V2G is only operated at cluster 3 of SSGs, i.e., V2G average delay corresponds to only one time slot. The minimum time required for V2G to supply power to the SSG in case of TPF, because it contains only battery that directly supply power without any external source. By considering the general case that TPF may be occurred in any cluster then battery will supply power with minimum time interval if CSR failed due to availability of insufficient power then.,

$$\lambda_{V2G} \ll \lambda_{CSRs} \tag{37}$$

whereas, in term of generalized expression in a form of closed loop probabilistic model as depicted in Fig. 3, eq. (36) can be rewritten as,

$$A = \sum_{i_{V2G}=1}^{n} \left[ \frac{1}{\lambda_{V2G}} \right] \tag{38}$$

Likewise, for V2G the reserve r(t) is, i.e., eq. (31) will become,

$$r(t) = \sum_{i=1}^{n} \left[ \left( G_{i}^{a}(t) - E_{i}^{a}(t) \right) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) \right]$$
(39)

For V2G the  $D^{a}(t)$  and  $G^{a}(t)$  can be followed as, i.e., eqs. (32) and (33) will now become,

$$D^{a}(t) = \sum_{i=1}^{n} \left[ \left( D_{i}^{f}(t) \right) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) + R_{D_{i}}(t) \right]$$
(40)  
$$G^{a}(t) = \sum_{i=1}^{n} \left[ \left( G_{i}(t-1) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) \right) + \left( G_{i}^{f}(t) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) + R_{G_{i}}(t) \right) \right]$$
(41)

Likewise, F(t) and B(t) in eqs. (34) and (35) will now becomes,

$$F(t) = \sum_{i=1}^{n} \left[ \left( E_{i}^{a}(t) - G_{i}^{a}(t) \right) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) \right]$$
(42)  
$$B(t) = \sum_{i_{V2G}=1}^{n} \left[ \frac{1}{\lambda_{i_{V2G}}} \right] \times \sum_{i=1}^{n} \left[ \left( E_{i}^{a}(t) - G_{i}^{a}(t) \right) \times \sum_{i_{V2G}=1}^{n} \left( \frac{1}{\lambda_{i_{V2G}}} \right) \right]$$
(43)

Moreover, in this case V2G technology may serve as a power buffer to handle the RERs' volatility, improving the RERs capacity to withstand power transients in the event of TPF in SSGs. Let's consider a scenario of forecasted demand like switching peak to be  $D_{t_1}^f$  and the forecasted RERs to cope up with this forecasted demand due to TPF in SSGs are  $V_{t_1}^f$ . Currently, it is quite challenging to deal with transitory impacts with RERs owing to TPF in SSGs because of the reduced power dispatchability of RERs. Therefore, the power dispatchability improvement and transients stability of RERs, we consider the concept of V2G in SSGs. Our control problem is to find out dispatched power schedule  $P_t^f(t+f)$  for power system in order to enhance transients stability and decreases power dispatchability of RERs via V2G. This can be settled by tuning  $P_t^f(t+f)$  to  $(D_t^f(t+f)-V_t^f(t+f)+r_0)$ , where  $r_0$  is constant value utilizing the ramp up and ramp down constraints of eqs. (21) and (22). Hence, the final equation for  $P_t^f(t+f)$  can be followed as,

$$\sum_{i=1}^{n} \left[ P_{t_i}^{f_i}(t_i + f_i) \right] = \sum_{i=1}^{n} \left[ \left( D_{t_i}^{f_i}(t_i + f_i) - V_{t_i}^{f_i}(t_i + f_i) \right) \right]$$
(44)

where,  $V_{t_1}^f(t+f)$  stands for the supply from the V2G to control power outages and transient stability problems.

The aforementioned probabilistic models of the SSN transmission network using TNP, non cooperative control based CSRs and cooperative control based V2G are used for the purpose of studying reliability indices for delay in the event of TPF in SSGs. On the receiving side of various power grid stations in SSGs, necessary corrective measures, such as load flow balancing and transients stability, may be simply done by considering the aforementioned scenario.

#### **III. SIMULATION RESULTS**

We have carried out a simulation using MATLAB/Simulink in order to validate our proposed model of load flow balancing and transients stability in order to analyze the importance of a SSN for load flow balancing in SSGs, a 148 bus system model has been simulated as illustrated in Fig. 14. A simulated model has been split into four clusters, and each cluster contains three RERs that are linked to the other clusters. Moreover, cluster to cluster interconnections can be done in a form of SSN transmission network as shown in Figs. 11 and 12.

Now form a literature point of view, as we already mentioned in Section I, i.e., comparison between proposed work and methodologies of [47], [48], [49], [50], [51], and [52] in the literature have been performed for load flow balancing and [104] for transients stability analysis, as in case of smart grids power infrastructure, this article is based on a general system model in which different smart grid clusters are combine with one another in order to develop a SSGs infrastructure. Moreover, the impact of TPF on load flow balancing and transients stability in SSGs instead of smart grids is considered to be a problem formulation. Therefore, the problem formulated in this paper is new and it is significantly deviated from all earlier proposed studies. it's concluded, the proposed algorithm presented in this paper cannot be directly compared with the previous proposed algorithms in [47], [48], [49], [50], [51], [52], and [104], which is purely based on smart grid approaches for load flow balancing and transients stability analysis. In this review article we proposed two different algorithms, to cater the mentioned problem, one incorporated smart grid based approach for load flow balancing as proposed in [49] in our proposed work of SSGs to clearly illustrates the difference and significance of our analytical approach of SSN transmission network using TNP with the proposed approach in [49] for load flow balancing in SSGs. Secondly, incorporated smart grid distributed controller based approach as proposed in [104] in our proposed scenario of SSGs for transients stability analysis to clearly illustrate the difference and significance of our distributed cooperative control based V2G approach with the proposed approach in [104] in terms of enhancing transients stability issues in SSGs. Moreover, our proposed approach of SSN transmission network and V2G cooperative control based technique is suitable to both smart and SSGs power infrastructure as compared to previous proposed approaches for load flow balancing and transients stability analysis [49], [104], which is only limited to smart grids power infrastructure, as we already discussed in literature review. Moreover, it also elaborates, that why these approaches as discussed in [47], [48], [49], [50], [51], and [52] for load flow balancing and for transients stability analysis [104] is not suitable to be utilized in SSGs. Now, before discussing these algorithms in details, we must first clarify the phenomena, that why SSN is a good way to represents an electrical transmission network in order to provide an optimum load flow balancing in SSGs. This is accomplished by comparing our suggested study with the prior proposed research in [49].

To validate that suggested approach in [47], [48], [49], [50], [51], and [52] are considered to be more computationally costly and the optimum solution in a form of load flow balancing is not guaranteed to achieved, if we consider a larger deviations between loads on the receiving side (KW) of different power grid stations (MW), which is a main reason from transforming smart grids infrastructure into SSGs, a power system proposed in [49] is examined. The main intention of the proposed work in [49] is to reduce an overall cost of an energy import from the public grid, i.e.,

$$\int_0^T P_i(t)C(t)dt \to minimize, \tag{45}$$

where, the active power is represented by  $P_i(t)$ , which is transferred to the grid from the bus. it is considered to be positive and negative for generator and load respectively. Where C(t) represents the price signal (\$/MW) and it is basically depend on time. The formulated problem in [49] is to find out an optimum energy/storage function E(t), that will minimizes the primary objective function in eq. (45). Considering eq. (45), the cost function V(.) can be represented as,

$$V(t, E) = \int_0^T P_i(t)C(t)dt \to minimize, \qquad (46)$$

The main objective function as represented by eq. (45) is equal to the minimizing V(0, 0) in eq. (46), i.e., minimizing an overall cost of the power system over an entire period, which initially starts from an empty storage E = 0. In this case scenario, the optimum solution is determine recursively through utilizing Bellman equation, i.e.,

$$V(t, E) = \min_{\mathbf{E}(t+dt)} \left( \Delta V(E, E(t+dt)) + V(t+dt, E(t+dt)) \right)$$
(47)

The cost function V(t, E) is to numerically find out by utilizing the backward recursion process. It starts from the final time, i.e., t=T, and the representation of cost function is V(T, E) = 0. Now, using eq. (47), the cost function can be minimized at T - dt, by determining V(T - dt, E) for all values of the energy. This operation will continue up till the condition of i.e., t = 0 is reached. Fig. 6 depicts a backward recursion procedure for a one storage device.

Likewise, same procedure will be followed, if we considered another storage device in power system network. The sole difference is that the power system network now includes a new storage device, a new dimension to an existing solution space will also become added. Therefore, in case of utilizing two storage devices in an electrical network, the (1-D) space problem for a single storage device is now considered to be a two dimensional (2-D) space problem, having two variables, i.e.,  $E_i(t)$  and  $E_j(t)$ , both of these variable are considered to be free. The primary difference is that the cost function V(.) is multidimensional now, i.e.,

$$V(t, E_i, E_j) = \int_0^T P_i(t)C(t)dt \to minimize, \qquad (48)$$

where,  $E_i(t) = E_i$  and  $E_j(t) = E_j$  are initial conditions. Therefore, eq. (47) will become,

$$V(t, E_i, E_j) = \min_{\substack{\mathbf{E}_i(t+dt), E_j(t+dt)}} \left[ \Delta V \left( E_i, E_i(t+dt), \\ E_j, E_j(t+dt) \right) + \left( V(t+dt, E_i(t+dt), \\ E_j(t+dt)) \right) \right]$$
(49)

The hybrid optimization technique has been proposed to solve out these problems more quickly. The hybrid optimization is based on AI technique to control and manage flow of energy in hybrid storage energy system [123]. The fast adaptive algorithm for optimization has been developed to solve theses problem more quickly and provide best solution of the problem. The rate of convergence for finding optimal solution has greater than other techniques proposed in [124].

A storage network having 2-D a backward recursion process is illustrated in Fig. 7.

The major limitations involved with the proposed methodology in [49] is that a proportional rise in the amount of storage devices, the power system network's computational difficulty also rises. This phenomenon is shown in Fig. 7, where the inclusion of a second storage device into a power system network also results in the addition of an additional dimension to the optimal solution. This clearly verified the limitation of the proposed work in [49] that an electrical network is only processed in seconds, if it contains only a single storage device, even if considering a large number of



FIGURE 6. Process of Backward recursion for one storage in power system network [49].



FIGURE 7. Process of Backward recursion for double storage devices in a power system network [49].

interconnections between different buses. However, an electrical network with the addition of four or five storage devices may be determined, if it includes only few number of buses. This verified the phenomena that the proposed algorithm as presented in [49] does not give the power system operators an optimum solution in terms of load flow balancing, the capacity of power system can be increases by emerging various RERs along with energy storage devices. The capacity can be increases further by increasing number of buses. Alternatively, the presented approach in [49] is only valid for limited smaller loads (KW) and not heavier loads (MW) on the receiving side of different power grid stations, which is a main reason behind transforming a smart grid power infrastructure into SSGs power infrastructure.

Moreover, we proposed an entirely different and completely new method for the usage of sophisticated storage devices, that results in a different and ideal solution that is capable of solving the mathematical complexity of power systems. Instead of focusing on the issue outlined in [49], as a function to minimized cost within constraints of utilizing few number of RERs and storage devices to adjust light loads (KW) as in case of smart grids, we integrate large number of RERs, CSRs and storage device (V2G) in a form of SSN transmission network using TNP proactive approach and utilized it as a resource to be assigned to different clusters and their corresponding power grid stations, as in case of SSGs. Instead of considering it as a cost minimization problem for suggested work in this article, we assign it as an allocation problem, where an optimum power is devoted to put up heavy loads (MW) in order to optimize the power import from different RERs, so in order to supply an minimum load flow sharing between various power grid stations of different clusters, as in case of SSGs power infrastructure.

The SSN transmission network scale free topological representation is shown in Fig. 11. This type of network representation is very useful, when calculating a risks related to random errors, since randomly disturbance at any node of a network, may relate to the high probability of low degree of connectivity in power system network. In probabilistic terms, we can classified it as, that those nodes in power system network, which are having less interconnections with one another to disturb. Due to such strong interconnectivity among different nodes in a form of SSN transmission network as in case of SSGs, we supply an autonomous flow of power control in a form of an equalize power sharing between every node and their corresponding power grid station, without taking into account specific transmission nodes links. In this scenario, proper assessment of the proposed problem has been formulated using TNP proactive approach in a form of

SSN transmission network by considering the uncertainties due to RERs and power quality disturbances caused by TPF that happened in SSGs, which accounts for both, i.e., load flow balancing and transients stability. From the point of view of regulators and policy makers, an optimal assessment in a form of a TNP, especially in case of SSGs power infrastructure is of great interest because the lack of proper coordination between the different clusters of SSGs can jeopardize the load flow sharing scenario between these clusters and therefore causes a significant power outage issues at different power grid stations of various clusters. Moreover, Fig. 11 shows the TNP in a form of SSN transmission network of just four clusters with every one including three RERs in order to give an prime, i.e., equalize load flow sharing among multi clusters and their corresponding power grid stations of SSGs. Furthermore, this TNP can be further extended for any number of generalized clusters in order to develop a more strengthen interconnected networks for SSGs power infrastructure.

Fig. 12 shows a stronger interconnectivity between various transmission substations/buses in a form of a SSN transmission network in order to provide an optimized load flow balancing in SSGs. Now, in order to consider high penetration RERs integration with one another for gaining an minimum load flow balancing in SSGs, these interconnection between different transmission substations/buses and generation planning face many problems as proposed in [125]. The flexibility of power systems during power quality disturbances, or its ability to meet the variations in load demands through using RERs in SSGs, is one such critical issue, which are receiving much attention from the researchers side. So, by taking into account the RER dependability issues that might be effectively realized in real time scenario, if we consider a case scenario of an unknown time varying additive disturbance in power system, having taken place as a result of TPF in SSGs, a more accurate evaluation of the flexibility of a power system for load flow balancing may be done. In order to achieve the best load flow balancing in SSGs, it is considered that linkage between various transmission substations/buses in the form of an SSN transmission network is a significant resource, considering the risk imposed onto the power system network by using a renewable generating resources due to its unreliable operations, especially when TPF occurred in SSGs.

Moreover, addressing an issue of variability in the RERs for load flow balancing and transients stability in SSGs, an additional level of quick response spinning reserves will also require in the SSN transmission network. For this purpose, a noncooperative control based CSRs and a cooperative control based strategies in a form of V2G is added to an SSN transmission network in order to circumvent critical situations, considering an uncertainties issues in SSGs, i.e., further optimized the power systems for load flow balancing and transients stability enhancement as depicted in Fig. 12. In the presence of power quality issues, these circumstances worsen, i.e., occurrence of TPF in SSGs, where the inter-

77184

action between different RERs in a form SSN for optimum load flow balancing, the amount of RERs curtailment, which causes a significant power quality issues in SSGs and proper optimization control algorithms using CSRs and V2G in order to compensate these issues became even more complex.

#### A. TPF IN SSGs WITH CORRESPONDING SMALLER DELAY

The Fig. 8 analyzed the fact that due to well established interconnection in a form of SSN transmission network, a load flow balancing in terms of an equalized power sharing is obtained on each power grid station of the receiving side, i.e., Buses B4, B8 and B12 of cluster 1, B28, B32 and B36 of cluster 2, B88, B92 and B96 of cluster 3 and B64, B68 and B72 of cluster 4, even in case of an occurrence of TPF at (t = 1s to t = 1.01s), due to reliability issues involve with the RERs, whereas reliability is the capacity of a system to operate as planned under defined conditions for a certain period of time, without any failure, and within the set of operating boundaries [126]. The capacity of the power system to meet demand at all times, which depends on the presence of enough generation and transmission infrastructure and the necessary amount of reserve, may be used to assess longterm performance. This capacity is known as adequate of power systems [127]. Due to the intermittent and irregular characteristics of renewable energy sources, the penetration of renewable energy has a considerable negative influence on reliability of the grid [128]. Therefore, the renewable energy coming from the wind or solar make the power grid is less reliable because we cannot control the wind or light coming from the sun, due to which make the power drop and it's difficult to maintain a reliability balance between demand and supply. Since the TPF interval time is very small, therefore it doesn't cause the tripping of wind turbines nearest to the fault location.

However the stability of the power system has been affected due to TPF, but its again restored due to the SSN network transmission. From Table 1, (*B*4, *B*8, *B*12, *B*28, *B*32, *B*36, *B*88, *B*92, *B*96, *B*64, *B*68) and (*B*72) represent active power buses when TPF occurred with corresponding smaller delay i.e. no tripping of wind turbine due to SSN model. The model of the wind turbine without tripping has been derived in [125], when fault occurred in power system.

In this case our demand response problem is to keep the  $(D_1^f(t), D_2^f(t), D_3^f(t), \dots, D_{12}^f(t)) = D_1^a(t), D_2^a(t), D_3^a(t), \dots, D_{12}^a(t))$  of the 12 receiving stations (RS), i.e., (RS1, RS 2..., RS12) to be synchronized with one another in terms of an equalized power sharing between these receiving side power grid stations. This can be done through an SSN transmission network in SSGs, which makes  $(G_1^f(t), G_2^f(t), G_3^f(t), \dots, G_{12}^f(t)) = G_1^a(t), G_2^a(t), G_3^a(t), \dots, G_{12}^f(t))$  and  $(D_1^f(t), D_2^f(t), D_3^f(t), \dots, D_{12}^f(t))$  for different receiving side power grid stations, i.e., (RS1, RS 2..., RS12) of SSGs. The scale free topological model of an SSN transmission network as



FIGURE 8. Super smart node analysis of four clusters.

shown in Fig. 11 is utilized in order to provide load flow balancing in terms of an equalized power sharing between 12 power grid (*PG*) stations *PG*1, *PG*2, *PG*3....*PG*12 as shown in Fig. 13.

A scale-free topology is a network whose degree of distribution follows a power law in the node, small number of nodes have large degree while greater number of nodes have small value of degree. The power law can be expressed as;

$$P(k) = \alpha K^{-} \gamma$$

where the value of  $\alpha$  and  $\gamma$  are the specified network parameters [129]. The node is intersecting point of the element in the network [130]. A free scale topology has also developed for the wireless network in which the location of node should be satisfied, but new node from time to time must be introduced in wireless mesh of smart grid [131]. The model of free scale topology for wireless mesh in smart grid have been developed in [132].

#### B. TPF IN SSGs WITH CORREPONDING LARGER DELAY

Now let's consider the TPF occurrence time interval is upto 0.1s, i.e., (t = 1s to t = 1.1s) as shown in Fig. 9, i.e., transients effects on different buses, which cause the tripping of the first generation resource of cluster 1 at t = 1s as shown in Fig. 10, i.e., Bus B1 active power and Fig. 14. Due to tripping of RERs 1, it's  $G_1^a(t) = 0MW$  whereas actual supply of other 11 RERs are,  $G_2^a(t) = 973.6MW$ ,  $G_3^a(t) = 973.6MW$ ,  $\dots, G_{12}^a(t) = 973.6MW$ . Now in order to accommodate  $D_1^f(t)$ , i.e., adjusting the the load requirements of tripping bus B1 of RERs 1, whose actual supply is  $G_1^a(t) = 0MW$ . We have to shift the active power from the other eleven RERs of PG1, PG2, PG3,  $\dots, PG12$  as shown in Fig 10 and Fig 11 in order to increased  $G_1^f(t)$ , in other words accommodating  $G_1^a(t)$  of RERs 1. Considering

above scenario, the SSN transmission network as shown in Fig. 11 equalize the whole power system of four clusters having 12 RERs as shown in Fig. 14, but with up to some extent. The receiving station 2 and 3 of cluster 1, 5 and 6 of cluster 2, 8 and 9 of cluster 3 and 11 and 12 of cluster 4 active power is balanced, i.e., 850.6 MW at Buses B8 and B12 of cluster 1, 892.3 MW at Buses B32 and B36 of cluster 2, 905.7 MW at Buses B92 and B96 of cluster 3 and 892.3 MW at Buses B68 and B72 of cluster 4, also referred Table 1, considering these buses with corresponding tripping of wind turbines in SSGs due to an occurrence of TPF, i.e., SSN Tripping of WT. Whereas receiving station 1, 4, 7 and 10 of cluster 1, 2, 3 and 4 active power drops to 771.2 MW, 858.6 MW, 897.2 MW and 858.6 MW as shown in Fig 7, also referred Table 1, Buses (B4, B28, B88) and B64) active power with an occurrence of TPF with corresponding larger delay, i.e., SSN-Tripping of WT. Due to this reason, an unequal power sharing is occurred between different power grid stations, i.e., (PG1 with PG2 and PG3) (771.2 MW, 850.6 MW, 850.6 MW) of cluster 1, (PG4 with PG5 and PG6) (858.6 MW, 892.3 MW, 892.3 MW) of cluster 2, (PG7 with PG8 and PG9) (897.2 MW, 905.7 MW, 905.7 MW) of cluster 3, (PG10 with PG11 and PG12) (858.6 MW, 892.3 MW, 892.3 MW) of cluster 4. Therefore, in order to compensate these issues, an incorporation of CSR 1, 2, 3 and V2G at cluster 1, 2, 4 and cluster 3 in SSN transmission network of SSGs is necessary in order to provide an equalized power sharing between (PG1, PG 2, PG 3) of cluster 1, (PG4, PG 5, PG 6) of cluster 2, (PG7, PG 8, PG 9) of cluster 3, and (PG10, PG 11, PG 12) of cluster 4. The CSRs and V2G in SSGs will actually minimizing the randomness  $R_G(t)$  in all clusters by providing an equalized power sharing between different clusters and their corresponding power grid stations. Before operating CSRs and V2G in SSN transmission network, the  $R_G(t)$  in 12 power grid stations of corresponding 4 clusters of SSGs can be expressed as,

$$R_{G_1}(t)(P_{G_1}(t)) \neq R_{G_2}(t)$$
  
(P\_{G\_2}(t)) = R\_{G\_3}(t)(P\_{G\_3}(t)) (50)

$$R_{G_4}(t)(P_{G_1}(t)) \neq R_{G_5}(t)$$
  
(P\_{G\_2}(t)) = R\_{G\_6}(t)(P\_{G\_3}(t)) (51)

$$R_{G_7}(t)(P_{G_1}(t)) \neq R_{G_8}(t)$$

$$(P_{G_2}(t)) = R_{G_9}(t)(P_{G_3}(t))$$
(52)  
$$R_{G_{10}}(t)(P_{G_1}(t)) \neq R_{G_{11}}(t)$$

$$(P_{G_2}(t)) = R_{G_{12}}(t)(P_{G_3}(t))$$
(53)

where  $R_G(t)$  represent the difference or variation between the actual and forecast supply  $G^a(t)$  and  $G^f(t)$  respectively. If the variation between Ga(t) and Gf(t) is greater, then the value of  $R_G(t)$  is greater, similarly if the value of  $R_G(t)$  is low, it will represent the low variation between them [7]. Due to an occurrence of TPF in SSG, which causes a tripping of RERs 1 of cluster 1, it cause a random variations  $R_G(t)$  between different power grid stations of their corresponding clusters due to an unequal power distributions between them. In this case



FIGURE 9. Bus B1, B4, B28, B88 and B64 active power graphs.



FIGURE 10. Bus B1, B4, B28, B88 and B64 active power drops graphs.

scenario, our main objective is to maintain a same amount of  $R_G(t)$  for different power grid stations of each clusters, so in order to provide an equalized power sharing between these power grid stations and their corresponding clusters. Eqs. 50, 51, 52 and 53 shows the randomness between  $R_{G_1}(t)$ with  $R_{G_2}(t)$  and  $R_{G_3}(t)$ ,  $R_{G_4}(t)$  with  $R_{G_5}(t)$  and  $R_{G_6}(t)$ ,  $R_{G_7}(t)$ with  $R_{G_8}(t)$  and  $R_{G_9}(t)$ , and  $R_{G_{10}}(t)$  with  $R_{G_{11}}(t)$  and  $R_{G_{12}}(t)$ which occurs due to an unequal power distribution of (*PG*1 with *PG*2 and *PG*3) of cluster 1, (*PG*4 with *PG*5 and *PG*6) of cluster 2, (*PG*7 with *PG*8 and *PG*9) of cluster 3, (*PG*10 with *PG*11 and *PG*12) of cluster 4.

The spinning reserves in a form of CSRs and V2G also provides a synchronous stability to each cluster and their corresponding power grid stations in terms of load flow balancing and transient stability in SSGs. CSRs optimizes SSGs for load flow balancing, whereas V2G is acting as a valuable resource both for load flow balancing and transients stability. In order to compensate for larger power outages, integrated CSR 1, 2 and 3 in SSN transmission network are operated in cluster 1, 2 and 4 of SSGs. These CSRs not only contributes to their corresponding clusters for load flow balancing, but they will be consider as a valuable resource for load flow sharing between different clusters as shown in Figs. 15, 16 and 17. CSR 1, 2 and 3 provides an active power of 140.9 MW, 61.8 MW and 62.1 MW, which will be distributed among different clusters as shown in Figs. 15, 16 and 17 in order to achieve a synchronous stability in terms of an equalized power sharing between different clusters and their corresponding power grid stations. The complete picture of load flow sharing between different clusters and their corresponding power grid stations through utilizing CSRs in SSN transmission network of SSGs is shown in Fig. 13. Moreover, Fig. 19 represents a better utilization of CSRs, as proved from buses B145 (CSR 1), B146 (CSR 2) and B148 (CSR 3), in terms of compensating large power outages of cluster 1, 2 and 4 and also shows, that inspite of achieving an equalized load flow sharing between different clusters and their corresponding power grid stations by utilizing CSRs in SSN transmission network of SSGs, still there is an issue of transients stabilities due to an occurrence of TPF in SSGs. Furthermore, with the additions of CSRs in SSN transmission network of SSGs, an equalized power sharing is achieved between PG1 (861.8 MW), PG2 (861.8 MW) and PG3 (861.8 MW) of cluster 1, i.e., refereed Table 1, Buses B4, B8 and B12 with operating (CSR 1 + CSR 2 + CSR 3) in SSN transmission network of SSGs. Whereas, for cluster 2, 3 and 4, there is still some deviations of PG4 (901.8 MW), PG7 (907.4 MW) and PG10 (901.8 MW) with respect to PG5 (861.8 MW) and PG6(861.8 MW), PG8 (913.8 MW) and *PG*9 (913.8 *MW*) and *PG*11 (902*MW*) and *PG*12 (902*MW*), i.e., refereed Table 1, Buses B4, B8, B12, B28, B32, B36, B88, B92, B96, B61, B68 and B72, with operating (CSR 1 + CSR 2 + CSR 3) in SSN transmission network of SSGs. In equation form, these randomness  $R_G(t)$  can be expressed as,

> $R_{G_1}(t)(P_{G_1}(t)) \neq R_{G_2}(t)$ (P\_{G\_2}(t)) = R\_{G\_3}(t)(P\_{G\_3}(t)) (54)

$$R_{G_4}(t)(P_{G_1}(t)) \neq R_{G_5}(t)$$

$$(P_{G_2}(t)) = R_{G_4}(t)(P_{G_2}(t))$$
(55)

$$R_{G_7}(t)(P_{G_1}(t)) \neq R_{G_8}(t)$$

$$(P_{G_2}(t)) = R_{G_9}(t)(P_{G_3}(t))$$

$$R_{G_{10}}(t)(P_{G_1}(t)) \neq R_{G_{11}}(t)$$
(56)

$$(P_{G_2}(t)) = R_{G_{12}}(t)(P_{G_3}(t))$$
(57)

Eq. 54 shows, that by incorporating CSRs in SSN transmission network of SSGs, randomness between  $R_{G_1}(t)$ ,  $R_{G_2}(t)$  and  $R_{G_3}(t)$  is minimized, i.e., referred eq. 50.

Similarly, in order to compensate for smaller power outages of 10.8 *MW* in order to achieve an equalized power sharing between different power grid stations and their



FIGURE 11. Topological model for Super Smart node.

corresponding clusters, integrated V2G in SSN transmission network isoperated along with CSR 1, 2 and 3. Simulation results in Fig. 21 verifies that due to an unexpected outage in terms of TPF at t = 1s to t = 1.1s, a V2G kicks in the power system at t=1s and regulates the power system in terms of active power, i.e., Bus *B*147 active power graph, also referred



FIGURE 12. Transmission Network of Super Smart node.

## IEEE Access



FIGURE 13. Load flow sharing of CSR 1, CSR 2, CSR 3 and V2G for each clusters and their corresponding power grid stations.



FIGURE 14. Super Smart node, CSR and V2G MATLAB schematic diagram.

## TABLE 1. Active power on different buses.

CLUSTER 1							
GENERATING STATIONS (GS)							
Buses	GS 1	Buses	GS 2	Buses	GS 3		
B1(TPF-No tripping of WT)	973.6	B5(TPF-No tripping of WT)	973.6	B9(TPF-No tripping of WT)	973.6		
B1(TPF-Tripping of WT)	-2.51	B5(TPF-Tripping of WT)	973.6	B9(TPF-Tripping of WT)	973.6		
RECEIVING STATIONS (RS)							
Buses	RS 1	Buses	RS 2	Buses	RS 3		
B4(TPF) - SSN- No tripping of WT	952.7	B8(TPF)-SSN-No tripping of WT	952.7	B12(TPF)-SSN-No tripping of WT	952.7		
B4(TPF) - SSN-Tripping of WT	771.2	B8(TPF)-SSN-Tripping of WT	850.6	B12(TPF)-SSN-Tripping of WT	850.6		
B4(CSR 1)	856	B8(CSR 1)	857.3	B12(CSR 1)	857.3		
B4(CSR 1+CSR 2)	858.9	B8(CSR 1+CSR 2)	859.6	B12(CSR 1+CSR 2)	859.6		
B4(CSR 1+CSR 2+CSR 3)	861.8	B8(CSR 1+CSR 2+CSR 3)	861.8	B12(CSR 1+CSR 2+CSR 3)	861.8		
B4(CSR 1+CSR 2+V2G+CSR)	862.1	B8(CSR 1+CSR 2+V2G+CSR	862.1	B12(CSR 1+CSR 2+V2G+CSR	862.1		
3)		3)		3)			
		CLUSTER 2	0				
		GENERATING STATIONS (GS	)				
Buses	GS 4	Buses	GS 5	Buses	GS 6		
B25(TPF-No tripping of WT)	973.6	B29(TPF-No tripping of WT)	973.6	B33(TPF-No tripping of WT)	973.6		
B25(TPF-Tripping of WT)	973.6	B29(TPF-Tripping of WT)	973.6	B33(TPF-Tripping of WT)	973.6		
	DC	RECEIVING STATIONS (RS)	DOF	B	DCC		
Buses	RS 4	Buses	RS 5	Buses	RS 6		
B28(TPF)-SSN-No tripping of	952.7	B32(TPF)-SSN-No tripping of	952.7	B36(TPF)-SSN-No tripping of	952.7		
B28(TPF)-SSN-Tripping of WT	858.6	B32(TPF)-SSN-Tripping of WT	892.3	B36(TPF)-SSN-Tripping of WT	892.3		
B28(CSR 1)	864.8	B32(CSR 1)	897	B36(CSR 1)	897		
B28(CSR 1+CSR 2)	899.7	B32(CSR1+CSR 2)	900.2	B36(CSR 1+CSR 2)	900.2		
B28(CSR 1+CSR 2+CSR 3)	901.8	B32(CSR 1+CSR 2+CSR 3)	902	B36(CSR 1+CSR 2+CSR 3)	902		
B28(CSR 1+CSR 2+V2G+CSR)	902.3	B32(CSR 1+CSR 2+V2G+CSR	902.3	B36(CSR 1+CSR 2+V2G+CSR	902.3		
3) 3) 3)							
		CLUSTER 3	<u>.</u>				
		GENERATING STATIONS (GS	)	8			
Buses	GS 7	Buses	GS 8	Buses	GS 9		
B85(TPF-No tripping of WT)	973.6	B89(TPF-No tripping of WT)	973.6	B93(TPF-No tripping of WT)	973.6		
B85(TPF-Tripping of WT)	973.6	B89(TPF-Tripping of WT)	973.6	B93(TPF-Tripping of WT)	973.6		
	DOP	RECEIVING STATIONS (RS)	DCO	<b>B</b>	Dae		
Buses	RS 7	Buses	RS 8	Buses	RS 9		
B88(TPF)-SSN-No tripping of WT	952.7	B92(TPF)-SSN-No tripping of	952.7	B96(TPF)-SSN-No tripping of	952.7		
B88(TPF)-SSN-Tripping of WT	897.2	B92(TPF)-SSN-Tripping of WT	905.7	B96(TPF)-SSN-Tripping of WT	905.7		
B88(CSR I)	901.5	B92(CSR 1)	909.3	B96(CSR I)	909.3		
B88(CSR 1+CSR 2)	904.5	B92(CSR 1+CSR 2)	911.5	B96(CSR 1+CSR 2)	911.5		
B88(CSR 1+CSR 2+CSR 3)	907.4	B92(CSR 1+CSR 2+CSR 3)	913.8 B96(CSR 1+CSR 2+CSR 3)		913.8		
B88(CSR 1+CSR 2+V2G+CSR	914.3	B92(CSR 1+CSR 2+V2G+CSR	914.3	B96(CSR 1+CSR 2+V2G+CSR	914.3		
3)		3)		3)			
		CENEDATING STATIONS (CS	3				
Dugos	CC	GENERALING STATIONS (GS		Durger	Ce		
Buses	GS	Buses	65	Buses	12		
DC1(TDE No trianing of W/E)	072.6	DCE(TDE No to:so is a of WT)	12	DCO(TDE No trianing of WT)	13		
DC1(TDE Trianing of WT)	973.0	B65(TPF-No tripping of WT)	973.0	B69(TPF-No tripping of WT)	973.0		
Bo1(1PF-Iripping of W1)	973.0	Bob(TPF-Impping of WT)	913.0	B69(1PF-1ripping of W1)	973.0		
Duran	DC	RECEIVING STATIONS (RS)	DC	Duran	DC		
Buses	R5 10	Buses	К5 11	Buses	12		
D64(TDE) SSN No tripping of	059.7	D69/TDE\ SSN No tripping of	050.7	D79/TDE) SSN No tripping of	059.7		
D04(IFF)-SSN-NO URIPPING OF D64(TDF) SSN Tripping of WT	950 6	DOO(IFF)-SSN-NO URIPPING OF	902.7	D12(IFF)-SSN-NO URPPING OF	902.7		
P64(CSP 1)	0.000	Doo(1FF)-SSN-Tripping of W1	092.3	$D_{12}(1\Gamma\Gamma)$ - $D_{12}(1\Gamma)$ -	092.3		
$\frac{D04(OSR 1)}{D64(CSR 1)}$	004.8	$\frac{D}{D} = \frac{D}{D} = \frac{D}$	091	$\frac{D}{2} \left( \frac{C}{2} D + \frac{1}{2} \right)$	091		
$= \frac{1}{1004} \frac{1}{100} \frac$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		$B_{12}(CSR 1 + CSR 2 + CSP 2)$	098.8			
$B64(CSR 1 \pm CSR 2 \pm V2C \pm CSP$	901.0	$B68(CSR 1 \pm CSR 2 \pm V2C \pm CSP$	902	$B72(CSR 1 \pm CSR 2 \pm V2C \pm CSP$	902		
3)	002.0	$\frac{1}{3}$	002.0	$\frac{1}{3}$	004.0		
	1 1	<b>9</b> ]		•)	I		

Table 1, i.e., Buses *B*4, *B*8, *B*12, *B*28, *B*32, *B*36, *B*88, *B*92, *B*96, *B*61, *B*68 and *B*72 active power with operating (CSR 1 + CSR 2 + V2G + CSR 3) in SSN transmission

network of SSGs. Table 1 shows, that with operating V2G at cluster 3, an equalized power sharing is achieved between (*PG*1 (862.1 *MW*), *PG*2 (862.1 *MW*) and *PG*3 (862.1 *MW*))



FIGURE 15. CSR 01 active power transfer to different clusters.







FIGURE 17. CSR 03 active power transfer to different clusters.

of cluster 1, (*PG*4 (902.3 *MW*), *PG*5 (902.3 *MW*) and *PG*6 (902.3 *MW*)) of cluster 2, (*PG*7 (914.3 *MW*), *PG*8 (914.3 *MW*) and *PG*9 (914.3 *MW*)) of cluster 3, (*PG*10



FIGURE 18. V2G active power transfer to different clusters.



FIGURE 19. Bus B145, B146, B148 active power graphs.

(902.3 *MW*), *PG*11 (902.3 *MW*) and *PG*12 (902.3 *MW*)) of cluster 4. Through utilizing V2G along with CSRs in SSGs, finally the randomness  $R_G(t)$  between different power grid stations of each cluster is minimized, i.e., equalized load flow sharing between different power grid stations of each cluster in SSGs is achieved, i.e., Eqs. 54, 55, 56 and 57 will becomes,

$$R_{G_1}(t)(P_{G_1}(t)) = R_{G_2}(t)$$
  
(P\_{G\_2}(t)) = R\_{G\_2}(t)(P\_{G\_2}(t)) (58)

$$R_{G_4}(t)(P_{G_1}(t)) = R_{G_5}(t)$$

$$(P_{G_2}(t)) = R_{G_6}(t)(P_{G_3}(t))$$

$$R_{G_7}(t)(P_{G_1}(t)) = R_{G_8}(t)$$
(59)

$$(P_{G_2}(t)) = R_{G_9}(t)(P_{G_3}(t))$$
(60)

$$R_{G_{10}}(t)(P_{G_1}(t)) = R_{G_{11}}(t)$$



FIGURE 20. Comparison analysis between controller and V2G method.

$$(P_{G_2}(t)) = R_{G_{12}}(t)(P_{G_3}(t))$$
(61)

Simulation results of V2G show that in spite of an outage of t = 0.1s, The main advantage of V2G over CSRs is that it manages the power systems quickly and without any transient problems. Furthermore, V2G cooperative control based probabilistic modeling shows effective results, in terms of accommodating power outage and transients issues in SSGs as compared to CSR non cooperative control based probabilistic modeling as shown in Fig. 26, i.e., comparison analysis of CSR 1, CSR 2 and CSR 3 with V2G. Moreover, by utilizing V2G cooperative control strategies in SSGs, the coordination problem in terms of an equalized power sharing and transients stability enhancement between different power grid stations and their corresponding clusters of SSGs, under high penetration of renewable generation is easily achieved. This observation can be further verified by comparing our proposed distributed cooperative control based approach of V2G technology with the proposed distributed controller based approach in [104], as in case of SSGs. As illustrated in Fig. 20, the V2G cooperative control based strategies can achieve transient stability enhancement within around 0.2s, considering latencies up to 0.1s, i.e., TPF arises time (1sto1.1s) in SSGs. The distributed non linear robust controller achieves the same performance, but it will take some time to achieve transient stability enhancement of power gird, i.e., around 3.2s as shown in Fig. 20, when the latency rate is just 50ms - 400ms. These results verified the effectiveness of V2G cooperative controller based technique in terms of compensating transient stability issues in SSGs, even considering the high latency rate of upto 0.1s, i.e., TPF arises time and even more.

Fig. 22, Fig. 23 and Fig. 24 identify the effect of transients arises due to an occurrence of TPF on a synchronized transmission network in a form of SSN, when CSR 1, CSR 2 and CSR 3 are operated in order to achieve a load flow balancing



FIGURE 21. Bus B147 active power graph.



FIGURE 22. Bus B145 transients graph.

on the receiving side of cluster 1, cluster 2 and cluster 4 of different power grid stations in SSGs. Although CSR 1, CSR 2 and CSR 3 provides an efficient load flow balancing on the receiving side of different power grid stations, but the effect of transients in terms of TPF introduces different power quality issues in these CSRs, which is its major disadvantage as compared to V2G. Moreover, the delays in terms of transients in CSR 1, CSR 2 and CSR 3 is shown in Fig. 24, that observed the pattern of Normal distribution.

Fig. 25 shows the effect of load flow balancing and transient stability improvement in the context of active power along with addition of V2G in cluster 3 receiving side power grid stations. It is evident from Fig. 25, that V2G cooperative control base model will only operates, when TPF arises in a SSG due to any type of power quality disturbances, due to which it provides a cost effective solution, in terms of



FIGURE 23. Bus B146, B148 transients graph.



FIGURE 24. Bus B145, B146 and B148 distribution pattern graphs.

Vehicle to Grid (V2G) (Bus 147) 12 10 TPF at t=1s to t=1.1s Active Power 2 0 .2 -6 1.1 1.2 1.3 1.4 1.5 1.6 1.7 Time

FIGURE 25. Bus B147 transients graph.



FIGURE 26. Buses B145, B146, B147 and B148 transients graphs.

load flow balancing and transients stability improvement as compared to CSR 1, CSR 2 and CSR 3 non cooperative control based model, which are continuously operated. This is due to the reason of transmission of information between different clusters of SSGs in case of any disturbance arises in power systems, i.e., fault occurrence time, which is available to power system operators, especially in case of cooperative control model of V2G as compared to CSRs non cooperative control based model, which are continuously operated and has limited information related to fault behavior on different clusters, as in case of an occurrence of TPF in SSGs. It is evident from time t = 1s, i.e., TPF arises time, where V2G is operated as shown in Fig. 25. Moreover, there is no transients issues arises, when V2G is utilized in order to achieve a synchronous stability in terms of load flow balancing between different clusters of SSG. These results can be further verified in terms of graphical comparison analysis between CSR 1, CSR 2, CSR 3 and V2G as shown in Fig. 26. The whole single line diagram of the SSGs is illustrated in Fig. 27.

## IV. COUNTRIES MOVING TOWARDS SUPER SMART GRIDS

Super grid of the transmission side takes an edge over smart grid in terms of utilizing renewable and clean power generation scattered over a wide area of network with longer distances and therefore supply large amount of power generation into different power grid stations. Smart grid on the other hand provides benefit to customers in terms of saving electricity and sell their excess renewable power generation to the national grid. SSG takes advantages of both technologies, i.e., super and smart grid in order to efficiently integrates different RERs. However, considering todays demand requirements

## IEEE Access



FIGURE 27. Single line diagram of SSGs.

worldwide, smart grids does not fit in a commonly accepted definition of power planning infrastructure. Due to this reason, some of the countries takes initiatives to devised their diverse blueprints for the development of SSGs according to their specific national conditions and demand requirements of their power grid stations. For instance, the united states are moving towards transforming their existent grid infrastructures by utilizing clean energy and wide use of plug in hybrid electric vehicles as proposed in [133]. European countries as already discussed retrofitting existent grid infrastructures through utilizing different RERs in order to reduce  $CO_2$  emissions [4]. Japan tackles the problem of energy crisis through utilizing large scale dispersed PV power in a form of SSGs as discussed in [134]. SAARC countries on other hand in collaboration with SAARC energy center Pakistan demonstrates different projects as proposed in [135] and [136] in order to provide a valuable resource in terms of planning and development of future SAARC SSGs, that will improvise the electrical infrastructure in terms of power sharing between different SAARC countries. Thus, for the purposes of this paper, SSG is considered to ba a valuable resource, that should be economically accepted in terms of transmitting renewable power from the supply side and also operating efficient power plants from the demand side.

## V. RISKS AND OPEN QUESTIONS IN CASE OF FUTURE SSGs VISION

The literature point of review based on a scenario of a future SSG vision clearly identifies the significant benefits of an importing large amount of electrical energy from North Africa based on a massive renewable energy resources, i.e., consider the proposed case studies of gregor czisch [137], [138], [139], the german aerospace center [140], [141], [142] and DESERTEC [143], [144]. All of these case studies concludes the fact, that an electricity imports from North Africa based on RERs will be considered to be much cheaper than the electricity production in a current state, even including transmission scattered over a wide area of network and/or a decentralized power energy system as proposed in [145], [146], [147], and [148], which includes the significance of a decentralized energy systems with the ability to provide two way flow of information between utility and consumers enables an optimum utilization of RERs for load management systems. These case studies concludes the fact, that implementing the future SSG vision is considered to be challenging with respect to technologically and economically perspective but it is also realistic. From literature resources, the main hurdle behind the future vision of developing SSG is a political and most importantly a financial problem as proposed in [149] survey results.

#### **VI. TECHNOLOGICAL UNCERTAINTIES**

Inspite of an investment and financial problems as discussed above, there are still several technological questions, which must be answered to clarify the vision of developing a future SSGs power infrastructure. Among these questions, high voltage direct current (HVDC) transmission system is of current interest. HVDC lines have been significantly utilized over a past decade. However, the main drawback in HVDC technology is that, commutation failures occurs at the inverter stations due to different types of power system disturbances, which have been significantly experienced by the power systems. Moreover, in order to reduce the risks of collapsing AC voltages and addressing an issues of power system instability, developing and testing of several power controls mechanisms, which are capable of handling different parallel schemes, such as regulation, coordination and also accommodates an interchange power control mechanisms among different RERs is necessary to be carried out as proposed in [150]. Due to available smart technologies, the remotely operation of a modern HVDC links can be easily performed. Such operations will provides a strengthen capability to power grid in terms of reliability and stability, i.e., provides a strong protection to the control units in order to reduce problems to be further cascaded down. However, the development of a protection and control scheme based on a multilaver technology, which interconnects different local measurement devices and also an incorporation of an advanced level control schemes into an overall control strategy as proposed in [4] is far from being fully developed or implemented. Virtual Power Plants (VPP) developing and demonstrations is a major future step which needs to be taken as an initiative to develop future SSGs. VPPs can acts as a neutral systems, which can be easily operated with utilizing different diverse energy sources, but incorporating VPPs smart grid approaches in SSGs is still in their pioneering phases [151].

Moreover, due to early stages of development of V2G technology in order to provide electrical energy to power grid stations and also considering the phenomena of an electro mobility with plug in hybrids vehicles, it is difficult for power system engineers to utilized these technologies in SSGs. Therefore detailed simulation studies regarding an incorporating V2G technology in SSGs is essential to be carried out before it's actual implementation in future, which is the main research objective of this research work. Similarly form technology portfolio point of view, the relative weight of storage and smoothing technologies will only become suitable in the process of setting up smart power infrastructure and it is highly uncertain and difficult task to utilized these technologies in future SSGs power infrastructures. The relative weight of smoothing technologies and storage itself in relation to stochastic smoothing from a wide area use of renewable will totally depend on the weights given to the super and the smart part of the SSG power infrastructure, respectively.

## VII. LOOKING AHEAD (EUROPEAN SUPER SMART GRID)

The EU renewables directive will defined a set of certain boundaries for the different member states of the Europe to take initiatives in order to reached 20 % Moreover, the large scale investments is required in an upcoming few years, especially for the construction of a concentrating solar thermal plants (CSP) in the deserts of North Africa and on

the Iberian Peninsula, along with the investments of developing of an onshore and offshore wind farms, as already proposed by the French EU presidency. Through utilizing CSP, the solar plan based on a Mediterranean region foresees 20 GW of RERs capacity in that region by 2020. If all of this capacity were totally based on CSP technology with the addition of overnight storage systems, this could lead to a power production of approximately 80100 TWh/a. This would reduce 80100 Mt CO2 emissions in environment, considering the power production from coal is replaced, which would be considered to be a significant step in order to achieved green energy target compliance in Europe. More importantly, by achieving power production of 20 GW CSP in the next few years would reduce technological and commercial uncertainties, thus proving a promising way for investors to invest their money in renewable electricity generation and transmission systems based on SSGs. Achievements in the RERs field until 2020 will provides further paths for defining future steps needed for shaping the European energy system in terms of achieving the 80 %

## **VIII. CONCLUSION**

SSGs may be susceptible to significant power quality disruptions because of their congestion and scalability problems. In order to identify the irregularity between generation and demand phenomena in the case of SSGs, particularly in terms of the occurrence of TPF, probabilistic modelling in terms of SSN, V2G cooperative control, and CSRs non cooperative control based models have been presented. On the receiving side of various power grid stations, these probabilistic modelling approaches decrease power outages and transient stability problems while effectively balancing the load flow. Additionally these modelling tools greatly increase the accuracy of taking the required steps to effectively prevent future SSG related TPF instabilities. Using the aforementioned probabilistic modelling, the synchronization of complex networks (SSG) is used to examine the dynamic behavior of V2G in terms of load balancing and transient stability. Model based simulation results demonstrated that the suggested load flow balancing and transient stability technique may be used to implement synchronous stability of demand levels in various clusters by achieving synchronous stability of various power grid stations in SSGs. Moreover, the conclusions of probabilistic modelling and simulation demonstrate that the technology based on V2G cooperative control can effectively balance load flows and minimize the problem of transient stability across various clusters in SSGs, and further optimizes the power demand performance more efficiently as compared to utilizing non cooperative control based CSRs approach and distributed controller based approach in SSGs.

## IX. FUTURE TECHNICAL ADVANCEMENT IN THIS PAPER

This paper primary mentioned the situations of utilizing V2G cooperative controller based technique for compensating small power outages in SSGs due to its high cost, which includes battery degradation, the requirement for active inter-

action between the grid and the vehicles, impacts on grid distribution networks, modifications to the infrastructure, and economic, political, ideological, and technological barriers. These limitations will make V2G is attractive fields in terms of its applicability to settled the larger power outages in SSGs. V2G implementation is expected to be more cost effective for both power companies and vehicle owners, even if it may mitigate the lifespan of plug in EVs. Moreover, EV batteries will be less expensive owing to mass manufacture in the future, making V2G a more viable option for compensating greater power outages in SSGs, and therefore eliminate the concept of utilizing CSRs in SSGs, as it provides more efficient results for load flow balancing and transient stability as compared to CSRs. Moreover, the proposed method is formulated and presented to select the optimal control mode in a form of SSN, CSRs and V2G in order to provide an equalized load flow sharing between different power grid stations of each clusters, i.e., cluster 1 (862.1 MW), cluster 2 (902.3 MW), cluster 3 (914.3 MW) and cluster 4 (902.3 MW). This method can be further extended to provide an equalized load flow sharing between all clusters and their corresponding power grid stations, i.e., as in our case different power grid stations of cluster 2 and 4 received the same power at the receiving side, therefore it can be easily extended for all 4 clusters in SSGs. Furthermore, in future, through probabilistic modeling of SSN transmission network for SSGs, we can also adjust the load flow sharing between different clusters in SSGs, according to their required demand load conditions. A scientific work based on a small potion of this idea for load flow balancing in power systems according to required demand conditions through using a smart node transmission network topology in case of smart grids was already proposed in [15]. Moreover, in this paper, our cooperative control strategy of load flow balancing and transients stability using V2G in SSGs does not take into an account certain aspects, i.e.,

- 1) EVs availability at a time of need.
- 2) The customer participation programs, which also plays an important role in an improving of overall power grids stability.
- 3) The load flow balancing using V2G under the constraints of EVs mobility is also considered to be the one of an important aspect.

These issues can be further highlighted in Sec. X-I and Sec. X-J. The computational environment for management and control of power in many clusters of SSGs as shown in Fig. 28. The smooth operations for controlling of power can be managed by SCADA. Every cluster in SSGs have own control center that communicate with other cluster through SCADA.

# X. UNSOLVED PROBLEMS AND TECHNIQUES NEED FURTHER RESEARCH

In this section, we focused on some of the activities that should be considered by the governments, utility companies,



FIGURE 28. Control block diagram of SSG.

regulators and technology firms to make meaningful progress toward the future development of SSGs power infrastructure.

#### A. FUTURE TRANSMISSION SYSTEMS

As, SSGs power infrastructure is based on long transmission lines, therefore a considerable amount of losses occurred due to the resistance of the wires and the equipments through which electricity passes. Considering this scenario, SSGs transmission infrastructure will be updated. New technologies will be incorporated in a power systems, such as flexible AC transmission system (FACTS) can be utilized for maximizing the power transfer capabilities in SSGs. HVDC technology can be utilized for the integration of different RERs, which can significantly reduces transmission losses as compared to HVAC transmission lines. Dynamic line rating (DLR) can be considered as a promising solution to identify the carrying capabilities of transmission network parts in order to prevent an overloading conditions. Moreover, high temperature superconductors (HTS) can be utilized to decrease the considerable amount of transmission losses.

## B. COMPLICATED DECISION MAKING PROCESS IN CASE OF SSGs

In order to provide a proper assessment of a failures probabilities in case of a SSGs power infrastructure, we must have to considered and solves an algorithms based on a much more complex decision problems keeping these failure probabilities in mind, but with in a shorter span of time as proposed in [152], [153]. As we already discussed that a commercial future SSGs power infrastructure may have a large number of nodes, therefore realizing these failure probabilities with respect to each node will be considered to be a challenging research issue. A possible solution to overcome these critical situations is to utilize such a power system network that is totally based on a distributed decision making process. Means, that a different number of failure controllers with each having its own capabilities could be placed in the SSGs power infrastructure at a different places, which are more vulnerable to any kind of power system disturbances. Each of the controller should takes care of different devices at the same time and therefore makes a suitable decisions locally based on different disturbances in power systems. This can correspondingly increases an accuracy in case of a decision making process and therefore reducing an overall failure probabilities response time in SSGs. However, an optimal locally decision making process is not always considered to be globally optimal. Therefore, we need to address properly the critical issues regarding, how to manage a balancing between the power system response time and the effectiveness of the local decision making process.

## C. IMPACT OF AN INCREASING IN ASSET UTILIZATION AND ENERGY CONSUMPTION OF THE CUSTOMERS

The different modern power grid stations are operating at the near edge of its comfortable operation in most of the cases and more often because of the reason of an increasing in energy consumption of the users and especially by considering a methodology from the utility side, which involves an increasing in asset utilization by the customers, i.e., electrical energy as much as possible through utilizing a modern tools and techniques as proposed in [152], [153]. Due to this reason, the power system reliability risk are correspondingly increases, especially in case of SSGs, which is purely based on unreliable RERs. Therefore, in order to reduce the power system reliability risks, as in case of SSGs, there is a need to develop an optimize reliability approaches, through which a power system operators can find out the reliability margins in advance in power systems, so in order to provide a reliable operation to an overall power system network. Secondly, to observe dynamically the reliability margins, the power system operators needs a real time monitoring methods to be utilized in a power system network. In addition, as previously discussed, that maximizing the asset utilization could significantly minimizes the reliability margins and hence increasing the probability of power system failure. Therefore, we must have to balance the asset utilization and the corresponding risk increases through it.

## D. INTEROPERABILITY

Since, SSGs is based on the interconnectivity of different technologies with one another, i.e., HVAC, HVDC, FACTS devices, etc. Therefore interoperability becomes a challenging research issue to make a SSGs power infrastructure properly work, in order for different components of the SSGs might coexist with interconnection between many heterogeneous standard and technologies. The general distribution network of HVDC system for different area is illustrated in Fig. (29).

## E. INTERDISCIPLINARY

The involvement of different organizations and societies in case of developing a future SSGs power infrastructure will make the SSGs research areas an interdisciplinary in nature. For example, the integration of wireless sensor networks, actuators and communication networking technologies with power systems infrastructure. Moreover, incorporation of different cooperative control strategies with one another and also an integration of security protocols in power systems are one such issues receiving much attentions, as in case of SSGs power infrastructure.

#### F. TESTBED

Testbeds of SSGs are important and essential for conducting future research and test results.

#### G. DEMAND RESPONSE PROGRAMS FOR V2G

Considering the utilization of V2G technology in SSGs, The implementation of V2G demand policy responses will be the main focus of upcoming work in order to more effectively modify the dynamics of V2G, as suggested in [154]. In other words, by giving the power grid and the EVs greater flexibil-

ity, these demand response may be efficiently used to enhance the integration of EVs into future SSGs power infrastructure.

## H. DYNAMICS OF THE POWER SYSTEM USING V2G

According to the theory presented [154], voltage dips are unavoidable when using V2G applications in SSGs. There haven't been many studies that take into account weak grid dynamics depending on V2G applications, despite their importance. Similarly, one of the excellent representation models that requires the consideration of weak grid scenarios is the integration of a sizable number of RERs into the power system, as in the case of SSGs, such as wind and solar energy sources with the inclusion of V2G technology. A thorough knowledge of the dynamic behaviors of the electric grid is essential to predicting the efficient and reliable functioning of the electric grid with V2G technology. The Fig. 30 depicts the concept of V2G technology and communications of vehicle with Every thing (V2X), while every thing represent the home, grid or another object. For transmission of DC power the DC-DC converter is required for interfacing purpose between grid and vehicle.

## *I. EVs OPTIMIZATION PROGRAMS UNDER CUSTOMER REQUIREMENT CONSTRAINTS*

Another future research work using EVs technology in SSGs is considering the EVs optimization problem, especially under the constraints of customer requirements. Customers requirements can be as, the requirement for consumers to charge their electric vehicles batteries to the required state of charge (SOC) prior to departing. This is considered to be the most significant objective, which must be incorporated in EVs optimization programs, due to the fact that EV customers are often more eager to engage in the charge management program and supply electrical energy to utility, i.e., V2G only, if the customer EVs are fully charged when needed.

## J. DEMAND RESPONSE MANAGEMENT USING V2G CONSIDERING EVS MOBILITY

Another important aspect is considering the network model of V2G energy network to performed DRM under the constraints of EVs mobility. This idea of DRM using V2G, considering EVs mobility was well defined for several districts in [55] for the case scenario of smart grids. This idea can be easily extended to SSGs, where we consider the scenario of DRM using V2G under the constraints of EVs mobility for different countries.

### K. PERFORMANCE

SSGs power infrastructure is a very complex due to heterogeneous network systems, large scale deployments, interdisciplinary research areas, i.e., control, communication, power, etc., and also due to consisting of various dynamic and nondeterministic power systems. Therefore, efficiency is a very important terms in case of SSGs power infrastructure, that need to be properly addressed in order to provide better, fast,



#### FIGURE 30. Flow of energy in different modes.

secure, robust controls and performing reliable operations in SSGs.

## L. ENERGY MANAGEMENT TECHNOLOGIES ON SSG

The main role of these technologies to control and manage the flow of energy along with distribution of power according to the requirements of the load. The following wireless and wired technologies that can be used for management of power in SSGs [155].

## 1) ZIGBEE WIRELESS TECHNOLOGY

The usage of power in Zigbee wireless technology is very low which is major advantages of Zigbee. It's mostly used for high level application and based on IEEE 802.15.4 specification.

The infrastructure of Zigbee consist of six layers as shown in Fig. (31) [156].

Zigbee has a range of 30–50 m and operates in the 2.4– 915 MHz frequency band; despite its small range and low data rate, it can still send data at up to 250 Kbps. It is a popular choice for smart grids since it enables providers to transmit and receive signals [157].

#### 2) WIRELESS MESH

Wireless mesh is the major components of smart grid structure due to safety and effectively communicate the power distribution in smart grid system [158]. Wireless access points are used by a wireless mesh network (WMN) to create connections inside a network infrastructure. The distinctiveness of WMN lies in its decentralization and straightforward permission of message transmission to only the subsequent nodes rather than taking into account the complete network from the starting state. To efficiently and effectively reach the whole network, message transfer to each subsequent node is moved in an algorithmic order. Additionally, this may be automated with suitable algorithms. Electric utilities benefit from the adoption of wireless mesh, particularly when processing output data. The transmission of WMN across mesh nodes, mesh clients, and gateways makes it possible to meet the demand to increase efficiency and services while also attracting more users to smart grid infrastructures. It is also crucial to establish connectivity between smart meters and data center by wireless mesh networks [157]. As beneficial as wireless mesh is for smart grid communication, installation timelines and operating costs might occasionally be long owing to highly optimized equipment used in the process or insufficient availability of such equipment in the intended usage locations [159].

#### 3) CELLULAR NETWORK COMMUNICATION

Cellular refers to network distribution across an area, that are made up of transceivers cell. Cellular Network Communication (CNC), is similar to Zigbee and wireless mesh, is recognized as one of the communication technologies suited for smart grids and their applications. It includes, GSM, GPRS, and 3 G. High capacity, increased speed, meaningful data, and voice features are characteristics of CNC [159]. It also supports mobile cellular devices and includes a multimedia roaming option. Using them in smart grids is more than acceptable because they are the only method of communication for sensitive commercial dealings and mission critical services. CNC is the hub of communications that easily connects end-to-end wireless nodes [160].

#### 4) WLAN

Computers and a number of other devices are frequently connected to the internet using the Wireless Local Area Network (WLAN) technology, which is considered to be industry standard [161]. Even though it is combined with ethernet technology, it is a modified form of that technology. It uses radio frequencies to broadcast and receive data quickly without connections. WLAN has a number of advantages for the smart grid [162]. Due to safety concerns, lack of availability, disruption, and data rates, WLAN has historically had a loose reputation in the energy industry. However, this is no longer the case, especially for smart grids [162], [163].

The following are few wired technologies that have been studied to take into account in the SSGs infrastructure.

#### 5) POWER LINE COMMUNICATION (PLC)

PLC uses a line cable to communicate from device to device at high speed [164]. Due to the presence of existing lines, PLC technology is the only one whose setup cost can be comparable to that of wireless technology [165]. Due to its connections with the meter, it has been the most effective communication technique for electricity meters [166].

#### 6) DIGITAL SUBSCRIBER LINES (DSL)

The subscriber lines for digital DSL is a technology that transmits digital data quickly over the voice telephone network. The fundamental features of DSL that help to keep installation costs down. This explains why so many businesses choose DSL technology for smart installation. This technology has been incorporated into several active international projects as a smart grid solution [167]. It makes use of a communication box that is installed on the site of the energy user, and data about the pattern of usage of energy is sent to the utility company that employs the technology through DSL [164].

#### 7) FIBER OPTIC

Fiber optics play a crucial role in information transfer that cannot be overstressed. The telecommunications framework has changed as a result of the novelty surrounding the usage of this material. In order to prevent interference and signal noise when transporting data from one location to another, this technology employs light waves instead of electricity [168]. The enormous volume of data that can be transmitted across a very long distance of more than one hundred kilometers in a relatively little period of time is another benefit of this wired media. With a layer of plastic cover that guards against damp, installation is simple. Fiber optics are available in many multimode and single mode configurations. Although they are employed in communication networks as well as in medicine and surgery, they also play a significant part in the smart grid, just as coaxial and twisted pair cables, which are members of the same wired communication group [169].

#### 8) SCADA

Remote control of industrial operations is made possible by SCADA, which consists of hardware and software components. SCADA has recently been used for energy management, such as smart grids [170]. The advantages of SCADA include its ability to evaluate real time data and communicate directly with equipment like pumps, sensors, and



FIGURE 31. Layers of Zigbee.

motors [171]. The Human Machine Interface (HMI) software is used to perform this. Additionally, Supervisory Computers, Remote Terminal Units, Programmable Logic Controllers, Communication Infrastructure, etc. are frequently utilized in SCADA systems. Fig. (32) depicts the fundamental architecture of a SCADA system. Additionally, it maintains event logs, boosts productivity, and reduces downtime. SCADA architecture, common operational protocols, data transfer technologies, and graphical user interfaces all improve the high level supervisory process. While challenged with cybersecurity difficulties that are now being controlled utilizing contemporary security solutions, a number of businesses, particularly the power industry with big SCADA systems, have evolved to become resourceful in vast distance site monitoring [172].

The Fig. (33) shows the various categories of wireline and wireless communication on SSGs [173].

## M. MODELLING AND INCORPORATION OF RENEWABLE ENERGY SOURCES IN SSGs

When an approximate mathematical model is given, it is simpler to verify the system compatibility before manufacturing without spending any money. The modeling method aids in decision making by enabling the identification and enhanced understanding of component features. Although accurate performance prediction is reflected in detail modeling, designing the ideal model is either too difficult or takes too much time. A good model should be matched between complexity and reliability.

The next step after creating a mathematical model is to deploy a computer simulation model to test the system outputs under various input scenarios and to check for stability. Before creating the final working model, these computer simulations may be used for additional analysis, to replace specific components, or to reconsider the system and optimize it [174]. The mathematical modelling of the system is also required to design appropriate controller for any proposed system. The performance of each dynamic state in model can be evaluated by designed controller [175].

#### 1) MATHEMATICAL MODELLING OF WIND TURBINE

In this review paper, we have considered the wind turbine as a renewable energy source linked with clusters. The Fig. (34) illustrates the fundamental parts of conventional wind turbine generator [176]. To design a robust nonlinear controller for MMPT of wind turbine a mathematical dynamical equations should be required to track the maximum power from the wind to fulfill the requirement of the clusters through SSN and designed energy management algorithm. This techniques can be implemented for SSGs having multiple cluster comprises of wind turbine. The formulation of mathematical modelling of wind turbine can be follow as;

Considering "K.E" is the kinetic energy of an object having mass and velocity "M" and "V" respectively, a constant acceleration "A" produced an object due to applied force



FIGURE 32. Main components of SCADA system.



FIGURE 33. Wire and wireless communication categories.

which covers some displacement "D", so, the work "W" done on the body is;

$$W = F.D \tag{62}$$

The work is also equal to change in energy therefore,

$$k.E = F.D \tag{63}$$

Now according to the Newton second law of motion;

$$F = MA \tag{64}$$

Putting the value of eq. (64) in eq. (63) we get;

$$K.E = MAD \tag{65}$$

The third equation of motion is given as;

$$u_f^2 = u_i^2 + 2AD \tag{66}$$

From eq. (66) the value of 'A' can be obtained as;

$$A = \frac{u_f^2 - u_i^2}{2D}$$
(67)

where  $u_f$  and  $u_i$  are the final and initial velocity of an object. Putting the value of 'A' from eq. (67) into eq. (65) by

assuming  $u_i = 0$  and  $u_f = V$ ;

$$K.E = \frac{1}{2}MV^2 \tag{68}$$

The assumption behind this kinetic energy eq. (68) is that the mass is constant. But for considering wind (moving air) like a fluid with variable density and velocity, having no fixed mass. Because of this, the kinetic energy law is expressed with a



FIGURE 34. Major blocks in wind turbine.

factor of 2/3 rather than 1/2 [177]. The power 'P' of wind is equal to the derivative of K.E with respect to time, therefore;

$$P = \frac{dK.E}{dt}MV_a = \frac{1}{2}\frac{dM}{dt}V_a^2$$
(69)

where  $V_a$  is the velocity of wind and  $\frac{dM}{dt}$  is the flow rate of mass which is equal to;

$$\frac{dM}{dt} = \rho S V_a \tag{70}$$

where  $'\rho'$  and 'S' are the density of wind and area in which air is flowing respectively.

Using the value of eq. (70) in eq. (69) to get;

$$P = \frac{1}{2}\rho SV_a^3 \tag{71}$$

The real power  $P_r$  of the wind is obtained from the blades of the rotor is equal to the difference between upstream and downstream power of the wind turbine [178].

$$P_r = \frac{1}{2}\rho SV_a (V_f^2 - V_i^2)$$
(72)

where  $V_f$  and  $V_i$  are the upstream velocity of the wind at which enter the rotor blade and downstream velocity of wind at the outer end of rotor blade respectively.

Now from eq. (70) the mass flow rate can be written as;

$$\rho SV_a = \frac{\rho S(V_f + V_i)}{2} \tag{73}$$

where  $V_a$  is the average of in and out velocities of the rotor blade in wind turbine, therefore, eq. (72) can be expressed as;

$$P_r = \frac{1}{2}\rho S(V_f^2 - V_i^2)\frac{V_f + V_i}{2}$$
(74)

The eq. (74) can be simplified as;

$$P_r = \frac{1}{2} \left[ \rho S \left( \frac{V_f}{2} (V_f^2 - V_i^2) + \frac{V_i}{2} (V_f^2 - V_i^2) \right) \right]$$
(75)

$$P_r = \frac{1}{2} \left[ \rho S \left( \frac{V_f^3}{2} - \frac{V_f V_i^2}{2} + \frac{V_f^2 V_i}{2} - \frac{V_i^3}{2} \right) \right]$$
(76)

$$P_r = \frac{1}{2} \left[ \rho S V_f^3 \left( \frac{1 - (\frac{V_i}{V_f})^2 + (\frac{V_i}{V_f}) - (\frac{V_i}{V_f})^3}{2} \right) \right]$$
(77)

or

where

or

$$P_r = \frac{1}{2}\rho S V_f^3 W_h \tag{78}$$

 $W_h = \frac{1 - (\frac{V_i}{V_f})^2 + (\frac{V_i}{V_f}) - (\frac{V_i}{V_f})^3}{2}$ 

1

$$W_h = \frac{(1 + \frac{V_i}{V_f})(1 - (\frac{V_i}{V_f})^2)}{2}$$
(79)

where  $W_h$  is known as Betz limit, which is the ratio of upstream power of the wind taken from the rotor blade of turbine.

However, the ratio downstream velocity to upstream velocity of wind turbine is denoted by C;

$$C = \frac{V_i}{V_f} \tag{80}$$

or

$$C = \frac{speed \ of \ blade}{speed \ of \ wind} \tag{81}$$

The value of speed of blade can be calculated from eq. (82) as;

speed of blade = 
$$\frac{\text{rotational speed of turbine} \times R}{\text{speed of wind}}$$
 (82)

where R is the radius of turbine and rotational speed can be measured in *radian/sec*.

Now put the value of eq. (80) in eq. (79) to yield;

$$W_h = \frac{(1+C)(1-C^2)}{2}$$
(83)

Most of the wind will flow through the gaps between the blades of a wind turbine if the rotor rotates too slowly, generating little electricity. In contrast, if the rotor spins too quickly, the rotor blades operate as a solid barrier, preventing the wind flow and lowering the amount of power that can be extracted. In order to get the most power out of the wind flow, the turbines must be built to run at their ideal wind tip speed ratio. Greater the value of C the performance of the generator will be better. But there are drawbacks as well. Due to the contact of sand or dust grains in the air, high causes deterioration of the blades leading edges. This would imply the use of a specific coating material that is resistant to erosion, which might raise the cost of electricity. In addition to producing noise and vibration, higher decreases rotor efficiency owing to drag and tip losses, and too-high rotor speeds can result in turbine failure [179].

The infrastructure of super smart grid system comprises various RERS. It can link generators that are radially connected to the grid [180]. SSGs is adaptable enough to operate dispersed generation systems and can even control voltage for deeply integrated RES [181]. Frequency regulated and responsive loads are added to the smart grid in order to increase reliability. High levels of integration of small and medium sized renewable energy generating units give rise to a number of problems, including synchronization between sources and generally with the grid, the reduction of harmonics to prevent distortion, and the smoothing of voltage variation [182]. The operation of optimization in super smart grid, that enables end users to effectively regulate energy usage and prevents power outages. Due to distributed power generation and storage, the smart grid has opened up a wide range of opportunities. This energy can be dynamically contributed to the system at different levels, such as the transmission and distribution sector [183]. With the aid of significant developments in communication, computation, and sensor networks, the issues of disturbance and power transfer can be resolved. The basic model of a single cluster smart grid is shown in Figure (35). Optimization tools is used to check and balance the requirements of the power between demand and supply sides along with optimization of other parameters.

To resolve the energy storage issue has frequently grown as a result of the practical uses of renewable energy sources. Electric vehicles may store energy from the smart grid system to become charge in G2V mode [184]. The Linkage of PEVs and PHEVs with RERs provide a numerous advantages including improving the environment, lowering vehicle operating costs, consuming less fossil fuel, and emitting fewer dangerous gases [185]. Fig. (36) shows the control and manage of power for smart home management system [186]. The power from the renewable energy resources can be directly supplied to the grid. The battery can be charged from the grid when the requirement of the loads is low and discharged when the demands of the load is high.

The combination of a large number of PHEVs act as a loads on the grid might lead to a number of issues, including voltage stability and overloading of transformers or lines [127]. Many approaches have been proposed to solve these problems in PHEVs [187]. Two types bidirectional converters required for smooth operation of G2V and V2G modes. The first converter operate as a boost rectifier while other convert act as a buck converter that directly connected to the battery of electric vehicle in G2V Mode. In V2G modes the operation of these converters become reverse. The working principle of these converters based on constant current-constant voltage (CC-CV) topology. The mathematical modelling of both converters have been derived in [188].

Batteries from PHEVs serve as distributed energy resources in a V2G system by directly fulfilling demand during peak times, which relieves stress on overburdened distribution networks. The aggregator, which a communication link is utilized for interaction between vehicle owners and distribution network service providers (DNSPs), estimates the quantity of electricity transferred by vehicles to the grid [189]. When batteries employed in V2G systems suddenly run out, the distribution networks to which they are attached may experience voltage variations, which in turn results in voltage stability issues. Additionally, power electronic converters serve as interfaces between the grid and the batteries of plug-in hybrid electric vehicles (PHEVs), therefore a reliable switching system is necessary to preserve the power quality and stability. A powerful controller has to be needed that can reduce the voltage fluctuation problem through reactive power management and improve the power quality of distribution networks in V2G system [190].

### 2) MODELLING OF V2G SYSTEM

In order to ensure the steady performance of such systems with excellent power quality, designing of linear and nonlinear controllers relying on the mathematical model of V2G systems would be essential. The Fig. (37) illustrates the interaction between grid and vehicle during G2V (Charging) and V2G (discharging) modes with the help of interfacing medium such as power converters [191].

The modelling of a V2G system is proposed in this section of the paper.However, a battery model is analyzed before developing the primary V2G system as this is a key component of a V2G system. The Fig. (38) depicts the model of the battery that mostly used in V2G system;



FIGURE 35. Infrastructure of single cluster in SSGs.

The electric circuit of the battery includes three resistors  $R_0$ ,  $R_1$ ,  $R_2$ , one capacitor  $C_1$  and internal voltage source  $E_m$  [192]. The Fig. (39) shows the V2G system with battery model.

Now after applying KCL law in Fig. (39), we have;

$$I_s = I_c = I_{R_1} + C_1 \frac{dV_{C_1}}{dt}$$
(84)

where  $V_{C_1}$  represent the voltage across capacitor  $C_1$ , the capacitor  $C_1$  and resistor  $R_1$  are parallel with each other therefore,

$$V_{C_1} = I_{R_1} R_1 \tag{85}$$

Using the value of eq. (85) in eq. (84) to get;

$$\frac{dI_{R_1}}{dt} = \frac{1}{\lambda} (I_c - I_{R_1})$$
(86)

where

$$\lambda = R_{1C1}$$

After applying KVL on the grid side of Figure (V2G system) we get;

$$\frac{dI_g}{dt} = -\frac{R_L}{L_L} + \delta \frac{V_c}{L_g} \tag{87}$$

where  $\delta$  is the duty ratio of the converter that depends on the modulation index and firing angle, while  $R_L$  and  $L_L$  represent the resistance and inductance of the line respectively.

The eqs. (86) and (87) show the model of V2G system in time variant domain. For designing the controller for V2G system the model must be converted into time invariant model that can be represented on dq frame [190], therefore;

$$\dot{I}_{R_1} = \frac{1}{\lambda} (\delta_d I_d + \delta_q I_q - I_{R_1})$$
(88)

$$\dot{I}_d = -\frac{R_L}{L_L}I_d + \omega I_q - \frac{E_d}{L_L} + \frac{V_C}{L_L}\delta_d$$
(89)

$$\dot{I}_q = -\frac{R_L}{L_L}I_q - \omega I_d - \frac{E_q}{L_L} + \frac{V_C}{L_L}\delta_q$$
(90)

where  $\omega$  represents the angular frequency. The  $\delta_d$  and  $\delta_q$  show the duty ratio of converter in d and q frame respectively. The  $E_d$  and  $E_q$  are the grid input voltage in d and q references respectively.

The active and reactive powers injected from vehicle to grid in dq frame can be given as;

$$P_{V2G} = E_q I_q + E_d I_d \tag{91}$$

$$Q_{V2G} = E_q I_d - E_d I_q \tag{92}$$



FIGURE 36. Monitoring and control system for smart home.



FIGURE 37. Directions of power in G2V and V2G mode.

The eqs. (91) and (92) show the active and reactive from the vehicle to grid respectively.

In dq reference frame the value of  $E_d = 0$  [193], therefore eqs. (91) and (92) becomes;

$$P_{V2G} = E_q I_q \tag{93}$$

$$Q_{V2G} = E_q I_d \tag{94}$$

The eqs. (88), (89) and (90) represent the complete dynamic model of V2G system. However the quality of power depends on the current as expressed in eqs. (10) and (11). Therefore a robust controller must be required to transfer the energy from vehicle to grid after TPF occurred in any cluster of the SSGs. The V2G technology is the possible way that fulfill the requirement of the load which is the main advantage of this topology while controlling of power flow is a challenging

task in super grid by introducing V2G technology [194]. The non linear controller to be needed for smooth operation of power converters in G2V and V2G modes. The regulation of voltage can be done through a boost converter while correction of power factor has performed by buck boost converter connected to the battery of electric vehicle [195], [196].

### **XI. CONTROL STRATEGIES OF SUPER GRID**

The control challenge is extremely difficult since it demands for achieving reliability, efficiency, and outstanding performance. The control designer are also responsible for the process modeling, simulation, and diagnosis of parameters [197]. The usage of PID controllers has decreased as a result of safer operating guidelines and the need for extra compensators to ensure PID controller functionality [198].



FIGURE 38. Battery model in V2G system.



Grid Supply Point

FIGURE 39. Circuit diagram of V2G System.

In addition, PID experiences integrator windup problems, which worsen disturbance responses when the actuators are saturated. An antiwindup control strategy is typically used to address this problem, as seen in Fig. (40).

For PID based control methods that reduce the impact of measurable disturbances, a feed forward controller is also essential [199]. The sensed disturbances are utilized to identify the controlled variable based on a defined model in order to maintain the control at the reference point. Errors in modeling are compensated by the feedback [200]. The implementation of the technique is possible in serial or parallel arrangement. the serial arrangement is shown in Fig. (41).

For regulating the grid system filtering process, inverting process, DC link, and controlling grid side parameters, a LQR based control has been implemented in [201]. In the cascade

control approach, the inner loop is used to regulate disturbances, while the outer loop is used to control the output of the process. A cascade control block diagram is shown in Fig. (42). A primary controller regulates the collector region by using secondary controller. The Distributed Control System (DCS) mode having cascading control of a solar array is used for regulating average temperature and combining oil demands that discussed in [202] and [203].

As seen in Fig. (43), Adaptive Stochastic Control (ASC) may optimize the sources and loads that are part of smart grids. These technologies can recognize the worst case events that might occur in the plants [204]. The control center based on adaptive system that linked with generation and consumption unit. The different line transformers are also used in transmission network according to the requirements of the load.



FIGURE 40. The antiwind setup with PID controller.





FIGURE 42. Cascaded control mechanism.

Volt-VAR Control (VVC) is also widely used to maintain the operating requirements for the distribution system of the smart grid. The main goal of VVC is to achieve appropriate voltage levels at all distribution system locations while taking into account all loading conditions. Due to "self healing," this control strategy improves efficiency while maintaining voltage level. It uses standalone, On Site Voltage Regulator (OVR), rule based DA control, heuristic voltage regulation, and distribution based model VVC [205]. AGC (Automatic Generation Control) guarantees the frequency and power setting schedule stability of the smart grid. The AGC technique reduces the load forecasting ambiguities and failures [206]. Autonomous Plug in Electric Vehicles (PEVs) employ optimum charging control algorithms. There are two different types of PEV coordinating strategies: distributed and decentralized. The interaction and utilization of PEVs in grid level carried out in [207].

The delay and phasor qualities prevalent in smart grid systems are corrected by the Adaptive Power Damping Controllers. Power System Stabilizers (PSS) and FACTS controllers are the two different types of Power System Electromechanical Mode Damping Controllers. A



FIGURE 43. Block diagram of adaptive control for grid system.

number of Phasor Measurement Units (PMU) for damping control are presented in [208]. The block diagram for the Adaptive Power Damping Controller is depicted in Fig. (44).

The Model Reference Control (MRC) technique is used to make sure the smart grid system is reliable and long lasting. Model reduction, Collective Control, and Control the process of inversion are the three phases of the MRC technique suggested in [209].

The flow of energy in solar systems, smart grids, storage devices, and consumer loads is managed by adaptive logic controllers (ALC). The balanced power flow and the effects of the weather are also taken into consideration. ALC can create its own controlling and optimizing algorithms for issues like energy demands, pricing signals, the availability of resources, loading schedules, etc. [210]. ALC block diagram is shown in Fig. (45). Several other approaches have been studied for the control and management of super smart grid which are discussed in below sections;

# A. ROLES OF OPTIMIZATION AND ARTIFICIAL INTELLIGENCE IN SUPER GRID

The optimization tool is proposed to optimize the cost of different parameters, while Artificial Intelligence (AI) used for prediction of output based on previous data.

#### 1) OPTIMIZATION TECHNIQUES

The term super smart grid refers to a complex system, and the components of a smart grid are classified according to a set of standards, including homogeneous parts, interconnected elements, and characteristics of general interest. For a residential area with low requirements of power consumption,



FIGURE 44. Damping controller for power.



FIGURE 45. Adaptive logic controller for control and operation of power in smart grid system.

a single optimization technique may be used. Optimization of a super smart grid system is a tough task due to its complex infrastructure. Typically, the classical and game theoretical approaches are used to optimize and guarantee the flexibility and sustainability of the grid [211].

The grey wolf optimization (GWO) technique has been proposed for the super grid of Iran having 400V voltage. The purpose of designed GWO to control the flow of power and reducing the losses in transmission lines along with other parameters. It's been proved that the proposed GWO shows better performance against other optimization techniques in term of cost minimization, power losses and voltage stability index (VSI) [212]. The distributed generation methods for improvement of Iraq's 400kV gid system has been discussed in [213]. A GwO technique has been proposed for the improvement of voltage profile in power system infrastructure [214].

The major problem in super grid is the flow of optimal power by designing a novel hybrid technique based on particle swarm optimization (PSO) algorithm and salp swarm optimization (SSO) algorithm. This technique is proposed for solving the optimal power flow problem which deals with non linear optimization problem in electric power system. It solved different designed objective functions such as cost, power losses and enhancement in voltage [215]. The selection of features plays main role in the field of AI which is based on availability of data to increase the performances, therefore this problem has been reviewed in [216] by designing GWO technique. A improved Differential Evolution algorithm (IDEA) has been proposed to optimize the flow of power. The effectiveness of proposed technique has been tested by solving different objective functions under standard IEEE 57 bus system and IEEE 30 bus system for the applicability and uses in real life scenario [217]. The linear programming (LP) and the sequential quadratic programming (SQP) tools have used that available in matlab environment to control the reactive optimal power flow and decreases the energy losses to make the voltage profile effective [218].

To determine the location for installation of PV system using GWO algorithm to optimize the losses caused by active power has studied in [219]. The Solar microgrid along with hybrid energy storage system (HESS) has been proposed to increase the storage of energy and it plays basic role in super grid infrastructure [220]. The integration of pV system with smart grid have been examined and the researcher proposed the line conditioner to extract the maximum power point tracking (MPPT) of PV system, for this purpose neuro fuzzy controller has been designed in [221].

A modified African Buffalo Optimization (MABO) has designed to control the two stage frequency among interconnecting many microgrid. It's performance has been checked by varying different parameters such as variation in voltage and disturbance on load. The enhancement and stability of grid has been proved by connecting electric vehicle to the grid [222]. The effect of electric vehicle in term of pros and cons on grid under various circumstances have been recorded in [223]. The different factors such as feasibility and cost of standalone and hybrid grid connected system have been analysed. The Particle Swarm Optimization (PSO) and Exchange Market Algorithm (EMA) have been proposed to mitigate the power losses and optimize the size along with better locations in [224]. The multi objective predictive energy management algorithm has been proposed for the HESS of residential grid connected system. The designed algorithm is based on theroy of machine learning [225]. The Swine Influenza Model Based Optimization with Quarantine (SIMBO-O) has been suggested to minimize the total cost of operations of grid along with battery storage system. The suggested algorithm based of probability of quarantine and treatment loop. This technology can be useful to optimize the total cost of super grid infrastructure [226].

Shunt VARs Compensators (SVCs) and Thyristor Controlled Series Compensators (TCSCs) have been suggested instead of Flexible AC Transmission Systems (FACTS) device due to its high cost. Particle Swarm Optimization (PSO) has been considered to optimize the location and size of device for stability and flexibility of grid [227]. A computational methods has employed to analyse the supply and demand response in distributed network (DR). Its proved that the quantum algorithms shows better performance in DR among overall computational methods [228]. The Fig. (46) represents the categories of analysis of power system along with classification of probabilistic technique, which may be useful for long transmission in super grid to minimize the cost along with other necessary factors. The different approaches have been discussed to analyze the modeling of super grid infrastructure. The modeling with uncertainty would be the better choice as compared to deterministic due to external perturbation and renewable energy resources in super grid system. However, it's also important for the management of energy system in grid infrastructure [229]. Furthermore, the uncertainty modeling can be done through possibilistic, probabilistic, optimization and analysis of interval.

#### 2) ARTIFICIAL INTELLIGENCE

The artificial intelligence plays a major role in super grid infrastructure, for the communication purpose between various clusters. Different artificial intelligence techniques have been proposed for renewable energy and smart grids [230]. The fault detection in smart grid can be detected by using concept of AI. The application of AI in smart grid has been noted and reviewed the control mechanism for smart grid using AI approach. The shunt hybrid filters (SHF) has suggested for enhancement of power quality in smart grid [231]. Adaptive Fuzzy Neural Network (AFNN) has employed to cater the various objectives in smart grid system at different load conditions [232]. Artificial neural networks based on machine learning has proposed solutions for many wireless communication problem in lights of pron and cons [233]. The self building and self learning of AI based on machine learning has been explored and mentioned the advantages and disadvantages of self learning algorithm along its application in smart cities [234].

The concept of super grid and super block concept have been suggested for china cities. Super grid consists of a number of cell called super block. The researchers explored the modern structure and future advancement in super block connectivity for application in smart cities [235]. The effect of AI in practical life has been studied that how the life affected by AI when coming in real world. The huge amount of data has been collected from smart cities by employing machine learning algorithms. It's labeled the wasted data and proposed the three level strategy to convert these wasted data into efficient data [236], [237]. The deep learning algorithms has been implemented for detection of wrong way driving. The detection of way can be automatically detected by rotating the camera, to reduces the chances of accident [238]. The different approach of machine learning have been proposed, to handle the problems related to energy management strategy by introducing electric vehicles charging system in smart cities [239].

The different research area of AI have been mentioned such as power equipment based on AI algorithm, management of power and Image recognition technology. The AI is combined with big data and IoT to improve the performance of power system [240]. The different applications of AI has been discussed such as production of power along with distribution and transmission, energy saving methods having proper policy to controls its consumption. The scholars also mentioned the proper way of communication for



FIGURE 46. Power system analysis based on data.

input output data along with display of output [241]. The artificial neural networks has been suggested to evaluate the heat required for non residential building in European countries during winter season [242]. The AI technique has been proposed for better economics situation and job creativity in developing countries, this concept can be extended for communication between developed and developing countries in term of power system such as super grid system.

In India the emission of  $CO_2$  has been predicted by employing ANN. its proved the better economic growth along with enhancement in energy sector [243]. The major role of AI has been highlighted in the field of business. The researchers show that how the business problem has been solved by implementing the concept of AI. It's expressed that The impact of AI and automation on different places and people have been studied in [245]. The application of AI in the filed of education specially tutoring and assessment have been explained in [246]. The relationship between AI and sustainable development have been studied in [247]. it's concluded that sustainable development is based on AI. The application of AI concept between online and offline application is presented in Fig. (47). It shows the AI operation on both cases online and offline. The online operation is based on online data that has taken from different IoT devices. The GPS has communicated with phasor measurement unit (PMU) to provide data to phasor data concentrator (PDC) in a wide area measurement system (WAMS). Real-time monitoring and control system

the 50 different companies take help from AI domain [244].

checked the accuracy and type of data before providing to AI toolbox.

While the offline communication of AI evaluating the prediction that depend on training data which is further based on simulated and previous data of the system.

#### **B. OTHER METHODS FOR PLANNING OF SSG SYSTEM**

In this section we have reviewed a different typologies that could be used in infrastructure of super smart grid system.

The United State (U.S) Caribbean super grid has proposed to linked Caribbean island to power grid of U.S via High Voltage Direct Current (HVDC). it's considered the power profile as a tool for estimation of parameters and voltage that fulfill the requirement on both receiving and transmission sides [248]. The concept of both super grid and super block have been defined. It explored the connection from radial and grid to super grid and super block respectively [249]. The term global super grid is referred to the system in which overall electrical power is based on renewable energy. The researchers studied the operational and blockage factors that effects the global super grid [250]. A plan for implementing of super grid in 2023 has been discussed among five countries of Northeast Asia (NEA). The optimal network structure has been taken into account for Myerson (1977) value to shows the advantages of Myerson in SSGs [251]. The infrastructure of Australian grid has been linked with Asian grid named Australian-Asian (Power) Grid (AAG) by using super grid theory. It's also discussed the consequences of super grid in proposed AAG system [252]. The contribution of photovoltaic (PV) system in Association of Southeast Asian Nations (ASEAN) lead to the sustainable development by using SWOT method [253]. The transmission of solar energy from high concentration region to low concentration region in Turkey has been examined. The analysis of total cost for flow of energy has been calculated for HVDC transmission [254].

The Japan-Taiwan-Philippines HVDC transmission and its connection to Asian super grid along with technical challenges has been reviewed. The analytical framework has presented that how to become Australian grid system is a large exporter of renewable energy to the Asian grid. It's also considered the implementation of policy due to variable geopolitics situation of energy [255]. The analysis from 2015 to 2050 has been done for the transmission of renewable energy in Philippines. The linear optimisation tool has been proposed to consider variable renewable sources such as PV and batteries [256]. The comparison between HVDC and HVAC have been done and the components required for the HVDC has been discussed such as Voltage Source Converters (VSCs) and Line Commutated Converters (LCCs) [257]. The emerging of photovoltaic panel with grid has been explored in term of pros and cons. The scenario for power on both sides of supply and demand have been analysed [210]. The coordinated controller has suggested for the HVDC of the wind and battery energy storage system (BESS) to provide flexibility to the power system [258].

The conversion of HVDC system into multiterminal HVDC (MT-HVDC) grids has been explained with merging of renewable energy sources in MT-HVDc grids along with other challenging issues that have been discussed in [259]. The power coming from Distributed Energy Resources (DERs) plays important role in improvement of voltage profile to reduces the losses. Different optimization techniques have been proposed for finding the condition point of DERs [260]. The challenges that faced by transmission of HVDC with integration of other sources in national grid of Pakistan. It's proved that the crisis of energy in Pakistan can be minimized by using proposed HVDC technology [261]. In a MT HVDC structure a Mayfly Algorithm (MA) has proposed to increase the control efficiency of voltage source converter (VSC) by designing its parameters [262].

The concept of regional power connectivity has been explained through out Asia. The relationship of Asia and China has been explained in term of power connection [263]. The Europe 2020 infrastructure has been presented for development and prosperity of people, which lead to be sustainable growth of Europe country [264]. The zero  $CO_2$  emission has been adapted by Australia country due to greater renewable sources and is exported to Asian country [265].

The solution for maximum power transfer across the boundaries of the country in Europe and Latin America. To reduce the transmission losses Ultra High Voltage Alternating Current (UHVAC) having voltage value is above  $800 \, kV$  has been suggested [266]. The installation of HVDC grid of Europe along with Challenging tasks and advantages of HVDC have been discussed in [267], and proved that this will lead to maintain a union in Europe countries in term of electric power. The advantages of renewable energies to stabilize the economic situation of the country has bee discussed in [268]. The effect of political leadership of China and Russia on Asian super grid has been noted, which is main reason on the super grid infrastructure to share the power from both side to make the economy strong [269].

The researcher presented the green energy based on the first step of Northeast Asian Super Grid having less effect on environmental pollution. The electric vehicle play main role in reduction of  $CO_2$  emission. The vehicle to grid (V2G) mode in electric vehicle application act as a back up source to provide power to the load during its peak demand. It support and stabilize the smart grid to keep better voltage profile under varying load conditions [270]. The effect of V2G technology has been studied in Modern Distribution Network (MDW). The reliability tool has been suggested to measure the reliability of MDW including V2G [271].

The plug in electric vehicle (PEV) consist of charging unit and HESS. The integration of different energy storage devices communication is depicted in Fig. (48) which is proposed for super grid infrastructure along with V2G concept. The Fig. (48) describes the flow of energy between different energy management system by using a common interface, that control and manage the flow of energy between various system. The various system such as energy management sys-



FIGURE 47. AI architecture model both online and off line.



FIGURE 48. Communication of energy between storage devices.

tem (EMS), device management system (DMS) and power conversion system (PCS) are interconnected with user by a common communication interface. This topology is helpful by designing a hybrid energy management system in SSGs [272]. A linear programming algorithm has proposed for the DC power flow in both directions in charging and

discharging mode of the battery. The solar panel and wind are considered to be DC source in the smart grid for the charging of the PEV battery [273]. Advanced Metering Infrastructure (AMI) has been installed in America for controlling of price. This technology has also lead to the smart grid system in the future [274]. A dual stage optimization technique based on day ahead optimization (DAO) and real time optimization (RTO) have been proposed for measuring charging and discharging time in V2G application at an office building [275]. The 5G V2G system has presented to emerged the 5G network with power grid to support far transmission network. The lightweight algorithms has employed for the security and privacy purpose in [276]. The different components of probabilistic load flow (PLF) have been discussed which have useful for the generation and distribution of PV power along with charging of EV [277]. The charging of electric vehicles may be possible by introducing Inductively Coupled Power Transfer (ICPT) technology. The circuitry parameters for ICPT have been designed for proper charging process [278], [279]. The technique for DC fast charging of electric vehicle has proposed and its impact on smart grid. A Frequency Containment Reserves (FCR) has employed for the frequency stabilization of power grid. The FCR affected the life of the battery [280]. In [281] proposed the FCR for both cases with and without battery in V2G application. Rain flow Counting (RFC) method has implemented to calculate the degradation factor.

The structure of super grid in china and US have been explored in [282]. The super grid in china consist of ultra high voltage alternating current (UHVAC) and line commutated converters high voltage direct current (LCC-HVDC), while the super grid in US consist of voltage source converter VSC HVDC [282]. The integration of smart African Super Grid system which are based on five African Power pools. A complicated controller has been designed for integration of these pools [283]. The effectiveness of Iraqi super grid has been explored by adding a solar power plant having capacity 1000MW [284]. The HVDC and HVAC links have been introduced for the comparison analysis and proposed a better solution to increase its performance. The stability of the system has been checked by performing the line and bus tests. The importance of power quality can not be ignored by proposing the structure of super grid because it directly impact the consumer in the form of voltage sag etc... [284]. The voltage sag is the major problem in power system that can be reduced by considering the CPDS such as DVR. Custom Power Devices (CPDs) have been for the distribution network to overcome the above mentioned problems. The Dynamic Voltage Restorer (DVR) is used to overcome the voltage sag occurring due to the unbalanced power grid [285]. The DVR has proposed for the distribution system due to its many advantages such as it keep the consumer online with maintaining high quality of voltage constant [286].

A fast decouple techniques has performed in power system tool to model the 330KV existing network along with Meshed

330kV super grid transmission line for the Nigerian system. It performed to reduces the transmission losses and increases the efficiency of the system [287]. The stability of bus has been checked by analysing fault on Benin IkejaWest Aiyede Oshogbo Benin (BIAOB) 330KV transmission connected to Nigerian grid. It's concluded that the reactive power is present on all buses while the lower value of voltage exist at Line-Line-Line-Ground (L-L-G) bus. Symmetrical components is the main part of transmission line [288], the scholars suggested that how the symmetrical components of transmission line is effected from tower structure along with the resistivity of ground. The resistivity of ground plays a major contribution on structure of tower according to locations [289]. The upgrading of radial 330KV to ring 750KV super grid system has been proposed for Nigerian grid because the ring system is low losses and more reliable than radial system. The Newton-Raphson Algorithm has employed to convert the existing 330KV to 750KV system [290]. The evaluation of bus voltage and efficiency of reactive compensation is carried out on 330kV networks of Ikeja West Transmission Company. It's observed that the station reactor greatly change the profile of voltage to improve the quality of power [291]. The researcher designed the electrical parameters along with mechanical structure of 750KV system for transmission of Extra High Voltage (EHV) to optimize the cost as well as the losses as compared to 330KV system in Nigeria [292].

The scholars proposed various strategies to converter smart grid system into Internet of Energy (IOE) for various communication purpose. The word IoE has used to represent companies and industries that need energy from the grid. It makes the decentralized system in which peer to peer communication possible through block chain technique [293]. The various techniques for saving of energy by utilities through block chain network in grid structure have been suggested in [294]. The wireless communication between different sections of grid and its application in future super smart grid have been explained, in which Various types of energies have emerged with each other by proposing the idea of Internet of Energy (IoE) like V2G technology, smart grid etc... [295]. The various application of block chain has been carried in modern grid system that make the grid is more smart and reliable along with enhancement of energy resources to fulfill the requirement of load at any condition due to fast communication that made possible by block chain [296].

The nonlinear controller has been proposed for the integration of renewable energy in DC grid having stable voltage and power profile to support frequency as just like in AC grid [302]. A nonlinear Backstepping controller has been suggested for the single phase grid connected PV system. It includes the three phase inverter and DC-DC converter. The controller designed for each converter is separately [300]. The supertwisting algorithm sliding mode controller (STASMC) has been proposed for the voltage control mode (VCM) and current control mode (CCM) of grid connected renewable

Year	Hardware	Communication Models	Security Protocols	Artificial Intelligence	Energy Management (G2V and V2G modes)	Controller Design	Optimization	Reference
2019	X	X	Х	X	$\checkmark$	Х	X	[269], [286], [62], [192]
2019	X	$\checkmark$	X	X	X	$\checkmark$	X	[143], [145]
2019	X	X	X	√	X	X	X	[243]–[245], [297]
2020	X	X	X	$\checkmark$	X	X	$\checkmark$	[214]–[230], [298]
2020	X	X	X	X	$\checkmark$	$\checkmark$	X	[260], [284], [191]
2021	X	X	X	X	$\checkmark$	X	$\checkmark$	[216], [223], [195], [196]
2021	X	$\checkmark$	X	$\checkmark$	X	X	X	[231], [263]
2021	X	$\checkmark$	X	X	X	X	X	[173], [233], [295], [299]
2022	X	X	X	X	$\checkmark$	X	$\checkmark$	[154], [228], [282]
2022	X	√	X	X	X	$\checkmark$	X	[144], [168], [300]
2023	X	X	X	√	$\checkmark$	X	X	[222], [237]
2023	X	X	$\checkmark$	$\checkmark$	X	X	$\checkmark$	[123], [124], [271], [275]
2023	$\checkmark$	X	X	X	X	$\checkmark$	X	[298], [300], [301]
2023	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	[This Work]

#### TABLE 2. Comparison with existing studies.

sources having single phase inverter [303]. The Adaptive supertwisting Controller has been designed to regulate the value of voltage to its required reference. The chattering effect has also minimized by proposed controller which decrease the losses in the system [304].

The proper Energy management systems (EMSs) has been introduced for the storage of energy from different renewable sources in grid. The optimization technique has proposed for the better usage of storage energy in grid. The integration of different renewable sources for transmission and distribution in smart or super grid is illustrated in Fig. (49) [305]. The fading Kalman filter algorithm has been developed for the communication in both way having far distance transmission of smart grid network. The proposed algorithm can also be used in super grid infrastructure for possible communication [306]. The voltage compensator and harmonic elimination technique have proposed for the reduction of sag in voltage and control of voltage profile respectively. The THD is lower than 5 % The lightening is very dangerous event for the electrical power system which may damage the components of power system, therefore the researchers designed surge arrester to arrest the lightening [308].

The effective way for controlling of power management has been discussed in developed country which should be to able to handle the control of power during power management [309]. The regulations of DC bus voltage in DC micro grid can be improved by proposed voltage modulated direct power control method. From the inner loop proved the stability of the system while the outer loop helps to regulate

VOLUME 11, 2023

the DC bus voltage. The energy can be import and export to grid by using electric vehicle in G2V and V2G mode [310]. The Energy Monitoring System (EMS) in electric vehicle has been proposed to minimize the consumption of fuel along with reduction of  $CO_2$  gas [311]. The computational intelligence (CI) technique has been suggested for the refrigeration system to increase the efficiency in order to reduce the heat losses and decreases the cost of the system. The proposed method has been compared with other AI technique such as ANN to proof the robustness of proposed algorithm [312].

The performance of photovoltaic system has been tested on small, medium and large scales according to the different locations and climate change in the world. The authors proposed seven different PV technologies for different weather conditions. The electrolysis of sea is used to provide green hydrogen by proposing a marine renewable energies [313]. It's proved that the proton exchange membrane electrolysis at sea level has best choice for gaining green hydrogen from the marine renewable energies [314]. Many methods have been described for the green and clean production of hydrogen at large scale in Morocco's. it's concluded that the electrolysis of alkaline earth water with renewable energy is best choice for generation of hydrogen in Morocco [315].

A dual-stage robust frequency risk-constrained unit commitment (FRCUC) has proposed for the smooth operation of system. The frequency violation risk has been considered to adapt the stability of the system. The proposed model has suggested for the HVDC system Which represented the better performance [301]. The comparison between



FIGURE 49. Integration of different renewable sources.

self-excited induction generator (SEIG) and permanent magnet synchronous generator (PMSG) have been highlighted by considering its application with renewable energy source. The performance of these both generator practically checked by using with micro hydro-turbine [316]. The parallel operation of Self-Excited Induction Generator (SEIG) for mini power generation in standalone. They have been tested experimentally without using any power electronic components. Capacitor bank has been taken online for excitation purpose of generator [317]. The three-phase Squirrel Cage Induction Machine (SCIM) has been used to create the setup for the single phase power through two capacitor methods in rural areas of India [318].

The relationship among renewable, non renewable energy, GDP and emissions of  $CO_2$  have been studied in Vietnam. The wavelets analysis has been suggested for evaluating this relationship [319]. The world investment energy 2017 explained the way of decision taken by government, energy sector along with other financial departments. The issue related to energy has also explored for 2017 and beyond it [320].

The mechanism of HVDC has been discussed along with the control strategy for DC and AC system have also explored with help of DC circuit breaker [321]. The electromagnetic transient has detected by using Multi Level Converter (MPC) instead of half bridge circuit. The accuracy of proposed MPC has been checked under operating different fault condition [322]. The equivalent model of multilevel converter (MPC) for Hardware In Loop (HIL) simulation has been developed. HIL has proposed to the check the performance of converter in real time scenario, which has been employed for the project in China in [323]. The equivalent model of multilevel converter (MPC) has proposed for the multi terminal DC (MTDC) system. The proposed model and system have based on HVDC breaker [324]. The different social factor and risk have been mentioned that effect the infrastructure of super grid in Europe and US. A latest policy has been created to analyze and develop the structure of super grid [325]. The transmission system of energy in future (2040) has been explored for the Europe. The optimization tool has applied to defined the future transmission system based proposed European zonal model [326]. The renewable energy source

like wind and solar have been optimized in term of cost based on weather data that can be applicable for year 2030 [327]. The Australia has sufficient source of energy that has fulfill its requirements from the fossil fuel. Therefore Australia grid export energy to Asian grid called Australian-Asian Grid (AAG). The connection of grid and green energy is the major step to lead towards the sustainability via Global Energy Interconnection (GEI) platform [252]. The highly highlighted the following fields such as smart grid, high voltage transmission system and interconnection of grid for the global world in the future [328]. The concept of Global grid has been studied that all grid system connected in global to one transmission system [329]. The green energy from renewable resources reduces the challenging task to Global grid transmission system. The various feature has defined that used the idea of Global grid in reality [330]. The socio-technical experiments has based on four various steps of energy transition that lead to energy democracy in South Africa [331]. The fault current for HVDC circuit breaker has proposed to interrupt the current in HVDC system. The procedure for fault current and its interruption have been measured by using different switching tests [332]. The low frequency alternating current (LFAC) has been proposed for transmission of wind energy for long distance. The technical issue related to LFAC has been discussed to evaluate the range of the distance [333]. The control mechanism of multilevel matrix converter has been noted under unbalanced grid condition [334]. A decoupled control algorithm has been designed for low frequency transmission system using multilevel converter. The current has been transferred from dq axis to DC signal to show the better performance [335]. The methods for implementation global grid has been proposed and it's concluded that HVDC is a best option for transmission of high voltage through long distance [336]. The concept of electrification has been introduced to reduces the deficiency of energy in Africa. It's proposed the renewable energy for transmission and distribution for reliability of the super grid [337]. The the West African power system model has proposed to model financial electric market at which supply the electricity outside the border. It will estimate the cost of generation and other factors [338].

In nutshell, we have studied many articles in the literature that summarized in Table [2].

#### **XII. COMPARISON WITH EXISTING SOLUTIONS**

In this section we have presented a comparative analysis of proposed work with existing literature in Table [2]. It analyzes the reviews on the presence of various parameters that are depicted in tabular form. The proposed review article cover all topics that supports the infrastructure of super grid to minimize the crises of energy through globe.

#### REFERENCES

 J. W. Feltes, B. D. Gemmell, and D. Retzmann, "From smart grid to super grid: Solutions with HVDC and FACTS for grid access of renewable energy sources," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2011, pp. 1–6.

- [3] M. Tariq and H. V. Poor, "Electricity theft detection and localization in grid-tied microgrids," *IEEE Trans. Smart Grid*, vol. 9, no. 3, pp. 1920–1929, May 2018.
- [4] A. Battaglini, J. Lilliestam, A. Haas, and A. Patt, "Development of Super-Smart grids for a more efficient utilisation of electricity from renewable sources," *J. Cleaner Prod.*, vol. 17, no. 10, pp. 911–918, Jul. 2009.
- [5] N. Acharya and C.-C. Liu, "Tripping of wind turbines during a system fault," in Proc. IEEE Power Energy Soc. Gen. Meeting, Convers. Del. Electr. Energy 21st Century, Jul. 2008, pp. 1–8.
- [6] H. Bulat, D. Franković, and S. Vlahinić, "Enhanced contingency analysis—A power system operator tool," *Energies*, vol. 14, no. 4, p. 923, Feb. 2021.
- [7] M. Adnan and M. Tariq, "Cascading overload failure analysis in renewable integrated power grids," *Rel. Eng. Syst. Saf.*, vol. 198, Jun. 2020, Art. no. 106887.
- [8] C. Cambini, R. Congiu, T. Jamasb, M. Llorca, and G. Soroush, "Energy systems integration: Implications for public policy," *Energy Policy*, vol. 143, Aug. 2020, Art. no. 111609.
- [9] M. Z. Zakariya and J. Teh, "A systematic review on cascading failures models in renewable power systems with dynamics perspective and protections modeling," *Electr. Power Syst. Res.*, vol. 214, Jan. 2023, Art. no. 108928.
- [10] S. Gordon, "Supergrid to the rescue," *Power Eng.*, vol. 20, no. 5, pp. 30–33, 2006.
- [11] H. Ergun, J. Beerten, and D. Van Hertem, "Building a new overlay grid for Europe," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2012, pp. 1–8.
- [12] A. Purvins, H. Wilkening, G. Fulli, E. Tzimas, G. Celli, S. Mocci, F. Pilo, and S. Tedde, "A European supergrid for renewable energy: Local impacts and far-reaching challenges," *J. Cleaner Prod.*, vol. 19, nos. 17–18, pp. 1909–1916, Nov. 2011.
- [13] M. B. Blarke and B. M. Jenkins, "SuperGrid or SmartGrid: Competing strategies for large-scale integration of intermittent renewables?" *Energy Policy*, vol. 58, pp. 381–390, Jul. 2013.
- [14] J. De Decker, J. Tambke, J. Völker, and K. Michalowska-Knap, "An offshore transmission grid for wind power integration: The European techno-economic study OffshoreGrid," in *Proc. IEEE PES Gen. Meeting*, Jul. 2010, pp. 1–8.
- [15] D. Van Hertem, M. Ghandhari, and M. Delimar, "Technical limitations towards a supergrid—A European prospective," in *Proc. IEEE Int. Energy Conf.*, Dec. 2010, pp. 302–309.
- [16] J. Torriti, "Demand side management for the European supergrid: Occupancy variances of European single-person households," *Energy Policy*, vol. 44, pp. 199–206, May 2012.
- [17] M. Düren, "Clean power from deserts," *Green*, vol. 1, nos. 3–4, pp. 263–275, 2011. Accessed: Jul. 18, 2023, doi: 10.1515/green. 2011.025.
- [18] A. Woyte, J. De Decker, and V. Van Thong, "A North Sea electricity grid (r) evolution. Electricity output of interconnected offshore wind power integration," 2008.
- [19] L. Ponta, M. Raberto, A. Teglio, and S. Cincotti, "An agent-based stockflow consistent model of the sustainable transition in the energy sector," *Ecol. Econ.*, vol. 145, pp. 274–300, Mar. 2018.
- [20] R. K. Ahir and B. Chakraborty, "A novel cluster-specific analysis framework for demand-side management and net metering using smart meter data," *Sustain. Energy, Grids Netw.*, vol. 31, Sep. 2022, Art. no. 100771.
- [21] I. Mechelen, J. Hampton, R. S. Michalski, and P. Theuns, "Categories and concepts: Theoretical views and inductive data analysis," Jan. 1993. [Online]. Available: https://blogs.sas.com/content/sgf/2017/07/17/whatare-the-characteristics-of-true-clusters/
- [22] M. Adnan, M. Tariq, Z. Zhou, and H. V. Poor, "Load flow balancing and transient stability analysis in renewable integrated power grids," *Int. J. Electr. Power Energy Syst.*, vol. 104, pp. 744–771, Jan. 2019.
- [23] B. Arockiasamy and P. J. Raj, "A future of the electrical energy system using smart grid," *Int. Res. J. Modernization Eng. Technol. Sci.*, vol. 5, no. 3, 2023. [Online]. Available: https://www.doi.org/10.56726/ IRJMETS35185, doi: 10.56726/IRJMETS35185.
- [24] T. Alsuwian, A. Basit, A. A. Amin, M. Adnan, and M. Ali, "An optimal control approach for enhancing transients stability and resilience in super smart grids," *Electronics*, vol. 11, no. 19, p. 3236, Oct. 2022.

- [25] M. Azeroual, A. E. Makrini, H. E. Moussaoui, and H. E. Markhi, "Renewable energy potential and available capacity for wind and solar power in Morocco towards 2030," *J. Eng. Sci. Technol. Rev.*, vol. 11, no. 1, pp. 189–198, Feb. 2018.
- [26] M. Azeroual, T. Lamhamdi, H. E. Moussaoui, and H. E. Markhi, "Simulation tools for a smart grid and energy management for microgrid with wind power using multi-agent system," *Wind Eng.*, vol. 44, no. 6, pp. 661–672, Dec. 2020.
- [27] R. Bayindir, I. Colak, G. Fulli, and K. Demirtas, "Smart grid technologies and applications," *Renew. Sustain. Energy Rev.*, vol. 66, pp. 499–516, Dec. 2016.
- [28] S. Jinlei, P. Lei, L. Ruihang, M. Qian, T. Chuanyu, and W. Tianru, "Economic operation optimization for 2nd use batteries in battery energy storage systems," *IEEE Access*, vol. 7, pp. 41852–41859, 2019.
- [29] S. Blumsack, P. Hines, M. Patel, C. Barrows, and E. C. Sanchez, "Defining power network zones from measures of electrical distance," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2009, pp. 1–8.
- [30] A. Moreira, D. Pozo, A. Street, and E. Sauma, "Reliable renewable generation and transmission expansion planning: Co-optimizing system's resources for meeting renewable targets," *IEEE Trans. Power Syst.*, vol. 32, no. 4, pp. 3246–3257, Jul. 2017.
- [31] E. E. Sauma and S. S. Oren, "Proactive planning and valuation of transmission investments in restructured electricity markets," *J. Regulatory Econ.*, vol. 30, no. 3, pp. 358–387, Dec. 2006.
- [32] F. D. Munoz, B. F. Hobbs, and S. Kasina, "Efficient proactive transmission planning to accommodate renewables," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2012, pp. 1–7.
- [33] A. H. van der Weijde and B. F. Hobbs, "The economics of planning electricity transmission to accommodate renewables: Using two-stage optimisation to evaluate flexibility and the cost of disregarding uncertainty," *Energy Econ.*, vol. 34, no. 6, pp. 2089–2101, Nov. 2012.
- [34] M. Madrigal and S. Stoft, "Transmission expansion for renewable energy scale-up: Emerging lessons and recommendations," 2012.
- [35] V. Krishnan, J. Ho, B. F. Hobbs, A. L. Liu, J. D. McCalley, M. Shahidehpour, and Q. P. Zheng, "Co-optimization of electricity transmission and generation resources for planning and policy analysis: Review of concepts and modeling approaches," *Energy Syst.*, vol. 7, no. 2, pp. 297–332, May 2016.
- [36] F. D. Munoz, B. F. Hobbs, J. L. Ho, and S. Kasina, "An engineeringeconomic approach to transmission planning under market and regulatory uncertainties: WECC case study," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 307–317, Jan. 2014.
- [37] J. H. Roh, M. Shahidehpour, and Y. Fu, "Market-based coordination of transmission and generation capacity planning," *IEEE Trans. Power Syst.*, vol. 22, no. 4, pp. 1406–1419, Nov. 2007.
- [38] J. D. Lyon, K. W. Hedman, and M. Zhang, "Reserve requirements to efficiently manage intra-zonal congestion," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 251–258, Jan. 2014.
- [39] Y. Chen, P. Gribik, and J. Gardner, "Incorporating post zonal reserve deployment transmission constraints into energy and ancillary service co-optimization," *IEEE Trans. Power Syst.*, vol. 29, no. 2, pp. 537–549, Mar. 2014.
- [40] M. Tariq and M. Adnan, "Stabilizing super smart grids using V2G: A probabilistic analysis," in *Proc. IEEE 89th Veh. Technol. Conf. (VTC-Spring)*, Apr. 2019, pp. 1–5.
- [41] T. Harighi, R. Bayindir, S. Padmanaban, L. Mihet-Popa, and E. Hossain, "An overview of energy scenarios, storage systems and the infrastructure for vehicle-to-grid technology," *Energies*, vol. 11, no. 8, p. 2174, Aug. 2018.
- [42] L. F. A. Roman, P. R. L. Gondim, and J. Lloret, "Pairing-based authentication protocol for V2G networks in smart grid," *Ad Hoc Netw.*, vol. 90, Jul. 2019, Art. no. 101745.
- [43] M. Aziz, T. Oda, T. Mitani, Y. Watanabe, and T. Kashiwagi, "Utilization of electric vehicles and their used batteries for peak-load shifting," *Energies*, vol. 8, no. 5, pp. 3720–3738, Apr. 2015.
- [44] Y. Li, P. Zhang, and Y. Wang, "The location privacy protection of electric vehicles with differential privacy in V2G networks," *Energies*, vol. 11, no. 10, p. 2625, Oct. 2018.
- [45] S. Chatzivasileiadis, T. Krause, and G. Andersson, "Supergrid or local network reinforcements, and the value of controllability—An analytical approach," in *Proc. IEEE Grenoble Conf.*, Jun. 2013, pp. 1–6.
- [46] A. A. Eajal, E. F. El-Saadany, and K. Ponnambalam, "Equal power sharing in islanded AC/DC hybrid microgrids," in *Proc. IEEE Electr. Power Energy Conf. (EPEC)*, Oct. 2016, pp. 1–6.

- [47] S. Sun, M. Dong, and B. Liang, "Distributed real-time power balancing in renewable-integrated power grids with storage and flexible loads," *IEEE Trans. Smart Grid*, vol. 7, no. 5, pp. 2337–2349, Sep. 2016.
- [48] S. Gill, I. Kockar, and G. W. Ault, "Dynamic optimal power flow for active distribution networks," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 121–131, Jan. 2014.
- [49] Y. Levron, J. M. Guerrero, and Y. Beck, "Optimal power flow in microgrids with energy storage," *IEEE Trans. Power Syst.*, vol. 28, no. 3, pp. 3226–3234, Aug. 2013.
- [50] M. Fan, V. Vittal, G. T. Heydt, and R. Ayyanar, "Probabilistic power flow analysis with generation dispatch including photovoltaic resources," *IEEE Trans. Power Syst.*, vol. 28, no. 2, pp. 1797–1805, May 2013.
- [51] M. Aien, M. Fotuhi-Firuzabad, and M. Rashidinejad, "Probabilistic optimal power flow in correlated hybrid wind–photovoltaic power systems," *IEEE Trans. Smart Grid*, vol. 5, no. 1, pp. 130–138, Jan. 2014.
- [52] M. N. Kabir, Y. Mishra, and R. C. Bansal, "Probabilistic load flow for distribution systems with uncertain PV generation," *Appl. Energy*, vol. 163, pp. 343–351, Feb. 2016.
- [53] D. Seo, H. Lee, and A. Perrig, "Secure and efficient capability-based power management in the smart grid," in *Proc. IEEE 9th Int. Symp. Parallel Distrib. Process. Appl. Workshops*, May 2011, pp. 119–126.
- [54] A. G. Tsikalakis and N. D. Hatziargyriou, "Centralized control for optimizing microgrids operation," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2011, pp. 1–8.
- [55] K. T. Tan, X. Y. Peng, P. L. So, Y. C. Chu, and M. Z. Q. Chen, "Centralized control for parallel operation of distributed generation inverters in microgrids," *IEEE Trans. Smart Grid*, vol. 3, no. 4, pp. 1977–1987, Dec. 2012.
- [56] H. Xin, Z. Qu, J. Seuss, and A. Maknouninejad, "A self-organizing strategy for power flow control of photovoltaic generators in a distribution network," *IEEE Trans. Power Syst.*, vol. 26, no. 3, pp. 1462–1473, Aug. 2011.
- [57] M. C. Chandorkar, D. M. Divan, and R. Adapa, "Control of parallel connected inverters in standalone AC supply systems," *IEEE Trans. Ind. Appl.*, vol. 29, no. 1, pp. 136–143, 1993.
- [58] G. Mokhtari, G. Nourbakhsh, and A. Ghosh, "Smart coordination of energy storage units (ESUs) for voltage and loading management in distribution networks," *IEEE Trans. Power Syst.*, vol. 28, no. 4, pp. 4812–4820, Nov. 2013.
- [59] S. Weckx and J. Driesen, "Load balancing with EV chargers and PV inverters in unbalanced distribution grids," *IEEE Trans. Sustain. Energy*, vol. 6, no. 2, pp. 635–643, Apr. 2015.
- [60] C. Wu, F. Wen, Y. Lou, and F. Xin, "Probabilistic load flow analysis of photovoltaic generation system with plug-in electric vehicles," *Int. J. Electr. Power Energy Syst.*, vol. 64, pp. 1221–1228, Jan. 2015.
- [61] W. Chen, H. Yan, X. Pei, and B. Wu, "Probabilistic load flow calculation in distribution system considering the stochastic characteristic of wind power and electric vehicle charging load," in *Proc. IEEE PES Asia–Pacific Power Energy Eng. Conf. (APPEEC)*, Oct. 2016, pp. 1861–1866.
- [62] R. Deng, Z. Yang, M.-Y. Chow, and J. Chen, "A survey on demand response in smart grids: Mathematical models and approaches," *IEEE Trans. Ind. Informat.*, vol. 11, no. 3, pp. 570–582, Jun. 2015.
- [63] N. Ul Hassan, Y. I. Khalid, C. Yuen, and W. Tushar, "Customer engagement plans for peak load reduction in residential smart grids," *IEEE Trans. Smart Grid*, vol. 6, no. 6, pp. 3029–3041, Nov. 2015.
- [64] W. Tushar, B. Chai, C. Yuen, D. B. Smith, K. L. Wood, Z. Yang, and H. V. Poor, "Three-party energy management with distributed energy resources in smart grid," *IEEE Trans. Ind. Electron.*, vol. 62, no. 4, pp. 2487–2498, Apr. 2015.
- [65] S. Maharjan, Q. Zhu, Y. Zhang, S. Gjessing, and T. Basar, "Dependable demand response management in the smart grid: A Stackelberg game approach," *IEEE Trans. Smart Grid*, vol. 4, no. 1, pp. 120–132, Mar. 2013.
- [66] O. Kilkki, A. Alahäivälä, and I. Seilonen, "Optimized control of pricebased demand response with electric storage space heating," *IEEE Trans. Ind. Informat.*, vol. 11, no. 1, pp. 281–288, Feb. 2015.
- [67] Y. Liu, C. Yuen, S. Huang, N. U. Hassan, X. Wang, and S. Xie, "Peakto-average ratio constrained demand-side management with consumer's preference in residential smart grid," *IEEE J. Sel. Topics Signal Process.*, vol. 8, no. 6, pp. 1084–1097, Dec. 2014.
- [68] Y. Liu, C. Yuen, R. Yu, Y. Zhang, and S. Xie, "Queuing-based energy consumption management for heterogeneous residential demands in smart grid," *IEEE Trans. Smart Grid*, vol. 7, no. 3, pp. 1650–1659, May 2016.

- [69] T. Ma, J. Zhou, M. Tang, Y. Tian, A. Al-Dhelaan, M. Al-Rodhaan, and S. Lee, "Social network and tag sources based augmenting collaborative recommender system," *IEICE Trans. Inf. Syst.*, vol. E98.D, no. 4, pp. 902–910, 2015.
- [70] S. Xie and Y. Wang, "Construction of tree network with limited delivery latency in homogeneous wireless sensor networks," *Wireless Pers. Commun.*, vol. 78, no. 1, pp. 231–246, Sep. 2014.
- [71] M. Dohler, S. Fortes, I. Haque, S. J. Johnson, and T. Maksymyuk, "Internet of Things," *IEEE Commun. Mag.*, vol. 61, no. 5, pp. 120–121, May 2023.
- [72] Z. Xia, X. Wang, X. Sun, and Q. Wang, "A secure and dynamic multikeyword ranked search scheme over encrypted cloud data," *IEEE Trans. Parallel Distrib. Syst.*, vol. 27, no. 2, pp. 340–352, Feb. 2016.
- [73] Z. Fu, X. Sun, Q. Liu, L. Zhou, and J. Shu, "Achieving efficient cloud search services: Multi-keyword ranked search over encrypted cloud data supporting parallel computing," *IEICE Trans. Commun.*, vol. E98.B, no. 1, pp. 190–200, 2015.
- [74] Y. Ren, J. Shen, J. Wang, J. Han, and S.-Y. Lee, "Mutual verifiable provable data auditing in public cloud storage," *J. Internet Technol.*, vol. 16, no. 2, pp. 317–323, 2015.
- [75] R. Yu, W. Zhong, S. Xie, C. Yuen, S. Gjessing, and Y. Zhang, "Balancing power demand through EV mobility in vehicle-to-grid mobile energy networks," *IEEE Trans. Ind. Informat.*, vol. 12, no. 1, pp. 79–90, Feb. 2016.
- [76] S. Deilami, A. S. Masoum, P. S. Moses, and M. A. S. Masoum, "Realtime coordination of plug-in electric vehicle charging in smart grids to minimize power losses and improve voltage profile," *IEEE Trans. Smart Grid*, vol. 2, no. 3, pp. 456–467, Sep. 2011.
- [77] C. Chen and S. Duan, "Optimal integration of plug-in hybrid electric vehicles in microgrids," *IEEE Trans. Ind. Informat.*, vol. 10, no. 3, pp. 1917–1926, Aug. 2014.
- [78] J. Zhao, F. Wen, Z. Y. Dong, Y. Xue, and K. P. Wong, "Optimal dispatch of electric vehicles and wind power using enhanced particle swarm optimization," *IEEE Trans. Ind. Informat.*, vol. 8, no. 4, pp. 889–899, Nov. 2012.
- [79] W. Tushar, C. Yuen, S. Huang, D. B. Smith, and H. V. Poor, "Cost minimization of charging stations with photovoltaics: An approach with EV classification," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 1, pp. 156–169, Jan. 2016.
- [80] H. Liu, Z. Hu, Y. Song, and J. Lin, "Decentralized vehicle-to-grid control for primary frequency regulation considering charging demands," *IEEE Trans. Power Syst.*, vol. 28, no. 3, pp. 3480–3489, Aug. 2013.
- [81] H. Liang, B. J. Choi, W. Zhuang, and X. Shen, "Optimizing the energy delivery via V2G systems based on stochastic inventory theory," *IEEE Trans. Smart Grid*, vol. 4, no. 4, pp. 2230–2243, Dec. 2013.
- [82] M.-A. Rostami, A. Kavousi-Fard, and T. Niknam, "Expected cost minimization of smart grids with plug-in hybrid electric vehicles using optimal distribution feeder reconfiguration," *IEEE Trans. Ind. Informat.*, vol. 11, no. 2, pp. 388–397, Apr. 2015.
- [83] J. J. Escudero-Garzás, A. García-Armada, and G. Seco-Granados, "Fair design of plug-in electric vehicles aggregator for V2G regulation," *IEEE Trans. Veh. Technol.*, vol. 61, no. 8, pp. 3406–3419, Oct. 2012.
- [84] S. De Breucker, P. Jacqmaer, K. De Brabandere, J. Driesen, and R. Belmans, "Grid power quality improvements using grid-coupled hybrid electric vehicles PEMD 2006," in *Proc. 3rd IET Int. Conf. Power Electron., Mach. Drives (PEMD)*, 2006, pp. 505–509.
- [85] M. Takagi, Y. Iwafune, K. Yamaji, H. Yamamoto, K. Okano, R. Hiwatari, and T. Ikeya, "Electricity pricing for PHEV bottom charge in daily load curve based on variation method," in *Proc. IEEE PES Innov. Smart Grid Technol. (ISGT)*, Jan. 2012, pp. 1–6.
- [86] S. G. Wirasingha, N. Schofield, and A. Emadi, "Plug-in hybrid electric vehicle developments in the U.S.: Trends, barriers, and economic feasibility," in *Proc. IEEE Vehicle Power Propuls. Conf.*, Sep. 2008, pp. 1–8.
- [87] C. Guille and G. Gross, "A conceptual framework for the vehicle-to-grid (V2G) implementation," *Energy Policy*, vol. 37, no. 11, pp. 4379–4390, Nov. 2009.
- [88] D. Dallinger, D. Krampe, and M. Wietschel, "Vehicle-to-grid regulation reserves based on a dynamic simulation of mobility behavior," *IEEE Trans. Smart Grid*, vol. 2, no. 2, pp. 302–313, Jun. 2011.
- [89] E. Keane and D. Flynn, "Potential for electric vehicles to provide power system reserve," in *Proc. IEEE PES Innov. Smart Grid Technol. (ISGT)*, Jan. 2012, pp. 1–7.

- [90] A. K. Srivastava, B. Annabathina, and S. Kamalasadan, "The challenges and policy options for integrating plug-in hybrid electric vehicle into the electric grid," *Electr. J.*, vol. 23, no. 3, pp. 83–91, Apr. 2010.
- [91] J. Zhong, L. He, C. Li, Y. Cao, J. Wang, B. Fang, L. Zeng, and G. Xiao, "Coordinated control for large-scale EV charging facilities and energy storage devices participating in frequency regulation," *Appl. Energy*, vol. 123, pp. 253–262, Jun. 2014.
- [92] H. Fan, L. Jiang, C. Zhang, and C. Mao, "Frequency regulation of multi-area power systems with plug-in electric vehicles considering communication delays," *IET Gener., Transmiss. Distrib.*, vol. 10, no. 14, pp. 3481–3491, Nov. 2016.
- [93] K. M. Rogers, R. Klump, H. Khurana, A. A. Aquino-Lugo, and T. J. Overbye, "An authenticated control framework for distributed voltage support on the smart grid," *IEEE Trans. Smart Grid*, vol. 1, no. 1, pp. 40–47, Jun. 2010.
- [94] I. Pavić, T. Capuder, and I. Kuzle, "Value of flexible electric vehicles in providing spinning reserve services," *Appl. Energy*, vol. 157, pp. 60–74, Nov. 2015.
- [95] A. Y. S. Lam, K.-C. Leung, and V. O. K. Li, "Capacity estimation for vehicle-to-grid frequency regulation services with smart charging mechanism," *IEEE Trans. Smart Grid*, vol. 7, no. 1, pp. 156–166, Jan. 2016.
- [96] J. Wei, D. Kundur, T. Zourntos, and K. L. Butler-Purry, "A flocking-based paradigm for hierarchical cyber-physical smart grid modeling and control," *IEEE Trans. Smart Grid*, vol. 5, no. 6, pp. 2687–2700, Nov. 2014.
- [97] A. Farraj, E. Hammad, and D. Kundur, "A cyber-enabled stabilizing control scheme for resilient smart grid systems," *IEEE Trans. Smart Grid*, vol. 7, no. 4, pp. 1856–1865, Jul. 2016.
- [98] M. Andreasson, D. V. Dimarogonas, K. H. Johansson, and H. Sandberg, "Distributed vs. centralized power systems frequency control," in *Proc. Eur. Control Conf. (ECC)*, Jul. 2013, pp. 3524–3529.
- [99] M.-H. Wang and H.-C. Chen, "Transient stability control of multimachine power systems using flywheel energy injection," *IEE Proc.-Gener.*, *Transmiss. Distrib.*, vol. 152, no. 5, pp. 589–596, 2005.
- [100] P. Mercier, R. Cherkaoui, and A. Oudalov, "Optimizing a battery energy storage system for frequency control application in an isolated power system," *IEEE Trans. Power Syst.*, vol. 24, no. 3, pp. 1469–1477, Aug. 2009.
- [101] M. Andreasson, D. V. Dimarogonas, H. Sandberg, and K. H. Johansson, "Distributed control of networked dynamical systems: Static feedback, integral action and consensus," *IEEE Trans. Autom. Control*, vol. 59, no. 7, pp. 1750–1764, Jul. 2014.
- [102] K. Kawabe and A. Yokoyama, "Improvement of transient stability and short-term voltage stability by rapid control of batteries on EHV network in power systems," *Electr. Eng. Jpn.*, vol. 188, no. 3, pp. 1–10, Aug. 2014.
- [103] M. Tucci, A. Floriduz, S. Riverso, and G. Ferrari-Trecate, "Kron reduction methods for plug-and-play control of AC islanded microgrids with arbitrary topology," 2015, arXiv:1510.07873.
- [104] M. Ayar, S. Obuz, R. D. Trevizan, A. S. Bretas, and H. A. Latchman, "A distributed control approach for enhancing smart grid transient stability and resilience," *IEEE Trans. Smart Grid*, vol. 8, no. 6, pp. 3035–3044, Nov. 2017.
- [105] R. H. Salim, M. Oleskovicz, and R. A. Ramos, "Power quality of distributed generation systems as affected by electromechanical oscillations—Definitions and possible solutions," *IET Gener, Transmiss. Distrib.*, vol. 5, no. 11, pp. 1114–1123, 2011.
- [106] M. Farina and R. Scattolini, "Distributed predictive control: A noncooperative algorithm with neighbor-to-neighbor communication for linear systems," *Automatica*, vol. 48, no. 6, pp. 1088–1096, Jun. 2012.
- [107] D.-C. Tomozei and J.-Y. Le Boudec, "Satisfiability of elastic demand in the smart grid," *CoRR*, vol. abs/1011.5606, 2010. [Online]. Available: http://arxiv.org/abs/1011.5606
- [108] V. S. Pappala, I. Erlich, K. Rohrig, and J. Dobschinski, "A stochastic model for the optimal operation of a wind-thermal power system," *IEEE Trans. Power Syst.*, vol. 24, no. 2, pp. 940–950, May 2009.
- [109] J. R. Birge and F. Louveaux, Introduction to Stochastic Programming. Springer, 2011, p. 485, doi: 10.1007/978-1-4614-0237-4.
- [110] J. R. Birge, "State-of-the-art-survey—Stochastic programming: Computation and applications," *INFORMS J. Comput.*, vol. 9, no. 2, pp. 111–133, May 1997.
- [111] M. V. F. Pereira and L. M. V. G. Pinto, "Multi-stage stochastic optimization applied to energy planning," *Math. Program.*, vol. 52, nos. 1–3, pp. 359–375, May 1991.
- [112] W. Zhang, K. Spence, R. Shao, and L. Chang, "Optimal scheduling of spinning reserve and user cost in vehicle-to-grid (V2G) systems," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2018, pp. 1058–1064.

- [113] A. Alsharif, A. A. Ahmed, M. M. Khaleel, and M. A. Altayib, "Ancillary services and energy management for electric vehicle: Minireview," *North Afr. J. Sci. Publishing*, vol. 1, no. 1, pp. 9–12, Jan./Mar. 2023.
- [114] V. Singh, T. Moger, and D. Jena, "Uncertainty handling techniques in power systems: A critical review," *Electr. Power Syst. Res.*, vol. 203, Feb. 2022, Art. no. 107633.
- [115] R. N. Allan, B. Borkowska, and C. H. Grigg, "Probabilistic analysis of power flows," *Proc. Inst. Elect. Eng.*, vol. 121, no. 12, pp. 1551–1556, Dec. 1974.
- [116] R. N. Allan and M. R. G. Al-Shakarchi, "Probabilistic ac load flow," *Proc. Inst. Electr. Eng.*, vol. 123, no. 6, pp. 531–536, 1976.
- [117] R. N. Allan, C. H. Grigg, and M. R. G. Al-Shakarchi, "Numerical techniques in probabilistic load flow problems," *Int. J. Numer. Methods Eng.*, vol. 10, no. 4, pp. 853–860, 1976.
- [118] R. N. Allan, A. M. L. da Silva, A. A. Abu-Nasser, and R. C. Burchett, "Discrete convolution in power system reliability," *IEEE Trans. Rel.*, vol. R-30, no. 5, pp. 452–456, Dec. 1981.
- [119] G. Valverde, A. T. Saric, and V. Terzija, "Stochastic monitoring of distribution networks including correlated input variables," *IEEE Trans. Power Syst.*, vol. 28, no. 1, pp. 246–255, Feb. 2013.
- [120] B. R. Prusty and D. Jena, "Sequence operation theory based probabilistic load flow assessment with photovoltaic generation," in *Proc. Michael Faraday IET Int. Summit*, Sep. 2015, pp. 164–169.
- [121] K. N. Hasan, R. Preece, and J. V. Milanović, "Existing approaches and trends in uncertainty modelling and probabilistic stability analysis of power systems with renewable generation," *Renew. Sustain. Energy Rev.*, vol. 101, pp. 168–180, Mar. 2019.
- [122] G. L. Doorman, K. Uhlen, G. H. Kjolle, and E. S. Huse, "Vulnerability analysis of the Nordic power system," *IEEE Trans. Power Syst.*, vol. 21, no. 1, pp. 402–410, Feb. 2006.
- [123] M. Thirunavukkarasu, Y. Sawle, and H. Lala, "A comprehensive review on optimization of hybrid renewable energy systems using various optimization techniques," *Renew. Sustain. Energy Rev.*, vol. 176, Apr. 2023, Art. no. 113192.
- [124] H. Attouch, R. I. Bot, and D.-K. Nguyen, "Fast convex optimization via closed-loop time scaling of gradient dynamics," 2023, arXiv:2301.00701.
- [125] E. Lannoye, D. Flynn, and M. O'Malley, "Transmission, variable generation, and power system flexibility," *IEEE Trans. Power Syst.*, vol. 30, no. 1, pp. 57–66, Jan. 2015.
- [126] M. F. Shaaban, M. H. Ahmed, T. H. M. El-Fouly, and M. M. A. Salama, "Impact of integrating PEV and renewable sources on power system reliability assessment," in *Proc. 4th Int. Conf. Electr. Power Energy Convers. Syst. (EPECS)*, Nov. 2015, pp. 1–6.
- [127] P. Papadopoulos, S. Skarvelis-Kazakos, I. Grau, L. M. Cipcigan, and N. Jenkins, "Electric vehicles' impact on British distribution networks," *IET Electr. Syst. Transp.*, vol. 2, no. 3, pp. 91–102, Oct. 2012.
- [128] T. Häring, T. M. Kull, R. Ahmadiahangar, A. Rosin, M. Thalfeldt, and H. Biechl, "Microgrid oriented modeling of space heating system based on neural networks," *J. Building Eng.*, vol. 43, Nov. 2021, Art. no. 103150.
- [129] A.-L. Barabási, "Scale-free networks: A decade and beyond," Science, vol. 325, no. 5939, pp. 412–413, Jul. 2009.
- [130] H. Lovell, "Nodes," in Understanding Energy Innovation: Learning From Smart Grid Experiments. Singapore: Springer, 2022, pp. 37–52, doi: 10.1007/978-981-16-6253-9\_3.
- [131] L. Jiang, X. Jin, Y. Xia, B. Ouyang, D. Wu, and X. Chen, "A scalefree topology construction model for wireless sensor networks," *Int. J. Distrib. Sensor Netw.*, vol. 10, no. 8, Aug. 2014, Art. no. 764698.
- [132] Y. Ding, X. Li, Y.-C. Tian, G. Ledwich, Y. Mishra, and C. Zhou, "Generating scale-free topology for wireless neighborhood area networks in smart grid," *IEEE Trans. Smart Grid*, vol. 10, no. 4, pp. 4245–4252, Jul. 2019.
- [133] N. C. Onat, M. Kucukvar, and O. Tatari, "Conventional, hybrid, plugin hybrid or electric vehicles? State-based comparative carbon and energy footprint analysis in the United States," *Appl. Energy*, vol. 150, pp. 36–49, Jul. 2015.
- [134] J. G. da Silva Fonseca Junior, T. Oozeki, H. Ohtake, K.-I. Shimose, T. Takashima, and K. Ogimoto, "Regional forecasts and smoothing effect of photovoltaic power generation in Japan: An approach with principal component analysis," *Renew. Energy*, vol. 68, pp. 403–413, Aug. 2014.
- [135] N. Abas, A. Kalair, N. Khan, and A. R. Kalair, "Review of GHG emissions in Pakistan compared to SAARC countries," *Renew. Sustain. Energy Rev.*, vol. 80, pp. 990–1016, Dec. 2017.

- [136] M. Adnan, A. Israr, N. Khan, and M. Irfan, "A generic scenario of a load flow study in SAARC countries super grid using high voltage alternating current (HVAC) and high voltage direct current (HVDC)," *Int. J. Sci. Eng. Res.*, vol. 7, pp. 210–223, Feb. 2016.
- [137] G. Czisch, J. Schmid, and D. Hein, "Scenarios for a future electricity supply: Cost-optimized variations on supplying Europe and its neighbours with electricity from renewable energies," Jan. 2011, pp. 1–581, doi: 10.1049/PBRN010E.
- [138] Potentiale der Regenerativen Stromerzeugung in Nordafrika, Universität Kassel, Kassel, Germany, 1999.
- [139] G. Czisch and G. Giebel, "Realisable scenarios for a future electricity supply based 100% on renewable energies," Energy Planning, Policy Economy, Gen. Miscellaneous, Denmark, Tech. Rep. RISO-R-1608(EN), 2007, pp. 186–195. [Online]. Available: http://www.risoe.dk/rispubl/ reports/ris-r-1608.pdf and http://inis.iaea.org/search/search.aspx?orig\_q= RN:38094616
- [140] F. Trieb, "Concentrating solar power for the Mediterranean region," German Aerosp. Center (DLR), Inst. Tech. Thermodyn. Sect. Syst. Anal. Technol. Assessment, Jan. 2005.
- [141] F. Trieb and H. Müller-Steinhagen, "Europe—Middle east—North Africa cooperation for sustainable electricity and water," *Sustainability Sci.*, vol. 2, pp. 205–219, Sep. 2007, doi: 10.1007/s11625-007-0025-x.
- [142] W. E. Alnaser, F. Trieb, and G. Knies, "Solar energy technology in the Middle East and North Africa (MENA) for sustainable energy, water and environment," in *Advances in Solar Energy*, vol. 17. London, U.K.: Plenum Publishing Corporation, 2007, p. 261.
- [143] F. Trieb and H. Müller-Steinhagen, "Sustainable electricity and water for Europe, Middle East and North Africa," in *EU-MENA Cooperation for Sustainable Electricity and Water*. Springer-Verlag, 2007, pp. 205–219.
- [144] M. Zhang, J. Ding, Y. Cai, and H. Wang, "Research on control strategy of MMC-MTDC system based on improved droop control," J. Phys., Conf. Ser., vol. 1176, Mar. 2019, Art. no. 062011.
- [145] D. Coll-Mayor, M. Paget, and E. Lightner, "Future intelligent power grids: Analysis of the vision in the European Union and the United States," *Energy Policy*, vol. 35, no. 4, pp. 2453–2465, Apr. 2007.
- [146] B. Buchholz, N. D. Hatziargyriou, I. Furones, and U. Schluecking, "Lessons learned: European pilot installations for distributed generation—An overview by the IRED cluster," 2006.
- [147] M. Chebbo, "EU SmartGrids framework 'electricity networks of the future 2020 and beyond," in *Proc. IEEE Power Eng. Soc. Gen. Meeting*, Jun. 2007, pp. 1–8.
- [148] A. Battaglini, J. Lilliestam, C. Bals, and A. Haas, "The SuperSmart grid," Jun. 2008.
- [149] A. E. Hammad, "Stability and control of HVDC and AC transmissions in parallel," *IEEE Trans. Power Del.*, vol. 14, no. 4, pp. 1545–1554, 1999, doi: 10.1109/61.796252.
- [150] K. Vu, M. M. Begouic, and D. Novosel, "Grids get smart protection and control," *IEEE Comput. Appl. Power*, vol. 10, no. 4, pp. 40–44, Oct. 1997.
- [151] K. Moslehi and R. Kumar, "A reliability perspective of the smart grid," *IEEE Trans. Smart Grid*, vol. 1, no. 1, pp. 57–64, Jun. 2010.
- [152] X. Fang, S. Misra, G. Xue, and D. Yang, "Smart grid—The new and improved power grid: A survey," *IEEE Commun. Surveys Tuts.*, vol. 14, no. 4, pp. 944–980, 4th Quart., 2012.
- [153] F. Mwasilu, J. J. Justo, E.-K. Kim, T. D. Do, and J.-W. Jung, "Electric vehicles and smart grid interaction: A review on vehicle to grid and renewable energy sources integration," *Renew. Sustain. Energy Rev.*, vol. 34, pp. 501–516, Jun. 2014.
- [154] H. N. T. Nguyen, C. Zhang, and Md. A. Mahmud, "Optimal coordination of G2V and V2G to support power grids with high penetration of renewable energy," *IEEE Trans. Transport. Electrific.*, vol. 1, no. 2, pp. 188–195, Aug. 2015.
- [155] S. Jain, N. V. Kumar, A. Paventhan, V. K. Chinnaiyan, V. Arnachalam, and M. Pradish, "Survey on smart grid technologies-smart metering, IoT and EMS," in *Proc. IEEE Students' Conf. Electr., Electron. Comput. Sci.*, Mar. 2014, pp. 1–6.
- [156] B. Ramezy, M. Saadatmand, and B. Mozafari, "Review of communication technologies for smart grid applications," in *Proc. Nat. Conf. New Approaches Power Ind.*, vol. 31, Tehran, Iran, 2017, pp. 1–9.
- [157] O. Alamu, B. Iyaomolere, and A. Abdulrahman, "An overview of massive MIMO localization techniques in wireless cellular networks: Recent advances and outlook," *Ad Hoc Netw.*, vol. 111, Feb. 2021, Art. no. 102353.

- [158] Y. Zhang, W. Sun, L. Wang, H. Wang, R. C. Green, and M. Alam, "A multi-level communication architecture of smart grid based on congestion aware wireless mesh network," in *Proc. North Amer. Power Symp.*, Aug. 2011, pp. 1–6.
- [159] H. Chidiebere, "Cellular network architecture: An overview," 2017. [Online]. Available: https://steemit.com/stemng/@henrychidiebere/ cellular-network-architecture-an-overview
- [160] COAI. (2017). Cellular Network Architecture. [Online]. Available: https://coai.com/indian-telecom-infocentre/telecom-infrastructure networks
- [161] A. Yarali and S. Rahman, "Wireless communication for smart grids," in Proc. 2nd Int. Conf. Adv. Collaborative Netw., Syst. Appl. (COLLA), 2012, pp. 1–9.
- [162] P. P. Parikh, M. G. Kanabar, and T. S. Sidhu, "Opportunities and challenges of wireless communication technologies for smart grid applications," in *Proc. IEEE PES Gen. Meeting*, Jul. 2010, pp. 1–7.
- [163] B. O. Barker, A. G. Frisbie, and K. R. Patrick, "Concepts: Broadening the definition of distance education in light of the new telecommunications technologies," *Amer. J. Distance Educ.*, vol. 3, no. 1, pp. 20–29, Jan. 1989.
- [164] V. C. Gungor, D. Sahin, T. Kocak, S. Ergut, C. Buccella, C. Cecati, and G. P. Hancke, "Smart grid technologies: Communication technologies and standards," *IEEE Trans. Ind. Informat.*, vol. 7, no. 4, pp. 529–539, Nov. 2011.
- [165] S. Galli, A. Scaglione, and Z. Wang, "For the grid and through the grid: The role of power line communications in the smart grid," *Proc. IEEE*, vol. 99, no. 6, pp. 998–1027, Jun. 2011.
- [166] R. P. Lewis, P. Igict, and Z. Zhou, "Assessment of communication methods for smart electricity metering in the U.K.," in *Proc. IEEE PES/IAS Conf. Sustain. Alternative Energy (SAE)*, Sep. 2009, pp. 1–4.
- [167] R. Makwana, J. Baviskar, N. Panchal, and D. Karia, "Wireless based load control and power monitoring system," in *Proc. Int. Conf. Energy Efficient Technol. Sustainability*, Apr. 2013, pp. 1207–1211.
- [168] A. Yokus and H. M. Baskonus, "Dynamics of traveling wave solutions arising in fiber optic communication of some nonlinear models," *Soft Comput.*, vol. 26, no. 24, pp. 13605–13614, Dec. 2022.
- [169] B. Elliott, B. J. Elliott, and M. Gilmore, *Fiber Optic Cabling*. Boston, MA, USA: Newnes, 2002.
- [170] C. Rohmingtluanga, S. Datta, N. Sinha, and T. S. Ustun, "SCADA based intake monitoring for improving energy management plan: Case study," *Energy Rep.*, vol. 9, pp. 402–410, Mar. 2023.
- [171] What is SCADA, Inductive Automat., Folsom, CA, USA, 2018.
- [172] M. Alanazi, A. Mahmood, and M. J. M. Chowdhury, "SCADA vulnerabilities and attacks: A review of the state-of-the-art and open issues," *Comput. Secur.*, vol. 125, Feb. 2023, Art. no. 103028.
- [173] N. Raza, M. Q. Akbar, A. A. Soofi, and S. Akbar, "Study of smart grid communication network architectures and technologies," *J. Comput. Commun.*, vol. 7, no. 3, pp. 19–29, 2019.
- [174] G. Janevska, "Mathematical modeling of hybrid renewable energy system," in *Proc. 3rd Virtual Int. Conf. Sci., Technol. Manage. Energy Energetics*, 2017, pp. 9–15.
- [175] H. K. Khalil, Nonlinear Control, vol. 406. New York, NY, USA: Pearson, 2015.
- [176] W. K. Ahmed, "Mechanical modeling of wind turbine comparative study," Int. J. Renew. Energy Res., vol. 3, no. 1, pp. 94–97, 2013.
- [177] R. M. Ochieng, F. N. Onyango, and A. O. Oduor, "Physical formulation of the expression of wind power," *Int. J. Energy, Environ. Econ.*, vol. 18, no. 1, p. 107, 2010.
- [178] K. E. Johnson, "Adaptive torque control of variable speed wind turbines," Univ. Colorado Boulder, 2004.
- [179] A. W. Manyonge, R. Ochieng, F. Onyango, and J. Shichikha, "Mathematical modelling of wind turbine in a wind energy conversion system: Power coefficient analysis," Tech. Rep., 2012.
- [180] B. Williams, M. Gahagan, and K. Costin, "Using microgrids to integrate distributed renewables into the grid," in *Proc. IEEE PES Innov. Smart Grid Technol. Conf. Eur. (ISGT Europe)*, Oct. 2010, pp. 1–5.
- [181] D. T. Ton, W. M. Wang, and W.-T.-P. Wang, "Smart grid R&D by the U.S. Department of Energy to optimize distribution grid operations," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2011, pp. 1–5.
- [182] G. M. Shafiullah, A. M. T. Oo, D. Jarvis, A. B. M. S. Ali, and P. Wolfs, "Potential challenges: Integrating renewable energy with the smart grid," in *Proc. 20th Australas. Univ. Power Eng. Conf.*, Dec. 2010, pp. 1–6.

- [183] M. Irfan, J. Iqbal, A. Iqbal, Z. Iqbal, R. A. Riaz, and A. Mehmood, "Opportunities and challenges in control of smart grids—Pakistani perspective," *Renew. Sustain. Energy Rev.*, vol. 71, pp. 652–674, May 2017.
- [184] Y. Deng, "Future vehicle trend: A comparative study of the fuel vehicle, electrical vehicle, and hybrid vehicles," *Highlights Sci., Eng. Technol.*, vol. 29, pp. 143–148, Jan. 2023.
- [185] L. Bozo, A. Hajderi, and F. Basholli, "The impact of hybrid cars on reducing urban pollution and global warming," *Adv. Eng. Days*, vol. 6, pp. 117–120, Mar. 2023.
- [186] A. Demeo and M. L. Peterson, "Community smart grid utilizing dynamic demand response and tidal power for grid stabilization," *Smart Grid Renew. Energy*, vol. 4, no. 7, pp. 465–472, 2013.
- [187] H. Yang, C. Y. Chung, and J. Zhao, "Application of plug-in electric vehicles to frequency regulation based on distributed signal acquisition via limited communication," *IEEE Trans. Power Syst.*, vol. 28, no. 2, pp. 1017–1026, May 2013.
- [188] I. Ahmed, M. Adnan, M. Ali, and G. Kaddoum, "Supertwisting sliding mode controller for grid-to-vehicle and vehicle-to-grid battery electric vehicle charger," *J. Energy Storage*, vol. 70, Oct. 2023, Art. no. 107914.
- [189] S. Han, S. Han, and K. Sezaki, "Development of an optimal vehicle-togrid aggregator for frequency regulation," *IEEE Trans. Smart Grid*, vol. 1, no. 1, pp. 65–72, Jun. 2010.
- [190] M. A. Mahmud, M. J. Hossain, H. R. Pota, and N. K. Roy, "Nonlinear controller design for vehicle-to-grid (V2G) systems to enhance power quality and power system stability," *IFAC Proc. Volumes*, vol. 47, no. 3, pp. 7659–7664, 2014.
- [191] A. Rachid, H. E. Fadil, F. Giri, and A. Lassioui, "Nonlinear output feedback control of V2G single-phase on-board BEV charger," *Asian J. Control*, vol. 22, no. 5, pp. 1848–1859, Sep. 2020.
- [192] M. Ceraolo, "New dynamical models of lead-acid batteries," *IEEE Trans. Power Syst.*, vol. 15, no. 4, pp. 1184–1190, Nov. 2000.
- [193] I.-S. Kim, "Sliding mode controller for the single-phase grid-connected photovoltaic system," *Appl. Energy*, vol. 83, no. 10, pp. 1101–1115, Oct. 2006.
- [194] L. Noel, G. Z. de Rubens, J. Kester, and B. K. Sovacool, "The potential benefits of V2G," in *Vehicle-to-Grid: A Sociotechnical Transition Beyond Electric Mobility*. Cham, Switzerland: Springer, 2019, pp. 33–64, doi: 10.1007/978-3-030-04864-8\_2.
- [195] I. Ahmed, I. Ahmed, S. Ahmed, and H. M. M. Adil, "Robust nonlinear control of battery electric vehicle charger in grid to vehicle applications," *J. Energy Storage*, vol. 42, Oct. 2021, Art. no. 103039.
- [196] I. Ahmed, H. M. M. Adil, S. Ahmed, I. Ahmad, and Z. Rehman, "Robust nonlinear control of battery electric vehicle charger in grid to vehicle and vehicle to grid applications," *J. Energy Storage*, vol. 52, Aug. 2022, Art. no. 104813.
- [197] E. F. Camacho, F. R. Rubio, M. Berenguel, and L. Valenzuela, "A survey on control schemes for distributed solar collector fields. Part I: Modeling and basic control approaches," *Sol. Energy*, vol. 81, no. 10, pp. 1240–1251, Oct. 2007.
- [198] L. Carotenuto, M. L. Cava, and G. Raiconi, "Regulator design for the bilinear distributed-parameter model of a solar-power plant," *Int. J. Syst. Sci.*, vol. 16, no. 7, pp. 885–900, Jul. 1985.
- [199] E. F. Camacho, F. R. Rubio, and F. M. Hughes, "Self-tuning control of a solar power plant with a distributed collector field," *IEEE Control Syst.*, vol. 12, no. 2, pp. 72–78, Apr. 1992.
- [200] M. Berenguel, F. R. Rubio, E. F. Camacho, and F. Gordillo, "Techniques and applications of fuzzy logic control of solar power plants," in *Fuzzy Theory Systems*. Amsterdam, The Netherlands: Elsevier, 1999, pp. 717–745.
- [201] S. Alepuz, S. Busquets-Monge, J. Bordonáu, J. Gago, D. Gonzalez, and J. Balcells, "Interfacing renewable energy sources to the utility grid using a three-level inverter," *IEEE Trans. Ind. Electron.*, vol. 53, no. 5, pp. 1504–1511, Oct. 2006.
- [202] R. N. Silva, L. M. Rato, J. M. Lemos, and F. Coito, "Cascade control of a distributed collector solar field," *J. Process Control*, vol. 7, no. 2, pp. 111–117, Jan. 1997.
- [203] L. Rato, R. N. Silva, J. M. Lemos, and F. J. Coito, "Multirate MUSMAR cascade control of a distributed collector solar field," in *Proc. Eur. Control Conf. (ECC)*, Jul. 1997, pp. 1009–1013.
- [204] R. N. Anderson, A. Boulanger, W. B. Powell, and W. Scott, "Adaptive stochastic control for the smart grid," *Proc. IEEE*, vol. 99, no. 6, pp. 1098–1115, Jun. 2011.

- [205] M. Manbachi, H. Farhangi, A. Palizban, and S. Arzanpour, "Smart grid adaptive volt-VAR optimization: Challenges for sustainable future grids," *Sustain. Cities Soc.*, vol. 28, pp. 242–255, Jan. 2017.
- [206] A. D. Domínguez-García, "Models for impact assessment of wind-based power generation on frequency control," in *Control and Optimization Methods for Electric Smart Grids*, A. Chakrabortty and M. D. Ilić, Eds. New York, NY, USA: Springer, 2012, pp. 149–165, doi: 10.1007/978-1-4614-1605-0\_7.
- [207] Z. Ma, D. Callaway, and I. Hiskens, "Optimal charging control for plugin electric vehicles," in *Control and Optimization Methods for Electric Smart Grids*, A. Chakrabortty and M. D. Ilić, Eds. New York, NY, USA: Springer, 2012, pp. 259–273, doi: 10.1007/978-1-4614-1605-0\_13.
- [208] J. H. Chow and S. G. Ghiocel, "An adaptive wide-area power system damping controller using synchrophasor data," in *Control and Optimization Methods for Electric Smart Grids*, A. Chakrabortty and M. D. Ilić, Eds. New York, NY, USA: Springer, 2012, pp. 327–342, doi: 10.1007/978-1-4614-1605-0\_17.
- [209] A. Chakrabortty, "A model reference approach for interarea modal damping in large power systems," in *Control and Optimization Methods for Electric Smart Grids*, A. Chakrabortty and M. D. Ilić, Eds. New York, NY, USA: Springer, 2012, pp. 343–362, doi: 10.1007/978-1-4614-1605-0\_18.
- [210] K. N. Nwaigwe, P. Mutabilwa, and E. Dintwa, "An overview of solar power (PV systems) integration into electricity grids," *Mater. Sci. Energy Technol.*, vol. 2, no. 3, pp. 629–633, Dec. 2019.
- [211] M. Ahat, S. B. Amor, M. Bui, A. Bui, G. Gurard, and C. Petermann, "Smart grid and optimization," *Amer. J. Oper. Res.*, vol. 3, no. 1, pp. 1–11, 2013.
- [212] M. Al-Kaabi, S. Q. Salih, I. B. Al Hussein, V. Dumbrava, and M. Eremia, "Optimal power flow based on grey wolf optimizer: Case study iraqi super grid high voltage 400 kV," in *Proc. 2nd Int. Conf. Emerg. Technol. Intell. Syst.*, M. A. Al-Sharafi, M. Al-Emran, M. N. Al-Kabi, and K. Shaalan, Eds. Cham, Switzerland: Springer, 2023, pp. 490–503.
- [213] I. M. Jawad and W. S. Majeed, "Enhancing and improving the performance of the Iraqi super grid 400 kV by connection distributed generation based on optimization techniques," *AIP Conf. Proc.*, vol. 2404, no. 1, 2021, Art. no. 080023.
- [214] D. P. Ladumor, I. N. Trivedi, R. H. Bhesdadiya, and P. Jangir, "A grey wolf optimizer algorithm for voltage stability enhancement," in *Proc. 3rd Int. Conf. Adv. Electr., Electron., Inf., Commun. Bio-Informat. (AEEICB)*, Feb. 2017, pp. 278–282.
- [215] R. A. El Schiemy, F. Selim, B. Bentouati, and M. A. Abido, "A novel multi-objective hybrid particle swarm and salp optimization algorithm for technical-economical-environmental operation in power systems," *Energy*, vol. 193, Feb. 2020, Art. no. 116817.
- [216] Q. Al-Tashi, H. M. Rais, S. J. Abdulkadir, S. Mirjalili, and H. Alhussian, "A review of grey wolf optimizer-based feature selection methods for classification," in *Evolutionary Machine Learning Techniques: Algorithms and Applications*, S. Mirjalili, H. Faris, and I. Aljarah, Eds. Singapore: Springer, 2020, pp. 273–286, doi: 10.1007/978-981-32-9990-0\_13.
- [217] M. Al-Kaabi, L. Al-Bahrani, V. Dumbrava, and M. Eremia, "Optimal power flow with four objective functions using improved differential evolution algorithm: Case study IEEE 57-bus power system," in *Proc. 10th Int. Conf. Energy Environ. (CIEM)*, Oct. 2021, pp. 1–5.
- [218] A. A. Al-Hadeethi, M. Wesołowski, and M. H. Fayyadh, "Minimization of energy loss based on var control in 400 kV super grid using MATLAB," *Elektronika, Konstrukcje, Technologie, Zastosowania*, vol. 59, no. 11, pp. 12–16, 2018.
- [219] C. Ghinea, M. Eremia, D. O. Sidea, and L. Toma, "Optimal location of a PV system using grey wolf optimizer and operation analysis," in *Proc.* 9th Int. Conf. Modern Power Syst. (MPS), Jun. 2021, pp. 1–6.
- [220] A. Khoramshahi, H. A. D. Ashtiani, A. Khoshgard, H. Damghani, and L. Damghani, "Modeling the hybrid battery/super-storage system for a solar standalone microgrid," *Int. J. Energy Power Eng.*, vol. 17, no. 2, pp. 51–56, 2023.
- [221] C. Makdisie, B. Haidar, and H. H. Alhelou, "An optimal photovoltaic conversion system for future smart grids," in *Handbook of Research on Power and Energy System Optimization*. Hershey, PA, USA: IGI Global, 2018, pp. 601–657.
- [222] B. Singh, S. K. Bishnoi, M. Sharma, P. Singh, and S. Dhundhara, "An application of nature inspried algorithm based dual-stage frequency control strategy for multi micro-grid system," *Ain Shams Eng. J.*, vol. 14, no. 9, Sep. 2023, Art. no. 102125.

- [223] A. Rabie, A. Ghanem, S. S. Kaddah, and M. M. El-Saadawi, "Electric vehicles based electric power grid support: A review," *Int. J. Power Electron. Drive Syst.*, vol. 14, no. 1, p. 589, Mar. 2023.
- [224] B. K. Das, M. A. Alotaibi, P. Das, M. S. Islam, S. K. Das, and M. A. Hossain, "Feasibility and techno-economic analysis of stand-alone and grid-connected PV/wind/diesel/batt hybrid energy system: A case study," *Energy Strategy Rev.*, vol. 37, Sep. 2021, Art. no. 100673.
- [225] K. Shivam, J.-C. Tzou, and S.-C. Wu, "A multi-objective predictive energy management strategy for residential grid-connected PV-battery hybrid systems based on machine learning technique," *Energy Convers. Manage.*, vol. 237, Jun. 2021, Art. no. 114103.
- [226] S. Sharma, S. Bhattacharjee, and A. Bhattacharya, "Operation cost minimization of a micro-grid using quasi-oppositional swine influenza model based optimization with quarantine," *Ain Shams Eng. J.*, vol. 9, no. 1, pp. 45–63, Mar. 2018.
- [227] A. N. Khan, K. Imran, M. Nadeem, A. Pal, A. Khattak, K. Ullah, M. W. Younas, and M. S. Younis, "Ensuring reliable operation of electricity grid by placement of FACTS devices for developing countries," *Energies*, vol. 14, no. 8, p. 2283, Apr. 2021.
- [228] U. Assad, M. A. S. Hassan, U. Farooq, A. Kabir, M. Z. Khan, S. S. H. Bukhari, Z. U. A. Jaffri, J. Oláh, and J. Popp, "Smart grid, demand response and optimization: A critical review of computational methods," *Energies*, vol. 15, no. 6, p. 2003, Mar. 2022.
- [229] M. Alipour, H. Chitsaz, H. Zareipour, and D. Wood, "Microgrid energy management: How uncertainty modelling impacts economic performance," *IET Gener., Transmiss. Distrib.*, vol. 13, no. 24, pp. 5504–5510, Dec. 2019.
- [230] B. K. Bose, "Artificial intelligence techniques in smart grid and renewable energy systems—Some example applications," *Proc. IEEE*, vol. 105, no. 11, pp. 2262–2273, Nov. 2017.
- [231] Z. Shi, W. Yao, Z. Li, L. Zeng, Y. Zhao, R. Zhang, Y. Tang, and J. Wen, "Artificial intelligence techniques for stability analysis and control in smart grids: Methodologies, applications, challenges and future directions," *Appl. Energy*, vol. 278, Nov. 2020, Art. no. 115733.
- [232] S. R. Das, P. K. Ray, A. K. Sahoo, K. K. Singh, G. Dhiman, and A. Singh, "Artificial intelligence based grid connected inverters for power quality improvement in smart grid applications," *Comput. Electr. Eng.*, vol. 93, Jul. 2021, Art. no. 107208.
- [233] M. Chen, U. Challita, W. Saad, C. Yin, and M. Debbah, "Artificial neural networks-based machine learning for wireless networks: A tutorial," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 4, pp. 3039–3071, 4th Quart., 2019.
- [234] D. Alahakoon, R. Nawaratne, Y. Xu, D. De Silva, U. Sivarajah, and B. Gupta, "Self-building artificial intelligence and machine learning to empower big data analytics in smart cities," *Inf. Syst. Frontiers*, vol. 25, pp. 221–240, Aug. 2020.
- [235] X. Chen, "Looking into the supergrid and superblock structure in Chinese cities: Taking Xi'an and Nanjing as examples," in *Chinese Urban Planning and Construction: From Historical Wisdom to Modern Miracles*, L. Bian, Y. Tang, and Z. Shen, Eds. Cham, Switzerland: Springer, 2021, pp. 37–62, doi: 10.1007/978-3-030-65562-4\_3.
- [236] Y. K. Dwivedi et al., "Artificial intelligence (AI): Multidisciplinary perspectives on emerging challenges, opportunities, and agenda for research, practice and policy," *Int. J. Inf. Manage.*, vol. 57, Apr. 2021, Art. no. 101994.
- [237] M. Mohammadi and A. Al-Fuqaha, "Enabling cognitive smart cities using big data and machine learning: Approaches and challenges," *IEEE Commun. Mag.*, vol. 56, no. 2, pp. 94–101, Feb. 2018.
- [238] A. Haghighat and A. Sharma, "A computer vision-based deep learning model to detect wrong-way driving using pan-tilt-zoom traffic cameras," *Comput.-Aided Civil Infrastruct. Eng.*, vol. 38, no. 1, pp. 119–132, Jan. 2023.
- [239] T. Mazhar, R. N. Asif, M. A. Malik, M. A. Nadeem, I. Haq, M. Iqbal, M. Kamran, and S. Ashraf, "Electric vehicle charging system in the smart grid using different machine learning methods," *Sustainability*, vol. 15, no. 3, p. 2603, Feb. 2023.
- [240] H. Yousuf, A. Y. Zainal, M. Alshurideh, and S. A. Salloum, "Artificial intelligence models in power system analysis," in *Artificial Intelligence* for Sustainable Development: Theory, Practice and Future Applications, A. E. Hassanien, R. Bhatnagar, and A. Darwish, Eds. Cham, Switzerland: Springer, 2021, pp. 231–242, doi: 10.1007/978-3-030-51920-9\_12.
- [241] M. C. Horowitz, "Artificial intelligence, international competition, and the balance of power," *Texas Nat. Secur. Rev.*, vol. 1, no. 3, pp. 36–57, May 2018, doi: 10.15781/T2639KP49.

- [242] G. Ciulla, A. D'Amico, V. Lo Brano, and M. Traverso, "Application of optimized artificial intelligence algorithm to evaluate the heating energy demand of non-residential buildings at European level," *Energy*, vol. 176, pp. 380–391, Jun. 2019.
- [243] M. H. Ahmadi, H. Jashnani, K.-W. Chau, R. Kumar, and M. A. Rosen, "Carbon dioxide emissions prediction of five Middle Eastern countries using artificial neural networks," *Energy Sources, A, Recovery, Utilization, Environ. Effects*, pp. 1–13, 2019, doi: 10.1080/15567036. 2019.1679914.
- [244] B. Marr, Artificial Intelligence in Practice: How 50 Successful Companies Used AI and Machine Learning to Solve Problems. Hoboken, NJ, USA: Wiley, 2019.
- [245] M. Muro, R. Maxim, and J. Whiton, "Automation and artificial intelligence: How machines are affecting people and places," Brookings India, Jan. 2019. [Online]. Available: http://hdl.handle.net/ 20.500.11941/2552\_Automation-and-Artificial-Intelligence-Howmachines-are-affecting-people-and-places
- [246] V. González-Calatayud, P. Prendes-Espinosa, and R. Roig-Vila, "Artificial intelligence for student assessment: A systematic review," *Appl. Sci.*, vol. 11, no. 12, p. 5467, Jun. 2021.
- [247] W. Leal Filho, P. Yang, J. H. P. P. Eustachio, A. M. Azul, J. C. Gellers, A. Gielczyk, M. A. P. Dinis, and V. Kozlova, "Deploying digitalisation and artificial intelligence in sustainable development research," *Environ., Develop. Sustainability*, vol. 25, pp. 4957–4988, Mar. 2022.
- [248] R. Itiki, M. Manjrekar, S. G. Di Santo, and C. Itiki, "Method for spatiotemporal wind power generation profile under hurricanes: U.S.-Caribbean super grid proposition," *Renew. Sustain. Energy Rev.*, vol. 173, Mar. 2023, Art. no. 113082.
- [249] B. Shelton, "Introduction: An introductory essay: From radials and grids to supergrids and superblocks," in *Supergrid and Superblock*. Evanston, IL, USA: Routledge, 2023, pp. 1–23.
- [250] K. Hojckova, H. Ahlborg, and B. A. Sandén, "A global super-grid: Sociotechnical drivers and barriers," *Energy, Sustainability Soc.*, vol. 12, no. 1, p. 42, Oct. 2022.
- [251] H.-I. Chang, Y. Chun, and Y. Her, "A fair and stable benefit-sharing for the Northeast Asia supergrid under flexible networks," *Energy Econ.*, vol. 103, Nov. 2021, Art. no. 105521.
- [252] E. Halawa, G. James, X. Shi, N. Sari, and R. Nepal, "The prospect for an Australian–Asian power grid: A critical appraisal," *Energies*, vol. 11, no. 1, p. 200, Jan. 2018.
- [253] H. Wang, X. Yang, Q. Lou, and X. Xu, "Achieving a sustainable development process by deployment of solar PV power in ASEAN: A SWOT analysis," *Processes*, vol. 9, no. 4, p. 630, Apr. 2021.
- [254] H. Acaroğlu and F. P. G. Márquez, "A life-cycle cost analysis of high voltage direct current utilization for solar energy systems: The case study in Turkey," J. Cleaner Prod., vol. 360, Aug. 2022, Art. no. 132128.
- [255] C. Downie, "Geopolitical leverage in the energy transition: A framework for analysis and the case of Australian electricity exports," *Energy Res. Social Sci.*, vol. 93, Nov. 2022, Art. no. 102826.
- [256] A. Gulagi, M. Alcanzare, D. Bogdanov, E. Esparcia Jr., J. Ocon, and C. Breyer, "Transition pathway towards 100% renewable energy across the sectors of power, heat, transport, and desalination for the Philippines," *Renew. Sustain. Energy Rev.*, vol. 144, Jul. 2021, Art. no. 110934.
- [257] A. Alassi, S. Bañales, O. Ellabban, G. Adam, and C. MacIver, "HVDC transmission: Technology review, market trends and future outlook," *Renew. Sustain. Energy Rev.*, vol. 112, pp. 530–554, Sep. 2019.
- [258] M. N. Acosta, F. Gonzalez-Longatt, J. M. Roldan-Fernandez, and M. Burgos-Payan, "A coordinated control of offshore wind power and BESS to provide power system flexibility," *Energies*, vol. 14, no. 15, p. 4650, Jul. 2021.
- [259] X. Liang and M. Abbasipour, "HVDC transmission and its potential application in remote communities: A review," in *Proc. IEEE Ind. Appl. Soc. Annu. Meeting (IAS)*, Oct. 2021, pp. 1–9.
- [260] P. R. Chowdhury, S. Essakiappan, M. Manjrekar, K. Schneider, and S. Laval, "Optimized dispatch of distributed energy resources for resiliency and power quality improvements at the grid-edge," in *Proc. IEEE Kansas Power Energy Conf. (KPEC)*, Apr. 2021, pp. 1–6.
- [261] K. Hafeez and S. A. Khan, "High voltage direct current (HVDC) transmission: Future expectation for Pakistan," *CSEE J. Power Energy Syst.*, vol. 5, no. 1, pp. 82–86, Mar. 2019.
- [262] M. Z. Yousaf, A. Raza, G. Abbas, N. Ullah, A. A. Al-Ahmadi, A. R. Yasin, and M. Jamil, "MTDC grids: A metaheuristic solution for nonlinear control," *Energies*, vol. 15, no. 12, p. 4263, Jun. 2022.

- [263] D. Del Barrio Alvarez, A. Sasakawa, K. Yamaguchi, and J. Wang, "Strategic priorities for regional power connectivity in Asia in the overlapping of subregional initiatives and China's global energy interconnection," *Asia Pacific Bus. Rev.*, pp. 1–23, 2022, doi: 10.1080/13602381.2022.2093528.
- [264] E. Širá, R. Kotulič, I. K. Vozárová, and M. Daňová, "Sustainable development in EU countries in the framework of the Europe 2020 strategy," *Processes*, vol. 9, no. 3, p. 443, Feb. 2021.
- [265] P. J. Burke, F. J. Beck, E. Aisbett, K. G. H. Baldwin, M. Stocks, J. Pye, M. Venkataraman, J. Hunt, and X. Bai, "Contributing to regional decarbonization: Australia's potential to supply zero-carbon commodities to the Asia–Pacific," *Energy*, vol. 248, Jun. 2022, Art. no. 123563.
- [266] R. Itiki, M. Manjrekar, and S. G. Di Santo, "Comparative evaluation of super grid topologies proposed for Europe and Latin America," in *Proc. North Amer. Power Symp. (NAPS)*, Oct. 2019, pp. 1–6.
- [267] E. Pierri, O. Binder, N. G. A. Hemdan, and M. Kurrat, "Challenges and opportunities for a European HVDC grid," *Renew. Sustain. Energy Rev.*, vol. 70, pp. 427–456, Apr. 2017.
- [268] Z. Gansukh, "Mongol dream beyond fossil fuels: Prosperity of greenification," *Renew. Energy*, vol. 171, pp. 95–102, Jun. 2021.
- [269] K. M. Belikova, N. V. Badaeva, and M. A. Akhmadova, "Cooperation of China and Russia in the framework of the Asian super grid: Political and legal aspects," *Int. J. Econ. Bus. Admin.*, no. 1, pp. 215–226, 2019. [Online]. Available: https://ideas.repec.org/a/ers/ijebaa/vviiy2019 i1p215-226.html
- [270] S. Jaiswal and S. Sharma, "Electric vehicles and smart grid interactions: Infrastructure, current trends, impacts and challenges," in *Smart Grids* for Renewable Energy Systems, Electric Vehicles and Energy Storage Systems. Boca Raton, FL, USA: CRC Press, 2023, pp. 15–44.
- [271] S. F. Myhre, O. B. Fosso, O. Gjerde, and P. E. Heegaard, "A study on V2G impact on the reliability of modern distribution networks," 2023, arXiv:2302.10069.
- [272] F. Zhang, L. Wang, S. Coskun, H. Pang, Y. Cui, and J. Xi, "Energy management strategies for hybrid electric vehicles: Review, classification, comparison, and outlook," *Energies*, vol. 13, no. 13, p. 3352, Jun. 2020.
- [273] M. J. M. Al Essa, "Power management of PEV using linear programming with solar panels and wind turbines in smart grids," *Electr. Eng.*, vol. 105, pp. 1761–1773, Feb. 2023.
- [274] S. Zhou, "The effect of smart meter penetration on dynamic electricity pricing: Evidence from the United States," *Electr. J.*, vol. 34, no. 3, Apr. 2021, Art. no. 106919.
- [275] Y. T. Chai, H. S. Che, C. Tan, W.-N. Tan, S.-C. Yip, and M.-T. Gan, "A two-stage optimization method for vehicle to grid coordination considering building and electric vehicle user expectations," *Int. J. Electr. Power Energy Syst.*, vol. 148, Jun. 2023, Art. no. 108984.
- [276] W. Hou, Y. Sun, D. Li, Z. Guan, and J. Liu, "Lightweight and privacy-preserving charging reservation authentication protocol for 5G-V2G," *IEEE Trans. Veh. Technol.*, vol. 72, no. 6, pp. 7871–7883, Jun. 2023.
- [277] U. H. Ramadhani, M. Shepero, J. Munkhammar, J. Widén, and N. Etherden, "Review of probabilistic load flow approaches for power distribution systems with photovoltaic generation and electric vehicle charging," *Int. J. Electr. Power Energy Syst.*, vol. 120, Sep. 2020, Art. no. 106003.
- [278] D. Sengeni, M. Kavitha, U. Durairaj, K. S. Subramaniam, V. Prabhu, K. Suriyakrishnaan, and H. Shivappa, "Analysis and implementation of a high switching frequency dynamic wireless power transfer technique for electric vehicle charging devices," *AIP Conf. Proc.*, vol. 2519, no. 1, 2022, Art. no. 050060.
- [279] B. Lalitha, S. Divyapriya, A. Amudha, G. E. Gopalakrishnan, D. Viyathukattuva, M. S. Ramkumar, and M. Nivetha, "Design, analysis and application of dynamic wireless power transfer technique at high switching frequency for a electric vehicle battery chargers," *Int. J. Psychosocial Rehabil.*, vol. 23, no. 4, pp. 567–581, Dec. 2019.
- [280] P. Franzese, D. D. Patel, A. A. S. Mohamed, D. Iannuzzi, B. Fahimi, M. Risso, and J. M. Miller, "Fast DC charging infrastructures for electric vehicles: Overview of technologies, standards, and challenges," *IEEE Trans. Transport. Electrific.*, early access, Jan. 23, 2023, doi: 10.1109/TTE.2023.3239224.
- [281] C. Harnischmacher, L. Markefke, A. B. Brendel, and L. Kolbe, "Twosided sustainability: Simulating battery degradation in vehicle to grid applications within autonomous electric port transportation," *J. Cleaner Prod.*, vol. 384, Jan. 2023, Art. no. 135598.

- [282] R. Itiki, M. Manjrekar, S. G. Di Santo, and L. F. M. Machado, "Topology design method for super grids based on experiences in China and North America," in *Proc. IEEE Power Energy Soc. Innov. Smart Grid Technol. Conf. (ISGT)*, Feb. 2020, pp. 1–5.
- [283] N. W. Ndlela and I. E. Davidson, "Power planning for a smart integrated African super-grid," in *Proc. 30th Southern Afr. Univ. Power Eng. Conf.* (SAUPEC), Jan. 2022, pp. 1–6.
- [284] I. I. Hussein, S. Essallah, and A. Khedher, "Improvement of the Iraqi super grid performance using HVDC/HVAC links by the integration of large-scale renewable energy sources," *Energies*, vol. 15, no. 3, p. 1142, Feb. 2022.
- [285] R. P. K. Naidu and S. Meikandasivam, "Power quality enhancement in a grid-connected hybrid system with coordinated PQ theory & fractional order PID controller in DPFC," *Sustain. Energy, Grids Netw.*, vol. 21, Mar. 2020, Art. no. 100317.
- [286] A. M. Eltamaly, Y. Sayed, A.-H. M. El-Sayed, and A. N. A. Elghaffar, "Mitigation voltage sag using DVR with power distribution networks for enhancing the power system quality," *IJEEAS J.*, vol. 1, no. 2, pp. 2600–7495, 2018.
- [287] A. A. Ehimen, M. I. Bala, and A. I. Adejoh, "Modeling of existing 330 kV and meshed 330 kV super-grid transmission line for the Nigerian network," *J. Electr. Electron. Eng.*, vol. 15, no. 2, pp. 11–15, 2022.
- [288] A. E. Airoboman, A. P. Oriaifo, and S. T. Wara, "Impact of faults on bus stability on an island 330 kV mesh network on the Nigerian grid," *J. Electr., Control Technol. Res.*, vol. 2, pp. 30–41, Nov. 2020.
- [289] L. M. Adesina, A. Abdulkareem, and O. Ogunbiyi, "Effects of ground resistivity and tower structural design on transmission line symmetrical components," *Int. J. Recent Technol. Eng.*, vol. 9, no. 1, pp. 2115–2126, May 2020.
- [290] A. Abdulkareem, T. E. Somefun, C. O. A. Awosope, and O. Olabenjo, "Power system analysis and integration of the proposed Nigerian 750-kV power line to the grid reliability," *Social Netw. Appl. Sci.*, vol. 3, no. 12, pp. 1–15, Dec. 2021.
- [291] D. C. Idoniboyeobu, S. L. Braide, and K. O. Adebisi, "Reactive power compensation study of Ikeja West 330 kV transmission line for improved power transmission," Tech. Rep.
- [292] A. Abdulkareem, V. Oguntosin, O. Dare, and A. L. Mutalub, "The electricity security in Nigeria: Design and analysis of 750-kV mega grid," *Int. J. Eng. Res. Technol.*, vol. 13, no. 9, pp. 2219–2229, 2020.
- [293] N. Kanimozhi, S. Saranya, and R. Devipriya, "Transformation of smart grid to Internet of Energy," in *Blockchain-Based Syst. for Modern Energy Grid.* Amsterdam, The Netherlands: Elsevier, 2023, pp. 37–49.
- [294] V. Rajasekar and K. Sathya, "Blockchain utility in renewable energy," in *Blockchain-Based Systems for the Modern Energy Grid*, S. Padmanaban, R. K. Dhanaraj, J. B. Holm-Nielsen, S. Krishnamoorthi, and B. Balusamy, Eds. Academic, 2023, ch. 8, pp. 115–134. [Online]. Available: https://www.sciencedirect.com/science/article/pii/ B978032391850300010X, doi: 10.1016/B978-0-323-91850-3.00010-X.
- [295] C. Vanitha, S. Malathy, S. Krishna, and K. Vanitha, "Wireless communications in energy grid," in *Blockchain-Based Systems for the Modern Energy Grid*. Amsterdam, The Netherlands: Elsevier, 2023, pp. 71–82.
- [296] V. Rajasekar and K. Sathya, "Blockchain utility in renewable energy," in *Blockchain-Based Systems for the Modern Energy Grid.* Amsterdam, The Netherlands: Elsevier, 2023, pp. 115–134.
- [297] N. Kshetri, "Artificial intelligence in developing countries," *IT Prof.*, vol. 22, no. 4, pp. 63–68, Jul. 2020.
- [298] R. P. P. Smeets and N. A. Belda, "HVDC fault current interruption technology," in *Proc. 5th Int. Conf. Electr. Power Equip., Switching Technol. (ICEPE-ST)*, Oct. 2019, pp. 780–787.
- [299] K. Hojčková, B. Sandén, and H. Ahlborg, "Three electricity futures: Monitoring the emergence of alternative system architectures," *Futures*, vol. 98, pp. 72–89, Apr. 2018.
- [300] C. Aouadi, A. Abouloifa, I. Lachkar, M. Aourir, Y. Boussairi, and A. Hamdoun, "Nonlinear controller design and stability analysis for single-phase grid-connected photovoltaic systems," *Int. Rev. Autom. Control*, vol. 10, no. 4, p. 306, Jul. 2017.
- [301] S. Jiang, C. Wu, S. Gao, G. Pan, Y. Liu, X. Zhao, and S. Wang, "Robust frequency risk-constrained unit commitment model for AC–DC system considering wind uncertainty," *Renew. Energy*, vol. 195, pp. 395–406, Aug. 2022.
- [302] A. Iovine, M. J. Carrizosa, G. Damm, and P. Alou, "Nonlinear control for DC microgrids enabling efficient renewable power integration and ancillary services for AC grids," *IEEE Trans. Power Syst.*, vol. 34, no. 6, pp. 5136–5146, Nov. 2019.

- [303] M. Barzegar-Kalashani, B. Tousi, M. A. Mahmud, and M. Farhadi-Kangarlu, "Robust nonlinear sliding mode controllers for single-phase inverter interfaced distributed energy resources based on super twisting algorithms," *ISA Trans.*, vol. 123, pp. 61–75, Apr. 2022.
- [304] M. Ashtary, S. M. Hakimi, S. V. Naghavi, and V. Behnamgol, "Applying adaptive super-twisting controller for voltage regulation in a DC micro grid," Feb. 2023, doi: 10.21203/rs.3.rs-2534812/v1.
- [305] R. H. Byrne, T. A. Nguyen, D. A. Copp, B. R. Chalamala, and I. Gyuk, "Energy management and optimization methods for grid energy storage systems," *IEEE Access*, vol. 6, pp. 13231–13260, 2018.
- [306] M. M. Rana, "Modelling the microgrid and its parameter estimations considering fading channels," *IEEE Access*, vol. 5, pp. 10953–10958, 2017.
- [307] H. Jo, W. Lee, and H. Cha, "Three-phase voltage sag compensator for smart grid and infrastructure," in *Proc. IEEE Vehicle Power Propuls. Conf.*, Oct. 2012, pp. 1463–1468.
- [308] V. M. Mansoor, P. Nagaveni, A. Amudha, S. Divyapriya, G. Emayavaramban, M. S. Ramkumar, and M. SivaramKrishnan, "Throwing light on lightning," *Mater. Today, Proc.*, vol. 37, pp. 2572–2577, Jan. 2021.
- [309] M. S. Ramkumar, A. Amudha, M. Sivaramkrishnan, S. Divyapriya, P. Nagaveni, G. Emayavaramban, and V. Mansoor, "Micro grid based low price residential home to grid—Modeling and control of power management system," in *Proc. 4th Int. Conf. I-SMAC (IoT Social, Mobile, Analytics Cloud) (I-SMAC)*, Oct. 2020, pp. 1179–1183.
- [310] Y. Gui, F. Blaabjerg, X. Wang, J. D. Bendtsen, D. Yang, and J. Stoustrup, "Improved DC-link voltage regulation strategy for grid-connected converters," *IEEE Trans. Ind. Electron.*, vol. 68, no. 6, pp. 4977–4987, Jun. 2021.
- [311] A. Alsharif, C. W. Tan, R. Ayop, A. Dobi, and K. Y. Lau, "A comprehensive review of energy management strategy in vehicle-to-grid technology integrated with renewable energy sources," *Sustain. Energy Technol. Assessments*, vol. 47, Oct. 2021, Art. no. 101439.
- [312] R. Ahmed, S. Mahadzir, N. E. M. Rozali, K. Biswas, F. Matovu, and K. Ahmed, "Artificial intelligence techniques in refrigeration system modelling and optimization: A multi-disciplinary review," *Sustain. Energy Technol. Assessments*, vol. 47, Oct. 2021, Art. no. 101488.
- [313] N. Bansal, S. P. Jaiswal, and G. Singh, "Comparative investigation of performance evaluation, degradation causes, impact and corrective measures for ground Mount and rooftop solar PV plants—A review," *Sustain. Energy Technol. Assessments*, vol. 47, Oct. 2021, Art. no. 101526.
- [314] R. d'Amore-Domenech, Ó. Santiago, and T. J. Leo, "Multicriteria analysis of seawater electrolysis technologies for green hydrogen production at sea," *Renew. Sustain. Energy Rev.*, vol. 133, Nov. 2020, Art. no. 110166.
- [315] I. Ourya and S. Abderaff, "Clean technology selection of hydrogen production on an industrial scale in Morocco," *Results Eng.*, vol. 17, Mar. 2023, Art. no. 100815.
- [316] V. B. M. Krishna, V. Sandeep, S. S. Murthy, and K. Yadlapati, "Experimental investigation on performance comparison of self excited induction generator and permanent magnet synchronous generator for small scale renewable energy applications," *Renew. Energy*, vol. 195, pp. 431–441, Aug. 2022.
- [317] M. K. Rajak, J. Samanta, and R. Pudur, "A hardware-based novel approach for parallel operation of two differently rated SEIGs," *Results Eng.*, vol. 17, Mar. 2023, Art. no. 100825.
- [318] S. Chakraborty and R. Pudur, "Supply of single-phase power for rural area using three-phase self-excited induction generator," in *Proc. Asian Conf. Innov. Technol. (ASIANCON)*, Aug. 2021, pp. 1–6.
- [319] T. H. Le, "Connectedness between nonrenewable and renewable energy consumption, economic growth and CO<sub>2</sub> emission in Vietnam: New evidence from a wavelet analysis," *Renew. Energy*, vol. 195, pp. 442–454, Aug. 2022.
- [320] World Energy Investment 2017, Int. Energy Agency, 2017, p. 191. [Online]. Available: https://www.oecd-ilibrary.org/content/publication/ 9789264277854-en, doi: 10.1787/9789264277854-en.
- [321] J. Beerten, O. Gomis-Bellmunt, X. Guillaud, J. Rimez, A. van der Meer, and D. Van Hertem, "Modeling and control of HVDC grids: A key challenge for the future power system," in *Proc. Power Syst. Comput. Conf.*, Aug. 2014, pp. 1–21.
- [322] U. N. Gnanarathna, A. M. Gole, and R. P. Jayasinghe, "Efficient modeling of modular multilevel HVDC converters (MMC) on electromagnetic transient simulation programs," *IEEE Trans. Power Del.*, vol. 26, no. 1, pp. 316–324, Jan. 2011.
- [323] W. Li and J. Bélanger, "An equivalent circuit method for modelling and simulation of modular multilevel converters in real-time HIL test bench," *IEEE Trans. Power Del.*, vol. 31, no. 5, pp. 2401–2409, Oct. 2016.

- [324] N. Ahmed, L. Ängquist, S. Mahmood, A. Antonopoulos, L. Harnefors, S. Norrga, and H.-P. Nee, "Efficient modeling of an MMC-based multiterminal DC system employing hybrid HVDC breakers," *IEEE Trans. Power Del.*, vol. 30, no. 4, pp. 1792–1801, Aug. 2015.
- [325] G. Z. D. Rubens and L. Noel, "The non-technical barriers to large scale electricity networks: Analysing the case for the U.S. and EU supergrids," *Energy Policy*, vol. 135, Dec. 2019, Art. no. 111018.
- [326] A. L'Abbate, G. Fulli, R. Calisti, P. Minnebo, A. Purvins, H. Gerbelova, M. Ardelean, and A. Zani, "Evolution of the pan-European transmission system towards the development of a prospective global power grid," in *Proc. 16th Int. Conf. Eur. Energy Market (EEM)*, Sep. 2019, pp. 1–5.
- [327] A. Aghahosseini, D. Bogdanov, L. S. N. S. Barbosa, and C. Breyer, "Analysing the feasibility of powering the Americas with renewable energy and inter-regional grid interconnections by 2030," *Renew. Sustain. Energy Rev.*, vol. 105, pp. 187–205, May 2019.
- [328] S. F. Rafique, P. Shen, Z. Wang, R. Rafique, T. Iqbal, S. Ijaz, and U. Javaid, "Global power grid interconnection for sustainable growth: Concept, project and research direction," *IET Gener., Transmiss. Distrib.*, vol. 12, no. 13, pp. 3114–3123, Jul. 2018.
- [329] S. Chatzivasileiadis, D. Ernst, and G. Andersson, "Global power grids for harnessing world renewable energy," in *Renewable Energy Integration*. Amsterdam, The Netherlands: Elsevier, 2017, pp. 161–174.
- [330] P. Deane and M. Brinkerink, "Connecting the continents—A global power grid," *IEEE Power Energy Mag.*, vol. 18, no. 2, pp. 121–127, Mar. 2020.
- [331] S. Bloem, M. Swilling, and K. Koranteng, "Taking energy democracy to the streets: Socio-technical learning, institutional dynamism, and integration in South African community energy projects," *Energy Res. Social Sci.*, vol. 72, Feb. 2021, Art. no. 101906.
- [332] R. Smeets, L. Van der Sluis, M. Kapetanovic, D. F. Peelo, and A. Janssen, Switching in Electrical Transmission and Distribution Systems. Hoboken, NJ, USA: Wiley, 2014.
- [333] X. Xiang, S. Fan, Y. Gu, W. Ming, J. Wu, W. Li, X. He, and T. C. Green, "Comparison of cost-effective distances for LFAC with HVAC and HVDC in their connections for offshore and remote onshore wind energy," *CSEE J. Power Energy Syst.*, vol. 7, no. 5, pp. 954–975, Sep. 2021.
- [334] S. Liu, M. Saeedifard, and X. Wang, "Analysis and control of the modular multilevel matrix converter under unbalanced grid conditions," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 6, no. 4, pp. 1979–1989, Dec. 2018.
- [335] S. Liu, X. Wang, Y. Meng, P. Sun, H. Luo, and B. Wang, "A decoupled control strategy of modular multilevel matrix converter for fractional frequency transmission system," *IEEE Trans. Power Del.*, vol. 32, no. 4, pp. 2111–2121, Aug. 2017.
- [336] M. Safiuddin and R. Finton, "Global renewable energy grid project— Integrating renewables via high-voltage direct current and centralized storage," in *Eissa MM*. New York, NY, USA: Academic, 2019, pp. 85–99.
- [337] R. Bucher, "Eliminating energy poverty in Africa by integrating topdown and bottom-up electrification concepts, i.e. Cross-border backbone networks & solar-hybrids," in *Proc. IEEE PES/IAS PowerAfrica*, Aug. 2020, pp. 1–5.
- [338] O. Adeoye and C. Spataru, "Quantifying the integration of renewable energy sources in West Africa's interconnected electricity network," *Renew. Sustain. Energy Rev.*, vol. 120, Mar. 2020, Art. no. 109647.



**MUHAMMAD ADNAN** (Member, IEEE) received the B.S. degree in electrical engineering from the National University of Computer and Emerging Sciences, Peshawar, Pakistan, in 2013, and the M.S. degree in electrical engineering from the COMSATS Institute of Information and Technology, Islamabad, Pakistan, in 2015. He is currently pursuing the Ph.D. degree in electrical engineering with the National University of Computer and Emerging Sciences. Since 2017,

he has been a Research Fellow with the Department of Electrical Power Engineering, National University of Computer and Emerging Sciences. His research interests include energy management systems, load flow balancing, load forecasting, power systems dynamic analysis, protection, stability, and intelligent control in renewable energy resources using a fuzzy controller and unified power flow controller.



**YAZEED GHADI** (Member, IEEE) received the Ph.D. degree in electrical and computer engineering from The University of Queensland. His dissertation on developing novel hybrid plasmonic photonic on-chip biochemical sensors received the Sigma Xi Best Ph.D. Thesis Award. He is currently an Assistant Professor of software engineering with Al Ain University. He was a Postdoctoral Researcher with The University of Queensland, before joining Al Ain University. He has pub-

lished more than 25 peer-reviewed journal articles and conference papers. He holds three pending patents. His current research is on developing novel electro-acoustic-optic neural interfaces for large-scale high-resolution electrophysiology and distributed optogenetic stimulation. He was a recipient of several awards.



**IJAZ AHMED** (Member, IEEE) received the B.S. degree in electrical engineering from the COM-SATS Institute of Information and Technology, Abbottabad, Pakistan, in 2017, and the M.S. degree from the School of Electrical Engineering and Computer Science (SEECS), National University of Science and Technology (NUST), Islamabad, Pakistan, in 2021. He is currently a Research Assistant (RA) in electrical engineering with the National University of Computer and

Emerging Sciences. His research interests include energy management systems, load flow balancing, load forecasting, power systems dynamic analysis, protection, stability, and intelligent control in renewable energy resources using a fuzzy controller and unified power flow controller.



**MANSOOR ALI** (Member, IEEE) received the B.S. degree in electrical engineering from the National University of Computer and Emerging Sciences, Peshawar, Pakistan, in 2013, the M.S. degree in electrical engineering from the CECOS University of IT and Emerging Sciences, Peshawar, in 2016, and the Ph.D. degree in electrical engineering with the National University of Computer and Emerging Sciences in 2020. Since 2021, he has been a Postdoctoral Research Fellow

with the Department of Electrical Engineering at École de technologie supérieure ÉTS, Montreal, Canada. His research interests include load forecasting in power system networks, fuzzy control, and power flow control under various disturbances.